Topical Review

Gravitational Decoherence

Angelo Bassi^{1,2}, André Großardt^{3,4} and Hendrik Ulbricht⁵

¹ Department of Physics, University of Trieste, Strada Costiera 11, 34151 Miramare-Trieste, Italy

 2 Istituto Nazionale di Fisica Nucleare, S
ezione di Trieste, Via Valerio 2, 34127 Trieste, Italy

³ ZARM, University of Bremen, Am Fallturm 2, 28359 Bremen, Germany

⁴ Centre for Theoretical Atomic, Molecular, and Optical Physics, School of

Mathematics and Physics, Queen's University Belfast, BT7 1NN, United Kingdom ⁵ School of Physics and Astronomy, University of Southampton, SO17 1BJ, United Kingdom

E-mail: bassi@ts.infn.it

Abstract. We discuss effects of loss of coherence in low energy quantum systems caused by or related to gravitation, referred to as gravitational decoherence. These effects, resulting from random metric fluctuations, for instance, promise to be accessible by relatively inexpensive table-top experiments, way before the scales where true quantum gravity effects become important. Therefore, they can provide a first experimental view on gravity in the quantum regime. We will survey models of decoherence induced both by classical and quantum gravitational fluctuations; it will be manifest that a clear understanding of gravitational decoherence is still lacking. Next we will review models where quantum theory is modified, under the assumption that gravity causes the collapse of the wave functions, when systems are large enough. These models challenge the quantum-gravity interplay, and can be tested experimentally. In the last part we have a look at the state of the art of experimental research. We will review efforts aiming at more and more accurate measurements of gravity (G and g) and ideas for measuring conventional and unconventional gravity effects on nonrelativistic quantum systems.

Submitted to: Class. Quantum Grav.

1. Introduction

The unification of quantum physics with general relativity is perhaps the most important and ambitious open problem in modern physics. Both theories work excellently when taken separately; when rigidly patched together to explain the evolution of the universe, they give impressive results, the discovery of the cosmic microwave background and its properties being the most important one, although the resolution of the dark matter and dark energy puzzle might ultimately show that such patching was wrong in the first place [1]. Yet, a coherent unified theory does not exist. The general attitude is to quantize gravity in some way, which produced a variety of sophisticated formulations, string theory [2] and loop quantum gravity [3] arguably being the most prominent ones. A minority of the scientific community favors the opposite approach, to "gravitize" quantum mechanics [4], or to radically reconsider both theories [5, 6, 7].

From the experimental point of view, not much is really known. We know that matter is quantum and, when in a classical (i. e. localized in *space*) state, it reacts to gravity as predicted by Newton or by Einstein. This has been verified for source masses ranging from 10^{-2} kg [8] to stars, galaxies, and beyond, and from distances ranging from 10^{-2} m [8] to cosmological distances (but here again, the resolution of the dark matter/energy problem might reveal surprises). We also know that matter in spatial *quantum superpositions* responds to an external classical gravitational field as predicted by simply adding the Newtonian gravitational potential to the Schrödinger equation [9, 10, 11].

However, the answers, both theoretical and experimental, to the big questions are still missing: Is gravity quantum? Do gravitons, the quantum counterpart of the recently discovered classical gravitational waves [12], really exist? What is the gravitational field generated by a quantum superposition of matter? Does quantum mechanics still hold true in presence of sufficiently strong gravitational fields? The fact that after almost one century of intense theoretical research we do not have a solid clue about the above questions, is perhaps the indication that radically new thinking is required to approach the problem of unifying quantum mechanics and gravity. And it certainly is not a viable option to simply wait until the Planck scale ($\ell_{\rm Pl} = 10^{-35}$ m, $t_{\rm Pl} = 10^{-44}$ s, $m_{\rm Pl} = 10^{-8}$ kg) becomes accessible experimentally in order to have an indication on how to proceed theoretically, as this will require a long time.

In recent years, however, pushed by the impressive technological developments, the scientific community is exploring the possibility that the quantum-gravity interplay can be seen far away from the Planck scale, in nonrelativistic quantum systems and in table-top experiments [13]. This is the context of this review article.

Whatever the final outcome of the quantum gravity riddle, one prediction seems to be rather robust: spacetime as a dynamical variable fluctuates, and these fluctuations cause decoherence in quantum systems, which can be revealed, at least in principle, in matter-wave interferometers, for example. These fluctuations can be classical, quantum, or both. Classical fluctuations can be a relic of primordial gravitational waves, generated during the early evolution of the universe, a relic of the big bang together with the cosmic microwave background, or the sum of the gravitational waves randomly produced by the large variety of physical sources scattered through the universe. Their spectrum is estimated theoretically, and constrained by observations [12]. Quantum fluctuations arise from the quantization procedure, as for any field. Not much is known, as a consistent quantum theory of gravity is missing.

Gravitational decoherence, as any noise, modifies the evolution of quantum systems with respect to what is predicted by the Schrödinger equation applied to

Gravitational Decoherence

isolated systems. Under general assumptions, Gorini–Kossakowski–Sudarshan [14] and independently Lindblad [15], proved that the general structure of the master equation for the density matrix ρ_t describing a quantum system interacting with an external environment is

$$\frac{\mathrm{d}}{\mathrm{d}t}\rho_t = -\frac{\mathrm{i}}{\hbar}[\rho_t, H] + \sum_n \lambda_n \left(L_n \rho_t L_n^\dagger - \frac{1}{2} \rho_t L_n^\dagger L_n - \frac{1}{2} L_n^\dagger L_n \rho_t \right),\tag{1}$$

where H is the standard quantum Hamiltonian, possibly containing a corrective term arising from the interaction with the environment, the Lindblad operators L_n generate decoherence and dissipation, and the positive coefficients λ_n set the strength of these effects. The goal of theoretical research on gravitational decoherence is to derive an explicit expression for the Lindblad operators and the coefficients, by assuming that quantum matter interacts with a stochastic gravitational background. We will review some of the models which have been presented in the literature.

A remark is due concerning the notion of "gravitational decoherence", which is used in quite different contexts with different meanings. For the purpose of this review article, we use it in a rather broad sense, referring to any loss of coherence of quantum *matter* fields, whose cause can be related to gravitational effects—be it due to a classical or quantum gravitational background (as discussed in section 2) or due to modifications of quantum mechanics motivated from gravity (as in section 3). As the main focus of this review article are experiments which are feasible in the laboratory (see section 4), our main concern are effects in the limit of nonrelativistic systems and weak gravitational fields.

The emergence of a classical spacetime structure from quantum gravity, as it is discussed, for instance, in minisuperspace models [16, 17] where the Friedmann universe acts as "system" and higher multipoles are treated as the "environment", is also often referred to as "gravitational decoherence". This, however, is an effect on where the quantized spacetime itself decoheres, rather than quantum matter in the laboratory, and is beyond the scope of this review article in which we are interested in understanding the effects of gravity on such laboratory quantum systems.

In section 2 we will consider the standard setup: (nonrelativistic) quantum mechanical matter interacting with a random gravitational field. It turns out that there exist a large number of quite different approaches towards this problem, many of them motivated by *quantum* gravitational fluctuations but nonetheless applicable to perturbations of a classical spacetime background. All the approaches we present make use of simplifications and approximations such as only considering the effect of conformal metric fluctuations. A general treatment of decoherence due to classical spacetime fluctuations is still lacking.

We will proceed to the discussion of quantum gravity effects as a possible sources of spacetime fluctuations causing decoherence. Many such proposals exist, and we will review effects of a thermal graviton bath, of a minimal length scale, and of fluctuations of the time variable. Finally, we have a look into a recently proposed decoherence effect due to gravitational time dilation which has been the starting point of an ongoing controversy. This effect is somewhat different from the previously discussed decoherence effects, as decoherence occurs due to a coupling of the center-of-mass motion (acting as "system") to internal degrees of freedom (acting as "bath") of a complex quantum system, and gravity (or rather relativity) solely enters in the role of a transmitter which enables this coupling.

In section 3 we will go beyond the standard quantum formalism. Quantum theory has originated a lively debate about its meaning, as it gives a fundamental and distinctive role, in its very formulation, to measurements and observations, while these should represent only a subclass of all possible physical events, described at least in principle by the dynamical equations of the theory. This is not the case with quantum mechanics. Several solutions to this conundrum have been proposed, among which is the idea that the Schrödinger equation should be modified and supplemented with nonlinear and stochastic terms, which end up in a collapse of the wave function in specific situations, like measurement processes. The phenomenology of models of spontaneous wave function collapse is well understood, and experiments are placing stronger and stronger bounds on the collapse parameters. However a relevant question remains open: if the collapse is real as predicted by these models, what is its origin? Of course one can always claim that it is an intrinsic property of nature, but intuition tells that there should be an explanation.

People, most notably Roger Penrose, proposed the idea that gravity might be the ultimate explanation for the collapse of the wave function. Gravity is universal, and its magnitude increases with the mass, which matches the requirements of the collapse, which is also universal, if quantum theory is, and becomes relevant when macroscopic systems are involved. In addition, there is no evidence thus far, nor a consensus among theoretical physicists, whether gravity should be quantized, therefore the possibility remains open for it to give the required nonlinear coupling to break quantum linearity and the associated measurement problem.

We will review four major attempts to link the collapse of the wave function to gravity: the Diósi–Penrose model, Adler's model, Károlyházy's model, and the Schrödinger–Newton equation.

The Diósi–Penrose model falls within the class of models of spontaneous wave function collapse, and it is specified by setting the correlation function of the noise equal to the Newtonian gravitational potential. The model was first proposed by Diósi, and later revived by Penrose, based on independent arguments on the behavior of spacetime in presence of quantum superpositions.

Adler's model also falls within the class of models of spontaneous wave function collapse, but now the assumption is that gravity has an irreducible, complex, rapidly fluctuating component. The correlation function of this noise is left unspecified, and eventually has to be explained by a yet-to-be-formulated underlying theory.

Károlyházy's model is based on the assumption that spacetime is fuzzy below a

specific scale, and this fuzziness causes a lack of quantum coherence in space (and time) for material systems, which is the stronger, the larger the systems. The quantum-gravity coupling is not modified, therefore it is linear and so technically this model falls under the class of decoherence models discussed in section 2, not under the class of collapse models. However, since Károlyházy's motivations spring more from the analysis of the quantum measurement problem, rather than from an attempt of arriving at a quantum theory of gravity, it is better discussed in context of the other models in section 3.

Finally, the Schrödinger–Newton equation is discussed, which is based on the assumption that matter is fundamentally quantum but gravity is fundamentally classical. Then the most natural way to couple matter and gravity leads to a nonlinearly modified Schrödinger equation of the solitonic type, where gravity attracts gravitationally different parts of the wave function. This equation does not fall within the class of models of spontaneous wave function collapse, although it also causes its "collapse" in a way which we will discuss.

In section 4 we move from theory to experiment, and we will review the state of the art in experimental research of the quantum-gravity interplay at low energy, typically in table-top experiments. We will summarize the most advanced and precise Newtonian and non-Newtonian gravity measurements by torsion pendulum type experiments, and also by matter-wave interferometry. Matter-wave interferometers, such as neutron, atom, and molecule interferometers, are sensitive to phase shifts triggered by gravitational effects, while the source mass, from which the gravitational field originates, is typically larger than 1 kg and usually not closer than 1 m to the test mass. Interestingly, gravity experiments involving source and test bodies with the smallest gravitational force between each other are torsion pendulum experiments. We will further summarize proposals for tests of gravity which aim to go beyond the present limitations in decreasing mass and distance in gravitational force sensing, as well as in increasing the mass in spatial quantum superpositions, and which aim to test general relativistic effects and even quantum gravity effects, amongst which are the tests of collapse type effects and gravitational decoherence.

2. Classical and quantum spacetime fluctuations

Any quantum state subject to some environment will become entangled to this environment, and undergo decoherence when environmental degrees of freedom are traced out. The gravitational interaction with the spacetime background will lead to decoherence as well. It is, however, different from the usual decoherence sources in at least two respects:

- The gravitational interaction cannot be omitted, contrary to the interaction with, for instance, gas particles and radiation which can—at least in principle—be evacuated or shielded completely.
- As, to date, gravity is only understood classically, the theoretical description of gravitational decoherence is necessarily incomplete, and predictions may depend

on assumptions about the correct behavior of gravity at the quantum scale.

These distinctive features of gravitational decoherence make it very interesting to understand the details of the possible decoherence mechanisms. Does gravitational decoherence put an ultimate bound on the scales at which quantum features of nature can be observed? And are there predictions that differ depending on whether the gravitational background is treated classically or quantum mechanically? Could such predictions even lead to observable consequences which give insight into a quantum theory of gravity?

On the other hand, due to the lack of a consistent and broadly accepted quantum theory of gravity from which decoherence effects at low energies could be derived in a rigorous way, there is no unanimous theoretical description for gravitational decoherence. Even the theoretical tools within quantum field theory on curved spacetime [18, 19] are not far enough developed for definite predictions at the level of low energy laboratory physics.

Decoherence effects are, therefore, discussed in very different contexts, where quantum gravitational features are introduced in a rather ad-hoc manner. Many of these ideas are based on either some specific quantum gravitational model, and are then model dependent, or on general ideas about gravity at the Planck scale, such as the existence of a minimal length and time scale, or spacetime fluctuations which can be implemented in multiple ways.

The latter idea of a fluctuating spacetime is often introduced in terms of stochastic perturbations of the *classical* background metric. Such perturbations can be of fundamental origin, e. g. as a consequence of quantum gravitational fluctuations. They can also stem from gravitational wave noise which itself may be modeled classically or in terms of quanta of the gravitational field, i. e. gravitons. However, as the theoretical description of laboratory quantum systems on curved spacetime background itself still poses many questions, the respective models are based on highly simplified assumptions, and the resulting predictions can differ considerably depending on the choice of these assumptions.

We first review some of the predictions and models for effects resulting from fluctuations of the classical spacetime background in section 2.1. Then, in section 2.2, we discuss how these classical fluctuations can be linked to quantum gravitational effects, and how decoherence is described in the formalism of perturbative quantum gravity.

Finally, we review the recent discussion about a decoherence effect which has been attributed to gravitational time dilation [20] in section 2.3. This effect is different in some regards from the previously mentioned ones, as it occurs in a homogeneous gravitational field, and gravity acts as a transmitter of entanglement between center of mass and internal degrees of freedom. Hence, the role of the environment which allows for decoherence is not played by gravity itself in this case.

2.1. Classical spacetime fluctuations

Decoherence can already occur in the straightforward approach of a quantum wave function that is evolving on a curved spacetime metric, if this spacetime background contains stochastic fluctuations. A common feature of all approaches to model such fluctuations is that they assume a weak gravitational field, such that the linear approximation to general relativity applies, i.e. the metric tensor can be written as $g_{\mu\nu} = \eta_{\mu\nu} + \epsilon h_{\mu\nu}$ with a small parameter ϵ .

2.1.1. Gravitational phase changes for matter-waves At the lowest significant order, gravitational effects on matter-waves reveal themselves in the form of a phase shift of the wave function. An instructive derivation can be given in terms of the Wentzel-Kramers-Brillouin (WKB) approximation [21, 22, 23]. For this, we start from the Klein-Gordon equation for a scalar field ϕ in a curved spacetime:

$$0 = \left[g^{\mu\nu} \nabla_{\mu} \nabla_{\nu} + \frac{m^2 c^2}{\hbar^2}\right] \phi$$
$$= \left[g^{\mu\nu} \left(\frac{\partial}{\partial x^{\mu}} \frac{\partial}{\partial x^{\nu}} - \Gamma^{\lambda}{}_{\mu\nu} \frac{\partial}{\partial x^{\lambda}}\right) + \frac{m^2 c^2}{\hbar^2}\right] \phi, \qquad (2)$$

where the covariant derivatives ∇_{μ} in the first line are expressed in terms of the Christoffel symbols and partial derivatives in the second line. Throughout this review article we adopt the mostly minus metric signature (+, -, -, -). The WKB approximation is based on the ansatz

$$\phi = \mathrm{e}^{\frac{1}{\hbar}\varphi} \tag{3}$$

for the field, followed by a series expansion in terms of \hbar . Inserting this ansatz into (2) yields

$$0 = g^{\mu\nu} \left(\frac{\mathrm{i}}{\hbar} \frac{\partial}{\partial x^{\mu}} \frac{\partial}{\partial x^{\nu}} \varphi - \frac{\mathrm{i}}{\hbar} \Gamma^{\lambda}{}_{\mu\nu} \frac{\partial}{\partial x^{\lambda}} \varphi - \frac{1}{\hbar^2} \frac{\partial}{\partial x^{\mu}} \varphi \frac{\partial}{\partial x^{\nu}} \varphi \right) + \frac{m^2 c^2}{\hbar^2} \,, \qquad (4)$$

which at lowest order in \hbar results in

$$g^{\mu\nu} \frac{\partial}{\partial x^{\mu}} \varphi \frac{\partial}{\partial x^{\nu}} \varphi = m^2 c^2 \,. \tag{5}$$

(5) is the equivalent to the eikonal equation in the case of geometrical optics (i. e. the case m = 0 [24]). Now, in the linear approximation $g_{\mu\nu} = \eta_{\mu\nu} + \epsilon h_{\mu\nu}$, the phase can be split as $\varphi = \varphi^{(0)} + \epsilon \varphi^{(1)}$ which results in the following set of two equations at the zeroth and first order in ϵ [25]:

$$m^2 c^2 = \eta^{\mu\nu} \frac{\partial}{\partial x^{\mu}} \varphi^{(0)} \frac{\partial}{\partial x^{\nu}} \varphi^{(0)}$$
(6a)

$$0 = 2 \eta^{\mu\nu} \frac{\partial}{\partial x^{\mu}} \varphi^{(0)} \frac{\partial}{\partial x^{\nu}} \varphi^{(1)} + h^{\mu\nu} \frac{\partial}{\partial x^{\mu}} \varphi^{(0)} \frac{\partial}{\partial x^{\nu}} \varphi^{(0)} .$$
 (6b)

(6*a*) is simply the eikonal equation in flat Minkowski space. Defining the constant vector $\xi^{\mu} = (E/c, p^1, p^2, p^3)$, where *E* is the energy and p^i the momentum of the

particle described by (2), the relativistic energy-momentum relation takes the form $\xi^{\mu}\xi_{\mu} = E^2/c^2 - p^2 = m^2c^2$, and it is evident that

$$\varphi^{(0)} = \eta_{\mu\nu} \,\xi^{\mu} \,x^{\nu} + \varphi^{(0)}_{0} \,, \tag{7}$$

with constant $\varphi_0^{(0)}$ solves (6*a*).

It can then be shown [25] that the gravitational contribution to the phase along the particle trajectory, parametrized by τ as $x^{\mu} = \tau \xi^{\mu} + x_0^{\mu}$, satisfies the differential equation

$$\frac{\mathrm{d}\varphi^{(1)}}{\mathrm{d}\tau} = \frac{1}{2} h_{\mu\nu} \xi^{\mu} \xi^{\nu} \,. \tag{8}$$

According to Linet and Tourrenc [25] (cf. also Refs. [26, 27]), the total phase difference resulting after integration along the particle trajectory can be categorized into three different types of gravitational phase shift effects:

(i) The term resulting from the h_{00} metric element is the contribution of the Newtonian gravitational potential,

$$\Delta \varphi = \frac{E}{2} \int_{t_0}^{t_1} h_{00} \,\mathrm{d}t \,. \tag{9}$$

In the Earth's homogeneous potential this yields the experimentally confirmed [9] phase factor $\Delta \varphi = m g \Delta z t$.

(ii) The contribution from the h_{0i} metric elements is

$$\Delta \varphi = c \int_{t_0}^{t_1} h_{0i} p^i \,\mathrm{d}t \,. \tag{10}$$

This effect can be understood as stemming from the rotation of the gravitational source (i.e. the Lense-Thirring effect [28]), which is closely related to the Sagnac effect [29] (phase shift in a rotating interferometer).

(iii) Finally, the third contribution from the spatial h_{ij} terms results from planar gravitational waves. The phase shift is

$$\Delta \varphi = \frac{c^2}{E} \int_{t_0}^{t_1} h_{ij} p^i p^j \,\mathrm{d}t \,. \tag{11}$$

In the transverse traceless (TT) gauge, where only the spatial components h_{ij} of the metric perturbation are nonzero, gravitational waves can be described by a mode decomposition [30]:

$$h_{ij}(x) = \sum_{\sigma \in \pm} \varepsilon_{ij}^{\sigma} \int \frac{\mathrm{d}^4 k}{(2\pi)^4} h^{\sigma}(k) \,\mathrm{e}^{-\mathrm{i}k_{\mu}x^{\mu}} \,, \tag{12}$$

where ε is the polarization tensor for the two possible polarizations $\sigma \in \pm$ and h^{σ} are the coefficients of the expansion for the respective polarizations. Reynaud *et al* [30] consider the special case of unpolarized, isotropic waves, in which case the metric components can be characterized by a spectral distribution $S(\omega)$, e.g. for the *xy*-component:

$$\langle h_{12}(t)h_{12}(0)\rangle = \int \frac{\mathrm{d}\omega}{2\pi} S(\omega) \,\mathrm{e}^{-\mathrm{i}\omega t} \,. \tag{13}$$

The phase difference then becomes a stochastic variable, $\delta\varphi$. If this gravitational noise is assumed to be Gaussian, the fringe visibility in an interferometer will be reduced as follows:

$$\mathcal{V} = \langle \exp\left(\frac{\mathrm{i}}{\hbar}\delta\varphi\right) \rangle = \exp\left(-\frac{\langle(\delta\varphi)^2\rangle}{2\hbar^2}\right) \equiv \exp\left(-\frac{\Delta\varphi^2}{2\hbar^2}\right) \,. \tag{14}$$

The variance in the exponent depends not only on the spectrum of the gravitational wave background S, but also on the geometry of the interferometer, encoded in an "apparatus response function" A. It can then be written in the frequency domain as [31]

$$\Delta \varphi^2 = \int \frac{\mathrm{d}\omega}{2\pi} S(\omega) A(\omega) F(\omega) , \qquad (15)$$

where F is a filter function ensuring that the integral only runs over the relevant (unobserved) frequencies.

The apparatus function $A(\omega)$ generally has a complicated form. In the idealized case of a rhombic Mach-Zehnder interferometer of arm length L, with small aperture angle α , and particles with nonrelativistic velocities $v \ll c$, it is given by [31]

$$A_{\rm MZ}(\omega) = \frac{4 m^2 v^4}{\hbar^2 \omega^2} \sin^2(\alpha) \left(1 - \cos\left(\frac{\omega L}{v}\right)\right)^2 \tag{16}$$

Lamine *et al* [31] argue that for a measurable decoherence effect, i. e. a variance $\Delta \varphi^2$ of the order of unity, even under ideal conditions in a wide angle (90°) Mach-Zehnder interferometer one would need, for instance, a femtogram particle with a velocity of 1 km/s and 1 m arm length (see section 4.4 C).

Nonetheless, such order of magnitude estimations can only provide a first impression, and the many approximations that need to be made in the course of arriving at this result (isotropy, Gaussianity, specific interferometer design, etc.) leave room for unexpected effects.

2.1.2. Conformal metric perturbations In order to go beyond the discussion based on the semi-classical WKB approximation and the phase change of a wave function on a gravitational background, one is looking for simple models for quantum matter in a fluctuating spacetime.

One possible approach is to model fluctuations of spacetime in the form of conformal transformations of the metric, in which case only a single scalar field determines the metric structure. This has been done by Sánchez-Gómez [32] for a non-propagating fluctuation and by Power and Percival [33] for conformal gravitational waves.

Conformal fluctuations of the flat Minkowski metric are such transformations which can be obtained by a local rescaling (Weyl transformations):

$$g_{\mu\nu} = \Omega(x)^2 \eta_{\mu\nu} , \qquad (17)$$

with some spacetime function Ω . Generally, Einstein's field equations can be derived from the Einstein-Hilbert action

$$S = \frac{1}{16\pi G} \int \mathrm{d}^4 x \sqrt{-g} R \,, \tag{18}$$

where R is the Ricci curvature scalar [28, 34]. In order to rewrite the Einstein-Hilbert action in a more suitable form, it is convenient to define $A(x) = \Omega(x) - 1$. The action then takes the form of the Klein-Gordon action[‡]

$$\frac{3}{8\pi G} \int d^4x \, \eta^{\mu\nu} \frac{\partial}{\partial x^{\mu}} A(x) \frac{\partial}{\partial x^{\nu}} A(x) \,, \tag{19}$$

and by variation of this action one finds that the "conformal field" A(x) satisfies the massless, flat space Klein-Gordon equation $\eta^{\mu\nu}\partial_{\mu}\partial_{\nu}A(x) = 0$. The wave solutions of this equations can be interpreted as gravitational waves traveling at the speed of light c.

A matter-wave evolving in this curved spacetime will, at lowest order, be evolving in a Newtonian gravitational potential Φ , which is related to the metric in the usual way [34] by $g_{00} = 1 + 2\Phi/c^2$. The evolution is then described by a nonrelativistic potential

$$V(x) = m\Phi = \frac{mc^2}{2} \left(g_{00} - 1\right) = mc^2 A(x) + \frac{mc^2}{2} A(x)^2.$$
(20)

Note that the gravitational constant G does not appear anywhere in (20). In general relativity, Einstein's field equations which follow from the action (18) ascertain that the metric components in the Newtonian limit depend on G, e. g. $g_{00} = 1 - 2Gm/(rc^2)$. In contrast, the Klein-Gordon equation which follows from the variation of the conformal action (19) does not depend on G. This is not surprising, as the conformal approach *violates* Einstein's field equations. The easiest way to see this is by noting that the continuity equation $\nabla^{\nu}T_{\mu\nu} = 0$ for the stress-energy tensor, which must be satisfied for a solution of general relativity, is invariant under conformal transformations $g_{\mu\nu} \rightarrow \Omega^2 g_{\mu\nu}$ only for trace free stress-energy tensors, i. e. for $T^{\mu}_{\mu} = 0$ [34], which is usually not the case. The conformal gravitational waves modeled by the field A(x), therefore, should be considered as fluctuations of the Minkowski vacuum rather than solutions of Einstein's equations.

If the conformal A(x) is supposed to be only a stochastic fluctuation around the Minkowski metric, then its average values must satisfy $\langle A(x) \rangle = \langle \partial_{\mu} A(x) \rangle =$ $\langle A(x) \partial_{\mu} A(x) \rangle = 0$. The first relation is due to the fact that there should not be an overall imprint of the stochastic fluctuations, and the last two relations are consequences of the conformal symmetry [35].

With the assumption of a Gaussian white noise in all four dimension, i.e.

$$\langle A(x)A(x')\rangle = A_0^2 \exp\left(-\frac{1}{L^2}\sum_{\mu=0}^3 (x^{\mu} - x^{\mu'})^2\right),$$
 (21)

Sánchez Gómez [32] derives the (one-particle) master equation

$$\frac{\partial}{\partial t}\rho(\mathbf{x},\mathbf{x}';t) = -\frac{\mathrm{i}}{\hbar} \left[H_0,\rho(\mathbf{x},\mathbf{x}';t)\right] - \frac{8m^2c^3A_0^2L}{\hbar^2} \min\left(\frac{(\mathbf{x}-\mathbf{x}')^2}{L^2};1\right)\rho(\mathbf{x},\mathbf{x}';t),\qquad(22)$$

where the first term is the standard unitary quantum evolution term for the Hamiltonian H_0 without the stochastic fluctuations.

[‡] We are only interested in the physical case of D = 4 dimensions, for $D \neq 4$ the redefinition of $\Omega(x)$ must be chosen differently.

Gravitational Decoherence

The gravitational constant enters again when the field amplitude A_0 is associated with the Planck scale—thus, by bringing quantum gravity argument into the game (cf. also section 2.2.3). The largest distance of two points which behave coherently under a metric fluctuation is the length scale L. Given the Gaussian distribution of the length rescalings, the uncertainty of this length scale due to the metric fluctuation will be $A_0^2 L^2$. Requiring that this amount equals the Planck length $\ell_{\rm Pl}^2 = \hbar G/c^3$ yields $A_0^2 \approx \hbar G/(L^2c^3)$. The pre-factor in (22) then is of the order of $Gm^2/(\hbar L)$.

For a single particle, effects are minuscule (e. g. for $L = 1 \ \mu m$ one finds a coherence over 1 cm for more than 10^{30} s). For a macroscopic system, however, the pre-factor Gm^2/L^3 in (22) becomes $Gm\rho$, and can provide a significant amount of decoherence. Fluctuations that are caused by astrophysical gravitational waves rather than quantum gravity can of course become arbitrarily large, in principle, depending on their source, and there is no a-priori reason for them to be related to the Planck length in any way.

The fluctuations considered above are isotropic and can be thought of as nonpropagating. Contrary to these non-propagating fluctuations, Power and Percival [33] consider conformal waves that are directed in positive and negative x-direction, traveling with velocity c. For this purpose, they split the wave A(x) in the parts propagating in those two directions:

$$A(t,x) = A_0 \left(\xi^+ (ct - x) + \xi^- (ct + x) \right) , \qquad (23)$$

where A_0 is the amplitude and ξ^{\pm} are the fluctuations for the two propagation directions.

The average values must vanish as before, specifically $\langle \xi^{\pm} \rangle = 0$. The first order correlation is $\langle \xi^{+}(x)\xi^{-}(x') \rangle = 0$ and $\langle \xi^{+}(x)\xi^{+}(x') \rangle = \langle \xi^{-}(x)\xi^{-}(x') \rangle = g^{(1)}(x-x')$. Power and Percival [33] use a Dyson expansion in order to determine the time evolution of the stochastic operator:

$$\rho(t) = \rho(0) - \frac{i}{\hbar} \int_0^t dt' \left(\langle H(t') \rangle \rho(0) + \rho(0) \langle H^{\dagger}(t') \rangle \right)
- \frac{1}{\hbar^2} \int_0^t dt' \int_0^{t'} dt'' \left(\langle H(t') H(t'') \rangle \rho(0) + \langle H(t') \rho(0) H^{\dagger}(t'') \rangle
+ \rho(0) \langle H^{\dagger}(t') H^{\dagger}(t'') \rangle \right) + \dots,$$
(24)

where H is the Hamiltonian describing the dynamics of the conformal field A. It turns out that the first order terms cancel completely, as one would expect, as they only yield an overall phase for the wave function.

Without further specification of the function $g^{(1)}$, the evolution of the density matrix is found to follow the equation

$$\rho(x, x'; t) = \rho(x, x'; 0) + \frac{m^2 A_0^4}{\hbar^2} \\
\times \left(\int_0^t dt' \int_0^t dt'' g^{(1)} (ct' - ct'' - x + x') g^{(1)} (ct' - ct'' + x - x') \rho(x, x'; 0) \right. \\
\left. -2 \int_0^t dt' \int_0^{t'} dt'' \left(g^{(1)} (ct' - ct'') \right)^2 \rho(x, x'; 0) \right) + \dots$$
(25)

Gravitational Decoherence

Note that although the second order terms in (24) in principle depend on the second order correlations $g^{(2)}$ as well, only the first order correlation function $g^{(1)}$ appears in (25). For a Gaussian correlation, $g^{(1)}(x) = \exp(-x^2/L^2)$, the second order master equation (25) takes the much simpler form

$$\frac{\partial}{\partial t}\rho(\mathbf{x},\mathbf{x}';t) = -\frac{\mathrm{i}}{\hbar} \left[H_0,\rho(\mathbf{x},\mathbf{x}';t)\right] - \sqrt{\frac{\pi}{2}} \frac{m^2 c^3 A_0^4 L}{\hbar^2} \left(1 - \mathrm{e}^{-2\frac{(x-x')^2}{L^2}}\right) \rho(\mathbf{x},\mathbf{x}';t) \,. \tag{26}$$

The main differences compared to the non-propagating fluctuations (22) are in the exponential dependence on spatial separation rather than the sharp cut-off in (22), as well as in that the field amplitude enters only with the fourth power instead of second, due to the fact that all first order terms cancel.

Both (22) and (26) exhibit the typical structure known also from collisional decoherence [36]: a quadratic decay for short distances which goes to zero (exponentially or incontinuously through the cut-off) for longer distances. This structure is also found in collapse models, and is discussed in section 3.1 in context of the Diósi–Penrose model.

2.1.3. Effective Schrödinger equation in a fluctuating spacetime Going beyond the conformal fluctuations in the previous section and considering generic fluctuations of the metric, Göklü and Lämmerzahl [37] derive an effective Schrödinger equation by considering a Klein-Gordon field on a fluctuating spacetime. The scalar field is coupled minimally to gravity, yielding the usual curved spacetime Klein-Gordon equation

$$g^{\mu\nu}\nabla_{\mu}\nabla_{\nu}\phi + \frac{m^2c^2}{\hbar^2}\phi = 0, \qquad (27)$$

where ∇_{μ} denotes the covariant derivative. Using the method of Kiefer and Singh [38], the Klein-Gordon equation is expanded in powers of $1/c^2$, yielding

$$i\hbar\partial_t\psi = -\frac{\hbar^2}{2m}\nabla^2\psi + \frac{m}{2}\left(\delta_{lm}h^{il}_{(0)}h^{jm}_{(0)} - h^{00}_{(0)}\right)\psi + \frac{1}{2}\left\{i\hbar\partial_i, h^{i0}_{(1)} + \delta_{lm}h^{il}_{(0)}h^{m0}_{(1)}\right\}\psi - \frac{i\hbar}{4}\partial_t\left(\delta_{ij}h^{ij}_{(0)} - \frac{1}{2}\delta_{ij}\delta_{lm}h^{il}_{(0)}h^{jm}_{(0)}\right)\psi + \frac{i\hbar}{4}h^{i0}_{(1)}\partial_i\delta_{ij}h^{ij}_{(0)}\psi,$$
(28)

where $\nabla^2 = \partial_i (\sqrt{g^{(3)}} g^{ij} \partial_j \psi) / \sqrt{g^{(3)}}$ is the Laplace–Beltrami operator in the threedimensional spatial hypersurface, and $g^{(3)}$ is the determinant of the spatial part of the metric. A simultaneous transformation of both the wave function and the Hamiltonian according to

$$\psi \to (g^{(3)})^{1/4}\psi, \qquad H \to (g^{(3)})^{1/4}H(g^{(3)})^{-1/4} + \frac{i\hbar}{4}\partial_t \ln g^{(3)}$$
 (29)

turns this into a Hermitian "flat-space form". An effective Schrödinger equation of the form $i\hbar\partial_t\psi = (H_0 + H_p(t))\psi$ is obtained after averaging over the spatial fluctuations, where H_0 corresponds to the usual, unperturbed evolution, and the term $H_p(t)$ encodes the effect of the time-dependent metric fluctuations.

Under the assumption that these fluctuations are isotropic, Breuer $et \ al$ [39] arrive at the master equation

$$\frac{\partial}{\partial t}\rho(t) = -\frac{\mathbf{i}}{\hbar} \left[H_0, \rho(t)\right] - \frac{\tau_c}{8m^2\hbar^2} \left[\mathbf{p}^2, \left[\mathbf{p}^2, \rho(t)\right]\right] \,, \tag{30}$$

ituda of the random r

where τ_c is a free parameter corresponding to the magnitude of the random metric fluctuations. This master equation is quite different from those discussed in the previous section. Although also being in Lindblad form, it describes localization in energy rather than position, implying energy conservation: $\frac{d}{dt}\langle H_0\rangle = 0$.

The origin of these differences remains unexplained, and the large variety of approaches with seemingly different outcomes makes a further investigation of the question how classical spacetime fluctuations should be modeled desirable.

2.2. Quantum spacetime fluctuations

Although there is no unanimously accepted high-energy quantum theory of gravity, it is often assumed that the perturbative quantization of general relativity is applicable at low energies. It is then treated as an effective field theory, the limiting case of the yet unknown full quantum gravity theory. In this regime, the non-renormalizability of said theory does not pose an immediate problem. One must, however, keep in mind that this treatment of gravity in analogy to the known quantum field theories for matter fields is a hypothesis which is, so far, lacking experimental evidence.

Even in a quantum theory of gravity, general relativity must emerge as a classical limit. In this regime, the semi-classical Einstein equations

$$R_{\mu\nu} - \frac{1}{2}g_{\mu\nu}R = \frac{8\pi G}{c^4} \left\langle \Psi \right| \hat{T}_{\mu\nu} \left| \Psi \right\rangle \tag{31}$$

are valid, i.e. quantum matter enters as a source of spacetime curvature as an expectation value of the stress-energy operator. However, since this stress-energy operator is derived from quantized fields, one must account for quantum fluctuations which lead to corrections to (31). A rigorous approach towards modeling these fluctuations and their consequences is provided by stochastic gravity, a review of which is given in [40].

2.2.1. Thermal graviton background If perturbative quantum gravity is assumed to be correct, decoherence effects arise due to background gravitons, in a very similar fashion in which photons are a source for decoherence. Such a decoherence effect due to a thermal spectrum of gravitons has been studied by Blencowe [41].

A scalar field ϕ of mass m on a spacetime with the metric $g_{\mu\nu}$ can be described by the action

$$S = \int d^4x \sqrt{-g} \left(\frac{R}{16\pi G} - \frac{1}{2} g^{\mu\nu} \partial_\mu \phi \partial_\nu \phi - \frac{m^2 c^2}{2\hbar^2} \phi^2 \right) , \qquad (32)$$

where g denotes the metric determinant. The first term is the Einstein-Hilbert action, describing the spacetime dynamics, and the second and third term describe the dynamics of the field evolving on the curved spacetime.§

 \S Note that for a scalar field the covariant derivative is simply the partial derivative and, hence, there is no need for writing a covariant derivative in (32).

In the linear approximation, where $g_{\mu\nu} = \eta_{\mu\nu} + \epsilon h_{\mu\nu}$, the action can then be decomposed [42] into a sum $S = S_{\text{matter}} + S_{\text{grav}} + S_{\text{int}}$ with the flat space Klein-Gordon action

$$S_{\text{matter}} = \int d^4 x \left(-\frac{1}{2} \eta^{\mu\nu} \partial_\mu \phi \partial_\nu \phi - \frac{m^2 c^2}{2\hbar^2} \phi^2 \right) \,, \tag{33}$$

and the graviton action

$$S_{\text{grav}} = \frac{\epsilon^2}{32\pi G} \int d^4x \left(-\frac{1}{2} \eta^{\alpha\beta} \partial_\alpha h^{\mu\nu} \partial_\beta h_{\mu\nu} + \eta^{\alpha\beta} \partial_\nu h^{\mu\nu} \partial_\alpha h_{\mu\beta} - \eta^{\alpha\beta} \partial_\mu h \partial_\nu h^{\mu\nu} + \frac{1}{2} \eta^{\mu\nu} \partial_\mu h \partial_\nu h \right) + \mathcal{O}(\epsilon^3) , \qquad (34)$$

where $h^{\mu\nu} = \eta^{\mu\alpha}\eta^{\nu\beta}h_{\alpha\beta}$ and $h = \eta^{\mu\nu}h_{\mu\nu}$. The interaction term is

$$S_{\rm int} = \int d^4x \left(\frac{\epsilon}{2} T^{\mu\nu} h_{\mu\nu} + \frac{\epsilon^2}{4} U^{\mu\nu\alpha\beta} h_{\mu\nu} h_{\alpha\beta} \right) + \mathcal{O}(\epsilon^3) \,. \tag{35}$$

The stress energy tensor of the scalar field is

$$T_{\mu\nu} = \partial_{\mu}\phi\partial_{\nu}\phi - \frac{1}{2}\eta_{\mu\nu}\eta^{\alpha\beta}\partial_{\alpha}\phi\partial_{\beta}\phi - \frac{m^{2}c^{2}}{2\hbar^{2}}\eta_{\mu\nu}\phi^{2}, \qquad (36)$$

and the tensor U is given by

$$U_{\mu\nu\alpha\beta} = -2\eta_{\nu\alpha}\partial_{\mu}\phi\partial_{\beta}\phi + \eta_{\mu\nu}\partial_{\alpha}\phi\partial_{\beta}\phi + \left(\frac{1}{2}\eta_{\mu\alpha}\eta_{\nu\beta} - \frac{1}{4}\eta_{\mu\nu}\eta_{\alpha\beta}\right)\left(\eta^{\rho\sigma}\partial_{\rho}\phi\partial_{\sigma}\phi + \frac{m^{2}c^{2}}{\hbar^{2}}\phi^{2}\right).$$
(37)

Blencowe [41] now uses the closed time path integral approach [43], from which the evolution equation for the density matrix follows immediately:

$$\rho_{\text{matter}}[\phi, \phi'; t] = \int d\phi_0 \, d\phi'_0 \int_{\phi_0}^{\phi} d\phi^+ \int_{\phi'_0}^{\phi'} d\phi^- \rho_{\text{matter}}[\phi_0, \phi'_0; 0] \\ \times \exp\left(\frac{i}{\hbar} \left(S_{\text{matter}}[\phi^+] - S_{\text{matter}}[\phi^-] + S_{\text{FV}}[\phi^+, \phi^-]\right)\right),$$
(38)

where $S_{\rm FV}[\phi^+, \phi^-]$ is the Feynman–Vernon influence action [44, 43].

In the case where the environment action S_{grav} is quadratic in the fields $h_{\mu\nu}$, the initial density matrix of the environment is Gaussian, and the interaction term S_{int} is bilinear, i.e. both $h_{\mu\nu}$ and ϕ appear linearly, the influence action takes a simpler form [43, 45]. The explicit calculation is straightforward but tedious, following chapter 3 of [43].

Blencowe [41] uses the Born approximation, i.e. the assumption that the environment is approximately constant over the time intervals of interest, and obtains the master equation

$$\frac{\partial}{\partial t}\rho(t) = -\frac{\mathrm{i}}{\hbar} \left[H_{\mathrm{matter}},\rho(t)\right]
-\int_{0}^{t} \mathrm{d}\tau \int \mathrm{d}^{3}r \,\mathrm{d}^{3}r' \left(N(\tau,\mathbf{r}-\mathbf{r}')\left(2\left[T_{\mu\nu},\left[\widetilde{T}^{\mu\nu},\rho(t)\right]\right] - \left[T^{\mu}_{\mu},\left[\widetilde{T}^{\nu}_{\nu},\rho(t)\right]\right]\right)
-\mathrm{i}\,D(\tau,\mathbf{r}-\mathbf{r}')\left(2\left[T_{\mu\nu},\left\{\widetilde{T}^{\mu\nu},\rho(t)\right\}\right] - \left[T^{\mu}_{\mu},\left\{\widetilde{T}^{\nu}_{\nu},\rho(t)\right\}\right]\right)\right),$$
(39)

where we simply write $\rho \equiv \rho_{\text{matter}}$ now, and define $T_{\mu\nu} = T_{\mu\nu}(\tau, \mathbf{r})$, $\tilde{T}_{\mu\nu} = T_{\mu\nu}(-\tau, \mathbf{r}')$. The expressions N and D are the noise and dissipation kernels [41], respectively

$$N(t, \mathbf{r}) = \frac{G}{4\pi^2} \int \mathrm{d}^3k \, \frac{\exp(\mathrm{i}\,\mathbf{k}\cdot\mathbf{r})}{k} \left(1 + 2\,n(k)\right) \, \cos kt \tag{40a}$$

$$D(t, \mathbf{r}) = \frac{G}{4\pi^2} \int \mathrm{d}^3k \, \frac{\exp(\mathrm{i}\,\mathbf{k}\cdot\mathbf{r})}{k} \, \sin kt \,, \tag{40b}$$

where n(k) is the Bose-Einstein occupation number at a given temperature T. These functions are a consequence of the assumption that the background field is described by a thermal distribution of gravitons.

In the nonrelativistic limit, the dominant contribution to the stress-energy tensor comes from the component $T_{00} \approx \frac{m^2 c^2}{2\hbar^2} \phi^2$, and the master equation (39) simplifies further. For decoherence effects, only the noise term proportional to N needs to be considered, while the dissipation term can be neglected.

Blencowe [41] proceeds with the introduction of a basis of coherent states

$$|\alpha\rangle = \exp\left[-\frac{1}{2}\int \mathrm{d}^{3}k \,|\alpha(\mathbf{k})|^{2} + \int \mathrm{d}^{3}k \,\alpha(\mathbf{k}) \,\hat{a}^{\dagger}(\mathbf{k})\right] |0\rangle, \qquad (41)$$

with

$$\alpha(\mathbf{k}) = \varphi_0 R^3 \left(\frac{m^2 c^2 + \hbar^2 k^2}{4\hbar^2}\right)^{1/4} e^{-i\,\mathbf{k}\cdot\mathbf{r}_0 - (kR)^2/2}, \qquad (42)$$

which satisfy

$$\langle \alpha \mid \phi(\mathbf{r}) \mid \alpha \rangle \sim \exp\left(-\frac{(\mathbf{r} - \mathbf{r}_0)^2}{2R^2}\right) \quad \text{and} \quad \langle \alpha \mid \dot{\phi}(\mathbf{r}) \mid \alpha \rangle = 0, \quad (43)$$

and can, therefore, be understood as models for macroscopic material objects in the shape of "Gaussian matter balls". The radius is supposed to be much larger than the Compton wavelength, $R \gg \lambda_c = h/(mc)$. One further assumes that these balls have negligible free spreading. For two such Gaussian matter distributions in a superposition of distinct rest mass energies $E_1 - E_2 = \Delta E$ one then finds a decoherence rate

$$\Gamma_{\rm dec} = \frac{k_B T}{\hbar} \left(\frac{\Delta E}{E_P}\right)^2 \,, \tag{44}$$

 $E_P = \sqrt{\hbar c^5/G}$ being the Planck energy.

2.2.2. Generic perturbations A more generic approach is taken by Anastopoulos and Hu [46] who derive a master equation for gravitational decoherence in a more generic way. Starting from the action (32) for a scalar field interacting with the gravitational field, the usual Arnowitt–Deser–Misner (ADM) formalism [47] is employed in order to perform a 3+1 decomposition of the action and the metric. An expansion to first order in $\kappa = 16\pi G$ yields the perturbations $h_{ij} = \delta_{ij} + \kappa \gamma_{ij}$ for the spatial metric, $N = 1 + \kappa n$ for the lapse, and $N^i = \kappa n^i$ for the shift function. A Legendre transformation then results in the Hamiltonian (in natural units)

$$H = \int d^3x \left[\frac{1}{\kappa} \left(\Pi^{ij} \Pi_{ij} - \frac{1}{2} \Pi^2 + \kappa^2 V_g \right) + \frac{1}{2} \left(\pi^2 + \partial_i \phi \partial^i \phi + m^2 \phi^2 \right) - \frac{\kappa}{2} \gamma^{ij} \partial_i \phi \partial_j \phi + \frac{\kappa \gamma}{4} (\partial_i \phi \partial^i \phi + m^2 \phi^2 - \pi^2) + n\mathcal{H} + n_i \mathcal{P}^i \right],$$
(45)

with $\gamma = \delta^{ij} \gamma_{ij}$. Π^{ij} are the conjugate momenta of the spatial perturbation γ^{ij} , π is the conjugate momentum of the scalar field ϕ , and the self-gravitational potential V_g (containing the second order terms in γ) is given by

$$V_g = -\frac{1}{2}\partial_k\gamma_{ij}\partial^i\gamma^{kj} - \frac{1}{2}\partial_k\gamma\partial^k\gamma + \partial_i\gamma\partial_k\gamma^{ik} + \frac{1}{4}\partial_k\gamma_{ij}\partial^k\gamma^{ij}.$$
 (46)

The Hamiltonian and momentum constraints \mathcal{H} and \mathcal{P}^i generate gauge transformations which correspond to temporal and spatial reparametrizations of the free fields. It is important to note that such reparametrizations in general relativity are pure gauge. As pointed out by Anastopoulos and Hu: "Any postulate of dynamical or stochastic fluctuations that correspond to space and time reparametrizations conflicts with the fundamental symmetries of GR" [46].

For the quantization of the Hamiltonian (45), Anastopoulos and Hu [46] fix a gauge in which both the longitudinal part of the metric fluctuation γ^{ij} and the transverse trace of the gravitational conjugate momentum Π vanish. This choice ensures that the Lorentz frame introduced by the 3+1 decomposition remains invariant under reparametrizations of space and time.

Writing the Hamiltonian operator as $\hat{H} = \hat{H}_0 + \kappa \hat{V}_g + \hat{H}_{int}$, the interaction Hamiltonian \hat{H}_{int} is expressed in terms of creation and annihilation operators $\hat{b}^{\dagger}_{\pm}(\mathbf{k})$ and $\hat{b}_{\pm}(\mathbf{k})$ corresponding to the transverse traceless metric perturbations and their conjugate momenta. The index \pm denotes the two polarizations, and with the operator $\hat{J}_{\pm}(\mathbf{k})$, derived from the normal ordered stress operator, the interaction Hamiltonian can be written as

$$\hat{H}_{\text{int}} = \sum_{\sigma \in \pm} \int \frac{\mathrm{d}^3 k}{(2\pi)^3} \left[|\mathbf{k}| \, \hat{b}^{\dagger}_{\sigma}(\mathbf{k}) \hat{b}_{\sigma}(\mathbf{k}) - \frac{1}{2} \sqrt{\frac{\kappa}{|\mathbf{k}|}} \left(\hat{b}_{\sigma}(\mathbf{k}) \hat{J}^{\dagger}_{\sigma}(\mathbf{k}) + \hat{b}^{\dagger}_{\sigma}(\mathbf{k}) \hat{J}_{\sigma}(\mathbf{k}) \right) \right] \,. \tag{47}$$

This Hamiltonian models the gravitational environment which consists of harmonic oscillators coupled to the matter degrees of freedom encoded in \hat{J} .

The next step is to derive a master equation for the matter degrees of freedom. For this purpose, one starts with the assumption that the initial state factorizes in matter and gravitational degrees of freedom, i.e. $\rho = \rho_{\text{mat}} \otimes \rho_{\text{grav}}$, and traces out the gravitational part. The only issue is the choice of the initial state ρ_{grav} of the gravitational field. Assuming that it is Gaussian at a "temperature" Θ , the Hamiltonian (47) straightforwardly yields the master equation ¶

$$\frac{\partial}{\partial t}\rho_{\rm mat}(t) = -i\left[\hat{H}_0 + \frac{\kappa}{2}\hat{V}, \rho_{\rm mat}\right]$$

|| See [46] for the precise definition of \hat{J} and \hat{b} .

¶ Note that the derivation of this master equation to second order in $\sqrt{\kappa}$ does not make use of the Born–Markov approximation.

Gravitational Decoherence

$$-\sum_{\sigma \in \pm} \int \frac{\mathrm{d}^{3}k}{(2\pi)^{3}} \frac{\kappa}{4|\mathbf{k}|} \left[\left(\coth\left(\frac{|\mathbf{k}|}{2\Theta}\right) + 1 \right) \left[\hat{J}_{\sigma}^{\dagger}(\mathbf{k}), \left[\hat{K}_{\sigma}(\mathbf{k}), \rho(t) \right] \right] \right] + \left(\coth\left(\frac{|\mathbf{k}|}{2\Theta}\right) - 1 \right) \left[\hat{J}_{\sigma}(\mathbf{k}), \left[\hat{K}_{\sigma}^{\dagger}(\mathbf{k}), \rho(t) \right] \right] \right]$$
(48*a*)

with

$$\hat{K}_{\sigma}(\mathbf{k}) = \int_{0}^{\infty} \mathrm{d}s \,\mathrm{e}^{-\mathrm{i}|\mathbf{k}|s} \mathrm{e}^{-\mathrm{i}\hat{H}_{0}s} \hat{J}_{\sigma}(\mathbf{k}) \mathrm{e}^{\mathrm{i}\hat{H}_{0}s} \,.$$

$$(48b)$$

In the limit of a nonrelativistic particle, restricted to one dimension, the momentum space master equation simplifies considerably, yielding

$$\frac{\partial}{\partial t}\rho(\mathbf{p},\mathbf{p}';t) = -\frac{\mathrm{i}}{2\,\hbar\,m}\left[p^2,\rho(\mathbf{p},\mathbf{p}';t)\right] - \frac{4\pi\,G\,\Theta}{9\,\hbar^2\,m^2\,c^5}\left[p^2,\left[p^2,\rho(\mathbf{p},\mathbf{p}';t)\right]\right]\,,\tag{49}$$

and the decoherence rate

$$\Gamma_{\rm dec} = \frac{G\Theta}{\hbar^2 m^2 c^5} \langle p \rangle^2 \Delta p^2 \tag{50}$$

if $\langle p \rangle$ is the momentum expectation value and Δp the variance of a momentum superposition state. As the decoherence described by (49) is due to a difference in *energy* rather than in momentum, a state peaked at +p and -p will not decohere. This is a feature this model has in common with the previously discussed decoherence effect due to thermal gravitons [41] discussed above in section 2.2.1.

A remark is necessary concerning the interpretation of the "temperature" variable Θ . As Anastopoulos and Hu [46] point out, an initial state ρ_{grav} as a thermal state at a certain temperature makes sense in the case where metric perturbations are of a *classical* nature. For fluctuations that are *inherently quantum*, the natural choice for the initial state is the quantum field theoretical vacuum state of the graviton field. As gravitons interact only weakly, their thermalization cannot be taken for granted. In this sense, the parameter Θ is to be understood as interpolating between classical and quantum gravitational fluctuations. The previously discussed model of thermal gravitons [41] should also be seen in this light.

A very similar approach, without the restriction to a scalar field, is followed by Oniga and Wang [48, 49] who arrive at a rather generic master equation for gravitational decoherence under minimal assumptions which has a structure similar to (2.2.2).

2.2.3. Minimal length The previously described decoherence effect due to a thermal graviton background only requires the assumption that gravity is quantized at low energies in a similar fashion as, e.g., electrodynamics. Another feature which is usually associated with quantum theories of gravity at high energies is the existence of a minimal length scale, which is often assumed to be of the order of the Planck length $\ell_{\rm Pl}$.

Power and Percival [33] make use of this idea in order to arrive at a master equation which depends only on this minimal length scale as a free parameter. For the discussion of this idea, we recall the master equation (26) in section 2.1.2, which resulted from considering conformal fluctuations of the (classical) spacetime metric. There, the master equation contained two free parameters: the amplitude A_0 of the conformal fluctuation,

Gravitational Decoherence

as well as the coherence length scale L. Relating this to quantized gravity, Power and Percival [33] estimate the amplitude of conformal quantum fluctuations to be of the order of $A_0 \approx (\ell_{\rm Pl}/l_{\rm cut-off})^2$, where the proposed minimal length scale is used as a cut-off length $l_{\rm cut-off}$. This result can be directly inserted into the master equation (26) obtained for the classical conformal fluctuation. The coherence length scale is assumed to coincide with the cut-off length, i. e. $L = l_{\rm cut-off}$, resulting in a prediction for the evolution of the density matrix which has only the cut-off scale as a free parameter:

$$\frac{\partial}{\partial t}\rho(\mathbf{x},\mathbf{x}';t) = -\frac{\mathrm{i}}{\hbar} \left[H_0,\rho(\mathbf{x},\mathbf{x}';t)\right] - \sqrt{\frac{\pi}{2}} \frac{m^2\hbar^2 G^4}{c^9 l_{\text{cut-off}}^7} \left(1 - \mathrm{e}^{-2\frac{(x-x')^2}{l_{\text{cut-off}}^2}}\right) \rho(\mathbf{x},\mathbf{x}';t) \,. \tag{51}$$

Note that, as in section 2.1.2, it is assumed that fluctuations travel at the speed of light.

Wang *et al* [50] provide a discussion which is based on the model by Power and Percival [33]. They add further degrees of freedom to the gravitational field allowing also for shearing actions. For this purpose, the metric is rewritten as $g_{\mu\nu} = (1+A)^2 \gamma_{\mu\nu}$, where A is the conformal field known from the previous discussion, but the rescaled metric $\gamma_{\mu\nu}$ takes the place of the Minkowski metric. Its time components are the same as in the Minkowski case, $\gamma_{00} = 1$ and $\gamma_{0i} = 0$, while the spatial components γ_{ij} vary and are normalized such that $\det(\gamma_{ij}) = -1$. In contrast to the conformal approach, this ansatz does not violate Einstein's equations.

Instead of the free Klein-Gordon equation $\eta^{\mu\nu}\partial_{\mu}\partial_{\nu}A(x) = 0$, the dynamics of the field A then are given by the conformal constraint Hamiltonian [50]

$$H^{(\rm CF)} = -\frac{3c^2}{8\pi G} \left(\dot{A}^2 + c^2 \gamma^{ij} \frac{\partial}{\partial x_i} A \frac{\partial}{\partial x_j} A \right) \,. \tag{52}$$

This is almost the Hamiltonian for a massless, scalar Klein-Gordon field, however belonging to negative energy due to the positive sign between temporal and spatial derivatives. The total Hamiltonian can be re-written as a sum of the conformal Hamiltonian and a second term describing gravitational waves: $H = H^{(CF)} + H^{(GW)}$.

Assuming that quantum fluctuations lead to a gravitational background for times shorter than $\tau_0 = \lambda t_{\rm Pl}$, i. e. a multiple of the Planck time defined by the factor λ , and modeling these fluctuations as standing waves in a box of unit volume with a frequency cut-off at $\omega_0 = 2\pi/\tau_0$, one can calculate the zero point energy

$$H_0^{(\text{GW})} \approx \frac{2\pi^2 c^2 m_{\text{Pl}}}{\lambda^4 \ell_{\text{Pl}}^3},$$
(53)

which would be of the order of $\lambda^{-4} \times 10^{98}$ kg/m³. However, Wang *et al* [50] point out that the conformal field gives a contribution of equal magnitude and opposite sign: $H^{(\text{CF})} = -H_0^{(\text{GW})}$. The interpretation they provide is that every pair of zero point energy gravitons of opposite helicities carries the energy $\hbar\omega/2$ but is accompanied by a quantum of the conformal field with *negative* energy $-\hbar\omega$. The gravitational wave terms, being of first post-Newtonian order, give no contribution in the Newtonian limit. The conformal field, on the other hand, leads to a Newtonian gravitational potential and decoherence effects as discussed in section 2.1.2. From (52) together with the condition $H^{(\text{CF})} = -H_0^{(\text{GW})}$ one can show that the amplitude of the conformal field is $A_0^2 = 4\pi/(3\lambda^2)$. This can be inserted into (26) and yields a lower bound on λ by comparison with experimental observations, where the dependence on experimental parameters is $\lambda \sim m^{2/3} (\delta \rho / \rho_0)^{-1}$. From atom interferometric data [51] one concludes that $\lambda \geq 7600$. Under the assumption that this model is an accurate description of reality, this implies that the scale where quantum gravitational effects become relevant cannot be lower than about four orders of magnitude above the Planck scale, and refined experiments would further increase this value.

Garay [52, 53] provides a model for decoherence based on the assumption of a spacetime foam [54]. The spacetime foam leaves an imprint also at low energies as a consequence of assumed nonlocal interactions. Specifically, given a basis $\{h_i[t]\}_i$ of the local, gauge-invariant interactions, the dominant contribution in the Euclidean action comes from bilocal interactions

$$I_{\rm int} = \frac{1}{2} \int dt \, dt' \, c^{ij}(t-t') \, h_i[t] \, h_j[t'] \,. \tag{54}$$

Garay considers three separate scales which are assumed to separate: the Planck scale $\ell_{\rm Pl}$, the scale l of low energy physics, and the scale r beyond which gravitational fluctuations are of relevance. The parameter $\epsilon \sim e^{-(r/\ell_{\rm Pl})^2} (r/\ell_{\rm Pl})^4 (l/\ell_{\rm Pl})^{-2}$ is introduced, which suppresses both gravitational fluctuations $(r \gg \ell_{\rm Pl})$ and low energy scales $(l \gg \ell_{\rm Pl})$ far from the Planck scale. It can be shown that the interactions for a scalar field ϕ are of the order $h_i[t] \sim \phi^2/\ell_{\rm Pl}^2$ and the bilocal interaction term is of the order $I_{\rm int} \sim e^{-(r/\ell_{\rm Pl})^2} (r/\ell_{\rm Pl})^4$, while further terms are of higher order in both ϵ and r/l.

Assuming that the correlation function $c^{ij}(t-t')$ is Gaussian, after a Wick rotation back into the Lorentz frame and making use of the influence functional formalism (cf. (38)) the resulting master equation at lowest order is

$$\frac{\partial}{\partial t}\rho(t) = -\frac{\mathrm{i}}{\hbar} \left[H_0,\rho\right] - \int_0^\infty \mathrm{d}\tau \, c^{ij}(\tau) \, \left[h_i, \left[h_j,\rho\right]\right] \tag{55}$$

where the second term describes the effect of the fluctuations while the first term is the usual Hamiltonian evolution. Dissipation terms are absent from (55), as they are of higher order in r/l. The correlation functions c^{ij} are not predicted by this model but are associated with a thermal bath, quite similar to the model by Blencowe [41]. The reasoning behind this approach comes from the comparison of the master equation (55) with the equations for quantum Brownian motion [55].

A critical view at this result [56] raises the question whether the performed separation of scales is physically reasonable, given that in general relativity the local structure of spacetime is strongly linked with the locality of interactions. It is also pointed out that the nonlocal interaction (54) is, at the level of the resulting master equation, indistinguishable from the effect of local additional phenomenological fields. The Born–Markov approximation may not be a good assumption for the dynamics induced by a spacetime foam. If the approximation does not hold, the resulting master equation could be of a vastly different shape. Similar criticisms [56] apply to a related

derivation [57] of decoherence in the flavor sector for neutrinos due to a spacetime foam motivated by string theory.

2.2.4. *Time fluctuations* A different approach to introduce quantum fluctuations of spacetime in quantum mechanics is through the effect of such fluctuations on the time parameter in the Schrödinger equation.

Intuitively speaking, a free quantum system evolves along a geodesic with the proper time τ given by the metric, for instance

$$|\psi\rangle \to e^{i\omega\tau} |\psi\rangle, \tag{56}$$

with some frequency ω . If the metric is considered a quantum object itself, which is in a quantum state $|g_{\mu\nu}\rangle$, then one expects the quantum state of the system to become entangled with the state of the metric: $|\psi\rangle \otimes |g_{\mu\nu}\rangle$. Specifically, if one allows for linear superposition states of metric fluctuations, i. e. $\frac{1}{2}(|g_{\mu\nu}\rangle + |g'_{\mu\nu}\rangle)$, then the system state $|\psi\rangle$ becomes entangled to the metric state and evolves according to

$$|\psi\rangle \otimes \frac{1}{2} \left(|g_{\mu\nu}\rangle + |g'_{\mu\nu}\rangle\right) \to \frac{1}{2} \left(e^{i\omega\tau} |\psi\rangle \otimes |g_{\mu\nu}\rangle + e^{i\omega\tau'} |\psi\rangle \otimes |g'_{\mu\nu}\rangle\right) , \quad (57)$$

where τ and τ' are the proper times corresponding to the metrics $g_{\mu\nu}$ and $g'_{\mu\nu}$, respectively. The fluctuations of the metric propagate to the system state and lead to a loss of coherence. Kok and Yortsever discuss decoherence on the grounds of these considerations [58].

A slightly more sophisticated model by Milburn [59] treats the dynamics of quantum states as probabilistic, finite jumps rather than a continuous unitary evolution. This is put on more rigorous mathematical grounds, motivated from time as a statistical variable, by Bonifacio [60], however with essentially identical physical implications.

Instead of the continuous evolution of the density matrix,

$$\rho(t+\tau) = \exp\left(-\frac{\mathrm{i}}{\hbar} H \tau\right) \rho(t) \exp\left(\frac{\mathrm{i}}{\hbar} H \tau\right) \,, \tag{58}$$

one assumes that a change of the system occurs with a certain probability, specifically that there is a probability $p_n(\tau_0, \tau)$ that *n* random time shifts by τ_0 lead to a total time shift of τ . The density matrix then satisfies an equation

$$\rho(t+\tau) = \sum_{n=0}^{\infty} p_n(\tau_0, \tau) \, \exp\left(-\frac{\mathrm{i}}{\hbar} \, H \, n \, \tau_0\right) \rho(t) \exp\left(\frac{\mathrm{i}}{\hbar} \, H \, n \, \tau_0\right) \,. \tag{59}$$

In the limit $\tau_0 \to 0$ this yields the continuous equation (58). For the specific choice of a Poisson distribution,

$$p_n(\tau_0, \tau) = \frac{\tau^n}{\tau_0^n \, n!} \,\mathrm{e}^{-\tau/\tau_0} \,, \tag{60}$$

the master equation takes the form

$$\frac{\partial}{\partial t}\rho(t) = \frac{1}{\tau_0} \left(\exp\left(-\frac{\mathrm{i}}{\hbar} H \tau_0\right) \rho(t) \exp\left(\frac{\mathrm{i}}{\hbar} H \tau_0\right) - \rho(t) \right), \tag{61}$$

or after expansion in the, presumably small, parameter τ_0 which indicates the time scale where deviations from continuous evolution become apparent:

$$\frac{\partial}{\partial t}\rho(t) = -\frac{\mathrm{i}}{\hbar} \left[H,\rho(t)\right] - \frac{\tau_0}{2\hbar^2} \left[H,\left[H,\rho(t)\right]\right] \,. \tag{62}$$

As pointed out by Anastopoulos and Hu [61], these considerations are in fact more general, and the semigroup equation (59) follows quite generally for any physical quantity and its corresponding generator. For example, the above results remain valid if time is replaced by position and the time translation generator H by the momentum p. They criticize that in Milburn's formalism there is no requirement to relate the incremental random time shifts with gravity or the Planck scale. This association is a mere intuitive hypothesis, and considering that many quantum effects are only poorly modeled by classical statistics, not necessarily a very compelling one.

Note also that, as Milburn points out himself [59], Lorentz invariance requires the random time shifts to be accompanied by a similar spatial structure, because the temporal jumps will appear as position fluctuations in another reference frame.

2.3. Decoherence due to gravitational time dilation

Decoherence can occur even in a classical and completely static spacetime, where the gravitational field itself exhibits no quantum or stochastic fluctuations. Instead, this type of decoherence effect stems from a coupling of internal and external degrees of freedom in a complex quantum system, which can be triggered by gravity.

This has been discussed as a consequence of time dilation in a quite intuitive fashion by Pikovski *et al* [20], based on earlier work by the same authors [62] on the interference of clocks; their results have since been the starting point of an ongoing controversy [63, 64, 65, 66, 67, 68, 69, 70, 71, 72, 73].

A "clock" in this context refers to any system which can be modeled as pointlike (i. e. with a single center-of-mass coordinate) and contains some oscillatory internal degrees of freedom. Quantum mechanically, such a "clock" is typically described by a product state

$$|\Psi\rangle = |\psi_{\rm cm}\rangle \otimes |\Psi_{\rm clock}\rangle, \tag{63}$$

with $\psi_{\rm cm}$ the center-of-mass position wave function and $\Psi_{\rm clock}$ the internal clock state. In nonrelativistic settings, the clock undergoes a dynamical evolution described by a Hamiltonian $H = H_{\rm cm} + H_{\rm clock}$. Hence, the evolution strictly separates into the center-of-mass motion and the internal dynamics ("ticking") of the clock:

$$|\Psi(t)\rangle = e^{-\frac{i}{\hbar}Ht} |\Psi\rangle_0 = e^{-\frac{i}{\hbar}H_{\rm cm}t} |\psi_{\rm cm}\rangle_0 \otimes e^{-\frac{i}{\hbar}H_{\rm clock}t} |\Psi_{\rm clock}\rangle_0.$$
(64)

In relativistic contexts, however, this strict separation does no longer hold.

The proper time measured by a clock between two spacetime events A and B depends on the spacetime trajectory the clock takes from A to B. In special relativity, this is of course known as the "twin paradox", the reason for which is special relativistic time dilation, i.e. the fact that moving clocks tick at a slower rate. On the other hand,

general relativity predicts that clocks in a stronger gravitational field tick at a slower rate (and, therefore, clocks on Earth tick the faster the higher their altitude).

Two clocks moving from A to B on different spacetime trajectories will, consequently, show a proper time difference $\Delta \tau$. For a quantum superposition of two clocks—assuming that the linear superposition principle for wave functions remains valid in this relativistic context—one expects constructive interference if $\Delta \tau$ is an integer multiple of both the period E/\hbar of the clocks center-of-mass matter-wave and the oscillation period of the clock. The first condition is simply the equivalent to the nonrelativistic requirement for the optical path length to be an integer multiple of the wavelength. The second condition poses an additional coherence criterion which is absent in the nonrelativistic limit.

If the considered system is not a "clock", containing one single internal degree of freedom, but rather a complex system (such as a large molecule) with many degrees of freedom of different proper frequencies, coherent superpositions become increasingly unlikely for proper time differences $\Delta \tau \neq 0$, leading to an effective decoherence of wave functions due to the time dilation effect [20].

Attempting to put this intuitive expectation on rigorous mathematical grounds, one faces several difficulties. Quantum mechanics, strictly speaking, makes no sense in a relativistic context as any interacting quantum theory is facing the problem that it must account for particle creation and annihilation [74]. In principle, relativistic effects should be described in the language of quantum field theory—and in the gravitational case in terms of quantum field theory on curved spacetime [18, 19].

In situations where the particle number is approximately conserved, the single particle Schrödinger equation can, nonetheless, be obtained from a relativistic Klein-Gordon field as the $c \to \infty$ limit, either by making use of the Foldy-Wouthuysen transformation [75], or by a WKB like expansion [38, 76]. Extending this derivation to the next order in $1/c^2$ one obtains an $O(1/c^2)$ -Hamiltonian for a single particle which includes relativistic corrections. In the special relativistic case, these corrections are well known, for instance from the theory of the hydrogen atom. In a homogeneous gravitation potential the same method yields the Hamiltonian⁺

$$H_{\text{one-particle}} = \frac{p^2}{2m} + \frac{p^4}{8m^3c^2} - g\,m\,z + \frac{g}{4mc^2}\left\{z, p^2\right\}\,.$$
(65)

In nonrelativistic physics one would assume the same single particle Hamiltonian to also describe the center of mass of a system with internal degrees of freedom. In the context of relativity, however, the internal energy contributes to the (inertial and gravitational) mass of the system. The rest mass energy mc^2 must then be replaced by the full relativistic energy $mc^2 + H_{\text{intern}}$ of the system, which contains the Hamiltonian for the internal degrees of freedom. The full Hamiltonian to $O(1/c^2)$ reads:

$$H = \frac{p^2 c^2}{2(mc^2 + H_{\text{intern}})} + \frac{p^4 c^4}{8(mc^2 + H_{\text{intern}})^3}$$

⁺ The Hamiltonian given here is the one following for a Rindler spacetime, while Lämmerzahl [76] derives the Hamiltonian in a Schwarzschild spacetime. The differences are discussed in [71].

$$-\frac{g}{c^2}(mc^2 + H_{\text{intern}}) + \frac{g}{4mc^2} \{z, p^2\} + H_{\text{intern}}$$
$$= \frac{p^2}{2m} - g\,m\,z + H_{\text{intern}} + \frac{p^4}{8m^3c^2} + \frac{g}{4mc^2} \{z, p^2\} - \frac{H_{\text{intern}}}{mc^2} \left(\frac{p^2}{2m} + g\,m\,z\right) \,. \tag{66}$$

The nonrelativistic part of this Hamiltonian strictly separates into center of mass and internal motion, as expected. The relativistic $O(1/c^2)$ corrections, on the other hand, contain a special relativistic coupling of H_{intern} to the center-of-mass energy, as well as a gravitational coupling term $-g z H_{\text{intern}}/c^2$. This coupling is the starting point for the derivation of the gravitationally induced decoherence effect [20].

A remark is at order concerning the meaning of the Hamiltonian (66): It is well known that the notion of center-of-mass coordinates is ambiguous in relativistic contexts [77]. Specifically, a definition of center-of-mass coordinates which satisfy canonical commutation relations, and at the same time transform covariantly under Lorentz transformations does not exist. One consequence is that any choice of centerof-mass coordinates that allows for a straightforward canonical quantization is frame dependent. Such a definition of center-of-mass coordinates based on the form of the Poincaré group generators has been introduced by Krajcik and Foldy [78] for Minkowski space, and recently has been extended to the situation of an accelerated observer (or a homogeneous gravitational field) [71]. The result is indeed the Hamiltonian (66).

Once the existence of the coupling term $H_{\rm coup} = -g z H_{\rm intern}/c^2$ between center of mass and the internal degrees of freedom is accepted, the quantum mechanical description is quite straightforward, and follows the usual prescription of open quantum systems. The full Hamiltonian can then be written as $H = H_{\rm cm} + H_{\rm intern} + H_{\rm coup}$, where the center of mass plays the role of the system and the internal degrees of freedom act as a bath. In the Born–Markov approximation, after tracing out the bath degrees of freedom, the master equation for the system is obtained as [20]

$$\frac{\partial}{\partial t}\rho_{\rm cm}(t) = -\frac{\mathrm{i}}{\hbar} \left[H_{\rm cm} + \frac{g}{c^2} z \left\langle H_{\rm coup} \right\rangle, \rho_{\rm cm}(t) \right] - \frac{\Delta E^2}{\hbar^2 c^4} \\ \times \int_0^t \mathrm{d}t' \left[\left(g \, z + \frac{p^2}{2m^2} \right), \left[\mathrm{e}^{-\frac{\mathrm{i}}{\hbar} H_{\rm cm} t'} \left(g \, z + \frac{p^2}{2m^2} \right) \, \mathrm{e}^{\frac{\mathrm{i}}{\hbar} H_{\rm cm} t'}, \rho_{\rm cm}(t) \right] \right], (67)$$

with $\Delta E^2 = \langle H_{\text{coup}}^2 \rangle - \langle H_{\text{coup}} \rangle^2$. Assuming that the center-of-mass motion is slow compared to decoherence, momentum terms can be neglected, and only the linear term in t is considered, which leads to

$$\frac{\partial}{\partial t}\rho_{\rm cm}(t) = -\frac{\mathrm{i}}{\hbar} \left[H_{\rm cm} + \frac{g}{c^2} z \langle H_{\rm coup} \rangle, \rho_{\rm cm}(t) \right] - \frac{\Delta E^2 g^2}{\hbar^2 c^4} t \left[z, \left[z, \rho_{\rm cm}(t) \right] \right], \tag{68}$$

and a decoherence rate

$$\Gamma_{\rm dec} = \frac{\Delta E \, g \, \Delta z}{\sqrt{2} \, \hbar \, c^2} \,, \tag{69}$$

which depends on both the variance of the internal energy and the variance in the width of the center-of-mass wave-function.

It is also interesting to assess to which extent this gravitationally induced decoherence effect can be detected in the lab. Equation (69) tells that one needs to

prepare a system with a large energy spread, and largely delocalized in space. A vacuum chamber is needed in order to decrease decoherence from gas particles, and a refrigerator in order to decrease thermal decoherence. An analysis performed in [68] shows that current technology cannot sufficiently suppress competing decoherence effects in order to allow for a detection of the time dilation induced decoherence; new schemes must be designed.

In addition to the question whether or not this decoherence effect is observable in practice, its compatibility with the equivalence principle has been questioned [64, 63, 65]. In the case of a free falling system, there is no contradiction, as long as one accepts that an accelerated observer can see effects which are absent in inertial frames—a fact that is well known, for instance, in the case of the Unruh effect [79, 80, 81]. The equivalence principle simply states that for a free falling particle both an accelerated observer (in empty space) and an observer at constant altitude on Earth ("earthbound") agree on their observations. It does not require these observations to be identical with the expectation in an inertial frame.

The requirement of a stationary system (in order to disregard the momentum terms in (67)), however, implies that not only the observer is earthbound but the system is as well. Then the full Hamiltonian must also include some external potential which keeps the system from falling and which, for a consistent description, must *include* relativistic corrections up to $O(1/c^2)$. A-priori, there is no reason to exclude the possibility that these correction terms contribute to the coupling of internal degrees of freedom and the center of mass, and thus can overshadow or even cancel the gravitational coupling. Whether or not the decoherence effect also exists in a system that is held at constant height by an external potential is still subject of a debate [71, 72, 73].

3. Spontaneous wave function collapse and the role of gravity

In the previous section we discussed decoherence effects induced by gravity, within the standard framework of quantum physics. In this section, we will review proposals which attempt to modify quantum theory, motivated from gravity.

The reason for such modifications is that, despite the success of Quantum Mechanics in explaining the structure of matter, the debate about its meaning and limits of validity never ceased. The main reason of concern lies in a straightforward consequence of the linearity of the Schrödinger equation, which is the superposition principle. It implies that superpositions of matter states are possible, and this is what one actually observes in experiments with atoms and molecules [82, 83, 84]. More than this, multi-particle superpositions, namely entangled states, show how different quantum physics can be from classical physics, in the form of non-locality [85, 86, 87]. They are also the basis for the new quantum technologies [88].

Quantum Mechanics contains no inner limit of validity, and it seems natural, until experiments prove the contrary, to extend it also to the description of the macroscopic domain of ordinary objects. In the end, macroscopic objects are made of atoms, which are quantum. But this poses a problem, first addressed by Schrödinger himself [89]: situations can occur, according to the Schrödinger equation, where *macroscopic* quantum superpositions of matter can be created, in plain contradiction with our daily experience. Therefore, a straightforward application of the Schrödinger equation to macroscopic object seems inconsistent.

The standard formulation of Quantum Mechanics provides a tentative resolution of the conflict by assuming the collapse of the wave function at the end of a measurement [90]. But clearly this is a phenomenological recipe, which cannot be accepted at the fundamental level. Weinberg among many, wrote: "The Copenhagen interpretation assumes a mysterious division between the microscopic world governed by quantum mechanics and a macroscopic world of apparatus and observers that obeys classical physics. During measurement the state vector of the microscopic system collapses in a probabilistic way to one of a number of classical states, in a way that is unexplained, and cannot be described by the time-dependent Schrödinger equation." [91]

To cope with this problem, the scientific community provided alternative formulations and theories: Bohmian mechanics [92, 93, 94, 95, 96, 97], many worlds [98, 99, 100] and models of spontaneous wave function collapse (collapse models) [101, 102, 103, 104] are among the most popular. Here we will review the third option, because of its potential relation to gravity.

The fundamental idea behind collapse models is that the Schrödinger equation is only approximately valid. It should be supplemented with additional *stochastic* and *nonlinear* terms, which account in a dynamical way for the collapse of the wave function in measurement-like situations, instead of assuming it phenomenologically as the Copenhagen interpretation does. Here a first problem arises: how can one consistently modify such a fundamental equation as the Schrödinger equation, without running into theoretical and experimental inconsistencies?

After decades of research [105, 106, 107, 108, 109], the situation seems now rather clear [108]. If one asks for i) norm conservation and ii) no superluminal signaling, then the structure of the modified Schrödinger equation is fixed, as we will now show. But first a comment about these two requirements. Norm conservation is needed if one wants to interpret the wave function in a reasonable way as objectively describing the state of a quantum systems. One consequence of its violation would be that mass or number of particles spontaneously decays in time. The second requirement, that no signal should be able to be sent faster than the speed of light, is needed for consistency with special relativity. A collapse that does not meet this requirement would allow for *instantaneous* communication at *arbitrarily large* distances [105], which is in contradiction to relativistic causality. Note that this instantaneous signaling has nothing to do with the fact that the Schrödinger equation is a *nonrelativistic* equation; it is only due to the nonlocality of the collapse.

The requirement of no superluminal signaling has a precise formulation [105]: whatever the evolution for the state vector, the dynamics for the density matrix must be *linear*. This basically implies that the master equation is of the Lindblad type [15, 14, 36], with possibly negative coefficients [109]. By adding complete positivity, one has the true Lindblad structure with positive coefficients.

This also implies that when adding nonlinear terms to the Schrödinger equation, one has to include appropriate stochastic terms, which wash away any nonlinearity at the density matrix level. Nonlinear and deterministic evolutions for the state vector unavoidably imply a nonlinear evolution for the density matrix. In other words, nonlinearity and stochasticity have to go hand in hand to have a consistent dynamics for the wave function. This is the basic reason why the structure of consistent nonlinear modifications of the Schrödinger equation is uniquely identified.

Let us then consider the modified Schrödinger equation, written in the Itô language [110]:

$$\mathrm{d}\phi_t = \left[-\mathrm{i}\hat{H}_0\,\mathrm{d}t + \sqrt{\lambda}\,\hat{A}\,\mathrm{d}W_t + \hat{O}\right]\phi_t,\tag{70}$$

with $\hbar = 1$. The term $\sqrt{\lambda} \hat{A} dW_t$ encodes the coupling of the quantum system (represented by the wave function ϕ_t) with the noise W_t (a Wiener process being the first and easiest choice), via the self-adjoint operator \hat{A} , on whose eigenstates the state will eventually collapse. The coupling constant λ sets the strength of the collapse mechanism. Note the absence of the imaginary unit in front of this term, which makes the dynamics non standard. This is necessary, in order to arrive at a nonlinear equation of the collapse type. The operator \hat{O} will be fixed by the requirement of no superluminal signaling.

The norm of ϕ_t is not conserved. Using Itô calculus, one can easily write down the equation for the normalized vector $\psi_t = \phi_t / \|\phi_t\|$:

$$d\psi_t = \left[-i\hat{H}_0 dt + \sqrt{\lambda} (\hat{A} - \langle \hat{A} \rangle_t) dW_t + \lambda \left(\frac{3}{2} \langle \hat{A} \rangle_t^2 - \frac{1}{2} \langle \hat{A}^2 \rangle_t - \hat{A} \langle \hat{A} \rangle_t \right) dt + \left(\hat{O} - \frac{1}{2} \langle (\hat{O}^\dagger + \hat{O}) \rangle_t \right) dt \right] \psi_t,$$
(71)

with $\langle \hat{A} \rangle_t = \langle \psi_t | \hat{A} | \psi_t \rangle$. As expected, the normalized vector evolves according to a nonlinear stochastic dynamics. The stochastic ensemble of pure states $\rho_t^W = |\psi_t\rangle\langle\psi_t|$ obeys the following dynamics:

$$d\rho_t^W = -i[H, \rho_t^W] +\lambda \left(4\langle \hat{A} \rangle_t^2 \rho_t^W - \langle \hat{A}^2 \rangle_t \rho_t^W - 2\hat{A}\langle \hat{A} \rangle_t \rho_t^W - 2\rho_t^W \hat{A}\langle \hat{A} \rangle_t - \hat{A}\rho_t^W \hat{A} \right) dt + \left(\hat{O}^{\dagger} \rho_t^W + \rho_t^W \hat{O} - \langle (\hat{O}^{\dagger} + \hat{O}) \rangle_t \rho_t^W \right) dt + (\text{extra terms}) dW_t.$$
(72)

When taking the expectation value to compute the dynamics for the density matrix $\rho_t = \mathbb{E}[\rho_t^W]$, the 'extra terms' average to 0, while the remaining terms generate a nonlinear evolution for the ensemble, as already stated. This leads to superluminal signaling, which can be avoided by choosing $\hat{O} = -(\lambda/2)\hat{A}^2 + 2\lambda(\hat{A} - \langle \hat{A} \rangle_t)\langle \hat{A} \rangle_t$, in

which case all nonlinear terms cancel, and the equation for ρ_t becomes of the Lindblad type:

$$\frac{\mathrm{d}}{\mathrm{d}t}\rho_t = -\mathrm{i}[\hat{H}_0, \rho_t] - \frac{\lambda}{2}[\hat{A}, [\hat{A}, \rho_t]].$$
(73)

In turn, the equation for the normalized state vector ψ_t reads:

$$d\psi_t = \left[-i\hat{H}_0 dt + \sqrt{\lambda} (\hat{A} - \langle \hat{A} \rangle_t) dW_t - \frac{\lambda}{2} (\hat{A} - \langle \hat{A} \rangle_t)^2 dt \right] \psi_t,$$
(74)

This is the structure of all collapse models. One can also proceed backwards [109] and show how, given (73), one unravels (74). It is not difficult to show that the new terms stochastically drive the state vector toward one of the eigenstates of the operator \hat{A} , with a probability equal to the Born rule [102, 111]. The above equation can be generalized to include more than one operator, therefore more than one noise [102]; non-self-adjoint operators [102]; more general noises, either complex [112], non white [113, 114, 115], or both [116].

So far, we presented the abstract formulation of collapse models. A *physical* model of spontaneous wave function collapse must justify the classical world emerging from an underlying quantum world: wave functions of macro-objects should be well-localized in *space*. The collapse should be negligible for microscopic system, and grow in strength with the *mass/size* of the system. This suggest taking for the collapse operator the *local* mass density $\hat{m}(\mathbf{x}) = \sum_{i} m_i \delta^{(3)}(\mathbf{x} - \hat{\mathbf{x}}_i)$, coupled to a white noise $w(\mathbf{x}, t)$ spread through space. Then (74) becomes

$$d\psi_t = \left[-\frac{\mathrm{i}}{\hbar} \hat{H} \,\mathrm{d}t + \sqrt{\lambda} \int \mathrm{d}^3 x \,(\hat{m}(\mathbf{x}) - \langle \hat{m}(\mathbf{x}) \rangle_t) \,\mathrm{d}W_t(\mathbf{x}) - \frac{\lambda}{2} \int \mathrm{d}^3 x \int \mathrm{d}^3 y \,\mathcal{G}(\mathbf{x} - \mathbf{y}) (\hat{m}(\mathbf{x}) - \langle \hat{m}(\mathbf{x}) \rangle_t) (\hat{m}(\mathbf{y}) - \langle \hat{m}(\mathbf{y}) \rangle_t) \,\mathrm{d}t \right] \psi_t, \quad (75)$$

(now we have re-introduced \hbar) where $W_t(\mathbf{x})$ is a family of Wiener processes, with spatial correlation function equal to $\mathcal{G}(\mathbf{x} - \mathbf{y})$. We assume that $\mathcal{G}(\mathbf{x} - \mathbf{y}) = \mathcal{G}(|\mathbf{x} - \mathbf{y}|)$, to respect translational and rotational symmetries. This equation was first proposed by Ghirardi, Pearle and Rimini and is the mass-proportional version [117, 118] of the Continuous Spontaneous Localization (CSL) model [102], if one takes a Gaussian correlation function: $\mathcal{G}(\mathbf{x} - \mathbf{y}) = \exp(-(\mathbf{x} - \mathbf{y})^2/4r_C^2)$, where the correlation length r_C (with dimension [L]) is a phenomenological parameter, and $\lambda = \lambda_{\text{CSL}}/m_0^2$, with m_0 a reference mass—equal to the mass of a nucleon, which is added to the definition of lambda to make the model mass-dependent—and λ_{CSL} (with dimension $[\mathrm{T}^{-1}]$) is the collapse rate, the second phenomenological parameter of the model.

This model has been extensively studied in the literature [103, 104]. Three properties are particularly important: i) The collapse drives the wave function towards states, which are localized in *space*, thus suppressing spatial superpositions. ii) If one considers composite systems such as rigid objects, then the center-of-mass dynamics satisfies a single-particle collapse equation, with an effective collapse rate $\lambda_{\text{CSL}}^{\text{CM}}$ which grows with the mass/size of the system ($\lambda_{\text{CSL}}^{\text{CM}} = f(N)\lambda_{\text{CSL}}$, where f(N) is a monotonically increasing function of the total number N of nucleons and of the geometry of the system). Therefore, with a suitable choice of the parameters (the original choice was $\lambda_{\rm CSL} = 10^{-16}$ s⁻¹ and $r_C = 10^{-7}$ m [101]) one can show that the collapse effects are negligible for microscopic systems, while for composite systems the amplification mechanism makes sure that the center-of-mass wave-function of macroscopic objects is always well-localized in space. iii) Quantum measurements are consistently described, and their entire phenomenology (definite outcomes, Born probability rule, von Neumann projection postulate, role of the operators as observables) emerges from the dynamics [119, 120].

For future reference, we write the master equation for the statistical operator
$$\rho_t = \mathbb{E}[|\psi_t\rangle\langle\psi_t|]$$
:

$$\frac{\mathrm{d}}{\mathrm{d}t}\hat{\rho}_t = -\frac{\mathrm{i}}{\hbar}[\hat{H}, \hat{\rho}_t] - \frac{\lambda}{2} \int \mathrm{d}^3x \int \mathrm{d}^3y \,\mathcal{G}(\mathbf{x} - \mathbf{y}) \left[\hat{m}(\mathbf{x}), \left[\hat{m}(\mathbf{y}), \hat{\rho}_t\right]\right],\tag{76}$$

which again is of the Lindblad type. We will refer to the second term—describing the collapse effect—as $\mathcal{L}[\hat{\rho}_t]$. For completeness, we write also the Fourier transformed equation:

$$\frac{\mathrm{d}}{\mathrm{d}t}\hat{\rho}_t = -\frac{\mathrm{i}}{\hbar}[\hat{H},\hat{\rho}_t] + \lambda \int \mathrm{d}^3k \, \frac{\mathcal{G}(\mathbf{k})}{(2\pi)^3} \left[\hat{m}(\mathbf{k})\,\hat{\rho}_t\,\hat{m}^\dagger(\mathbf{k}) - \frac{1}{2}\left\{\hat{m}^\dagger(\mathbf{k})\,\hat{m}(\mathbf{k}),\hat{\rho}_t\right\}\right],\tag{77}$$

where $\hat{m}(\mathbf{k})$ and $\mathcal{G}(\mathbf{k})$ are the Fourier transforms of $\hat{m}(\mathbf{x})$ and $\mathcal{G}(\mathbf{x})$, respectively.

The relevant open question for the present review is what causes the collapse, or equivalently, what is the origin of the noise $W_t(\mathbf{x})$. A tempting answer is that it has a gravitational origin, and this idea is supported by two considerations. The first one is that gravity naturally couples to the local mass density of each physical system, which is the natural choice for a collapse model, as discussed here above. The second one is that there is no proof so far that gravity is quantum, therefore the possibility is still open for it to provide the anti-hermitian coupling (see (70)), which is necessary to have a collapse equation. We will review four proposals, which connect gravity to the collapse of the wave function: The Diósi–Penrose model [121, 122, 123, 124], Adler's proposal [125], Károlyhaźy's model [126, 127, 128, 129, 130, 131], and the Schrödinger– Newton equation [132, 133, 134].

3.1. The Diósi-Penrose model

Diósi [121, 122, 123, 124] postulated (75) and assumed that the spatial correlator of the noise is proportional to the Newtonian gravitational potential [135, 136]:

$$\mathcal{G}(\mathbf{x}) = \frac{G}{\hbar} \frac{1}{|\mathbf{x}|};\tag{78}$$

this is the connection to gravity. Having defined $\mathcal{G}(\mathbf{x})$ this way, λ is a dimensionless constant, which Diósi set equal to 1. Equivalently, one can define $\mathcal{G}(\mathbf{x}) = 1/|\mathbf{x}|$ and $\lambda = G/\hbar$. The structure of the Diósi–Penrose equation being the same as that of the CSL equation, the collapse mechanism works the same way, with a different strength due to the different choice of the correlation function and collapse rate.

Gravitational Decoherence

One advantage of this equation, which was one of Diósi's motivations for proposing it, is that it is defined without free parameters, contrary to the CSL model, which contains two phenomenological parameters, the collapse rate λ and the spatial correlation length of the noise r_c . Here the collapse rate is governed by Newton's constant G, while the spatial correlation function is given by the "width" of the $1/|\mathbf{x}|$ function. However the model needs to be regularized, because $1/|\mathbf{x}|$ does not vanish for $|\mathbf{x}| \to \infty$ fast enough to make integrals converge. This is best seen by considering the Lindblad term in (77), which for Diósi's choice of the correlation function and for a fixed number N of particles becomes ($\mathbf{Q} = \hbar \mathbf{k}$):

$$\mathcal{L}_{\rm DP}^{\rm N}[\hat{\rho}_t] = \frac{G}{2\pi^2\hbar^2} \sum_{j,l=1}^N m_j \, m_l \, \int \frac{\mathrm{d}^3Q}{Q^2} \, f(Q) \\ \times \left[\mathrm{e}^{\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}_j} \, \hat{\rho}(t) \, e^{-\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}_l} - \frac{1}{2} \left\{ e^{\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}_j} \mathrm{e}^{-\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}_l}, \hat{\rho}(t) \right\} \right], \tag{79}$$

with f(Q) = 1 at this stage; for j = l integrals clearly diverge.

A cut-off can be introduced at the level of (76), by replacing the point-like density operator with a coarse-grained mass density operator, with a spatial resolution R_0 . Diósi originally introduced the coarse-grained mass density as follows:

$$\hat{m}'(\mathbf{x}) = \frac{3}{4\pi R_0^3} \int d^3 y \,\theta(R_0 - |\mathbf{x} - \mathbf{y}|) \,\hat{m}(\mathbf{y}),\tag{80}$$

where $\theta(x)$ is the Heaviside step function. Subsequently, Ghirardi *et al* [117] introduced the coarse-graining as follows:

$$\hat{m}'(\mathbf{x}) = (2\pi R_0^2)^{-3/2} \int d^3 y \, \exp\left(-\frac{|\mathbf{x} - \mathbf{y}|^2}{2R_0^2}\right) \, \hat{m}(\mathbf{y}). \tag{81}$$

Note that $\hat{m}'(\mathbf{x})$ is meant to replace $\hat{m}(\mathbf{x})$ in (76). These two cut-offs are practically equivalent, and in the following we will consider only the second one. (79) remains the same, with now

$$f(Q) = \exp\left(-\frac{Q^2 R_0^2}{\hbar^2}\right).$$
(82)

Now integrals converge. Diósi proposed to set $R_0 \simeq 10^{-15}$ m, which is the nucleon's radius. It is a reasonable assumption, given that this model is intrinsically nonrelativistic.

The cut-off can be also introduced directly at the level of (79), by limiting the momentum integral to $|\mathbf{Q}| \leq Q_{\max} = \hbar/R_0$. It is easy to see that the new equation is mathematically equivalent to (79) with $f(Q) = \theta(Q - Q_{\max})$, which eventually can be approximated by a damping Gaussian function, $f(Q) \approx e^{-Q^2/Q_{\max}^2}$, such as (82). In this way one can assign an interpretation to the cut-off and even justify a specific value for it. In fact, Q_{\max} can be read as an upper limit for the modes of the collapse field that are the dominant modes contributing to the collapse. These modes are also small enough to justify the nonrelativistic approach. This interpretation is very similar to Bethe's nonrelativistic computation of the Lamb shift (see [137]). Thus, if we argue

Gravitational Decoherence

that Q_{max} is the bound justifying the nonrelativistic approach, then one can replace R_0 by the Compton wavelength; that is to say: $R_0 = \frac{2\pi\hbar}{mc}$, which is $R_0 \approx 10^{-15}$ m for a nucleon. Accordingly, in this way, one can provide a justification for the cut-off and its chosen value. This choice, however, is incompatible with experimental data. We will discuss this issue soon.

We now discuss how efficient the collapse is, one measure of it being how fast the offdiagonal elements of the density matrix decay in time. Let us consider the one-particle version of (79):

$$\mathcal{L}_{\rm DP}^{1}[\hat{\rho}_{t}] = \int d^{3}Q \,\Gamma_{\rm DP}(\mathbf{Q}) \left(e^{\frac{i}{\hbar} \,\mathbf{Q} \cdot \hat{\mathbf{r}}} \,\hat{\rho}(t) \, e^{-\frac{i}{\hbar} \,\mathbf{Q} \cdot \hat{\mathbf{r}}} - \hat{\rho}(t) \right), \tag{83}$$

with

$$\Gamma_{\rm DP}(\mathbf{Q}) = \frac{Gm^2}{2\pi^2\hbar^2} \frac{1}{Q^2} \exp\left(-\frac{Q^2 R_0^2}{\hbar^2}\right).$$
(84)

A measure of the collapse rate is given by:

$$\Lambda_{\rm DP} = \int d^3 Q \,\Gamma_{\rm DP}(\mathbf{Q}) = \frac{Gm^2}{\sqrt{\pi}\hbar R_0},\tag{85}$$

with $[\Lambda_{\rm DP}] = {\rm s}^{-1}$. For a nucleon it gives: $\Lambda_{\rm DP} \simeq 10^{-15} {\rm s}^{-1}$, implying that coherences for microscopic particles are stable over extremely long times. On the other hand, as anticipated, for macroscopic objects the collapse is amplified. Assuming a rigid many-body system and tracing out the relative coordinates, the dynamical equation for the center-of-mass density-matrix $\hat{\rho}_t^{\rm M}$ takes the same form as in (83), where $\Gamma_{\rm DP}(\mathbf{Q})$ is replaced by [136]:

$$\Gamma^{\rm M}_{\rm DP}(\mathbf{Q}) = \frac{G}{2\pi^2 \hbar^2 Q^2} \left| \tilde{\mu}_{\rm rel}(\mathbf{Q}) \right|^2 \exp\left(-\frac{Q^2 R_0^2}{\hbar^2}\right) \,, \tag{86}$$

with $\tilde{\mu}_{\rm rel}(\mathbf{Q}) = \int d^3x \, e^{i\mathbf{Q}\cdot\mathbf{x}/\hbar} \, \mu_{\rm rel}(\mathbf{x})$, where $\mu_{\rm rel}(\mathbf{x})$ is the internal mass density. For example, for a homogeneous rigid sphere of mass M and radius R, we get: $\tilde{\mu}_{\rm rel}(\mathbf{Q}) \approx M \exp(-Q^2 R^2/8\hbar^2)$. Accordingly, we find:

$$\Gamma^{\rm M}_{\rm DP}(Q) \simeq \frac{G M^2}{2\pi^2 \hbar^2} \frac{1}{Q^2} \exp\left(-\frac{Q^2(R^2 + 4R_0^2)}{4\hbar^2}\right).$$
(87)

Similar to (85), here the total collapse rate becomes:

$$\Lambda_{\rm DP}^{\rm M} = \int d^3 Q \,\Gamma_{\rm M}(\mathbf{Q}) \simeq \frac{2GM^2}{\hbar \sqrt{\pi (R^2 + 4R_0^2)}}.$$
(88)

For example, $\Lambda_{\text{DP}}^{\text{M}}$ is of order 10^{-5} s^{-1} for a typical optomechanical nanosphere with $M \simeq 10^9$ u and $R \simeq 50$ nm [138]. Evidently, the collapse rate of a nanosphere is ten orders of magnitude larger than that of a nucleon, and would be even larger for a truly macroscopic system. Therefore, the model is capable of describing both the quantum properties of microscopic systems and the classical properties of macroscopic objects.

It is interesting to write down the explicit time evolution of the off-diagonal elements; this will give the connection between Diósi's model and Penrose's idea, which eventually justifies why the model is called the Diósi–Penrose model. Given the master equation (76), with Diósi's choice for the collapse rate and spatial correlator of the noise, and using the characteristic function [139, 140], the one-particle state at time t in the position representation, $\langle \mathbf{x} | \hat{\rho}^1(t) | \mathbf{x}' \rangle = \rho(\mathbf{x}, \mathbf{x}', t)$, is found to be

$$\rho(\mathbf{x}, \mathbf{x}', t) = \int d^3 y \int \frac{d^3 p}{(2\pi\hbar)^3} \rho_0(\mathbf{x} + \mathbf{y}, \mathbf{x}' + \mathbf{y}, t) \\ \times \exp\left(-\frac{\mathrm{i}}{\hbar}\mathbf{y} \cdot \mathbf{p} - \frac{1}{\hbar} \int_0^t d\tau \left(U\left(\frac{\mathbf{p}\,\tau}{m} + \mathbf{x} - \mathbf{x}'\right) - U(\mathbf{0})\right)\right), \tag{89}$$

where $\rho_0(\mathbf{x}, \mathbf{x}', t)$ is the solution of the free Schrödinger dynamics and

$$U(\mathbf{x}) = -G \int d^3r \int d^3r' \, \frac{m'(\mathbf{r}) \, m'(\mathbf{r}')}{|\mathbf{x} + \mathbf{r} - \mathbf{r}'|} = -Gm^2 \, \frac{\operatorname{Erf}(|\mathbf{x}|/R_0)}{|\mathbf{x}|},\tag{90}$$

is the Newtonian self-interaction where Erf(x) is the Gauss error function. If one neglects the pure Schrödinger contribution in (76), which is justified on the short time scale, then (89) reduces to an exponential decay of the form:

$$\rho(\mathbf{x}, \mathbf{x}', t) = \exp\left(-\frac{t}{\tau(\mathbf{x}, \mathbf{x}')}\right) \rho(\mathbf{x}, \mathbf{x}', 0),$$
(91)

where the characteristic damping time τ is:

$$\tau(\mathbf{x}, \mathbf{x}') = \frac{\hbar}{U(\mathbf{x} - \mathbf{x}') - U(\mathbf{0})}.$$
(92)

Again, (91) implies that spatial superpositions of positions \mathbf{x} and \mathbf{x}' decay with the rate $\tau(\mathbf{x}, \mathbf{x}')$. This is precisely the endpoint of Penrose's idea [141, 142, 143, 133, 134, 4].

According to Penrose, given two macroscopic lumps of matter at different locations, each being in a stationary state, i.e. an energy eigenstate, with the same eigenvalue, then also their superposition is a stationary state and persists in time, according to quantum theory. This is another way to express the Schrödinger's cat problem. However, if one takes also gravity into account, Penrose argues, things change. Each state will give rise to a stationary spacetime geometry, and the two differ macroscopically from each other. When a superposition of the two lumps is considered, then also the two spacetime geometries will be superimposed, and this will not be stationary any longer. Penrose writes: "We have to consider carefully what a 'stationary state' means in a context such as this. In a stationary spacetime, we have a well-defined concept of 'stationary' for a quantum state in that background, because there is a Killing vector T in the spacetime that generates the time-translations. Regarding T as a differential operator (the $\partial/\partial t$) for the spacetime), we simply ask for the quantum states that are eigenstates of T, and these will be the stationary states, i. e. states with well-defined energy values. $[\dots]$ However, for the superposed state we are considering here we have a serious problem. For we do not now have a specific spacetime, but a superposition of two slightly differing spacetimes. How are we to regard such a 'superposition of spacetimes'? Is there an operator that we can use to describe 'time-translation' in such a superposed spacetime? Such an operator would be needed so that we can identify the 'stationary states' as its eigenvectors, these being the states with definite energy. It will be shown that there is a fundamental difficulty with these concepts, and that the notion of time-translation operator is essentially ill defined" [133].

This ill-definedness leads to an uncertainty in the energy, which makes the superposition unstable. Penrose provides a quantitative measure of the energy uncertainty: it is the gravitational self-energy of the *difference* between the two mass distributions ρ and ρ' of the two superposed states,

$$\Delta E = -4\pi G \int d^3r \int d^3r' \frac{(\rho(\mathbf{r}) - \rho'(\mathbf{r})) \left(\rho(\mathbf{r}') - \rho'(\mathbf{r}')\right)}{|\mathbf{r} - \mathbf{r}'|} \,. \tag{93}$$

As Penrose points out, this coincides with the use of the Newtonian gravitational interaction energy of the corresponding mass distribution at two positions according to (90), as long as both superposed states have the same gravitational self-energy (i. e. for a displacement, but not for a shape-changing superposition). Using the energytime uncertainty relation, this can imply that the superposition decays to one of the localized states with the lifetime $\tau = \hbar/\Delta E$, which is equivalent to (92). Therefore, Diósi's dynamical equation implements Penrose's idea. This is why one speaks of the Diósi–Penrose model.

We have seen that collapse models couple quantum systems to an external classical noise. As such, the system undergoes a Brownian motion and energy is exchanged. In particular, the energy of the quantum system is not conserved anymore. The CSL model, and the Diósi–Penrose model as well, do not include dissipative effects, therefore the energy of the system steadily increases and eventually diverges, as if the noise is at infinite temperature. The rate of energy increase can be easily calculated and turns out to be:

$$\frac{\mathrm{d}E_{\rm DP}(t)}{\mathrm{d}t} = \frac{2\pi}{m} \int_0^\infty \mathrm{d}^3 Q \,\Gamma_{\rm DP}(Q) \,Q^4 = \frac{m \,G \,\hbar}{4\sqrt{\pi} \,R_0^3},\tag{94}$$

with $E_{\rm DP}(t) = {\rm Tr}[\hat{\rho}(t)\hat{H}]$ where $\hat{\rho}(t)$ satisfies the Diósi–Penrose dynamics. From this relation, one can easily evaluate the different implications of the cut-off proposed, respectively, by Diósi [135] and Ghirardi *et al* [117]. In the former case, $R_0 = 10^{-15}$ m, one gets a rate for the energy increase of order 10^{-4} K/s for a proton, which means a thermal catastrophe! Precisely this fact induced Ghirardi *et al* to introduce a much larger value, $R_0 = 10^{-7}$ m, in which case the rate is of order 10^{-28} K/s for a proton, which is indeed a much more reasonable value. Although in this way the problem of overheating has been partially resolved, it is clear that the introduction of a cut-off $R_0 = 10^{-7}$ m is much less justified than the one originally proposed by Diósi. One of the main motivations of the Diósi–Penrose model, namely, to provide a phenomenological model without free parameters, is in this way lost.

In [144] the possibility has been explored, whether the original value $R_0 = 10^{-15}$ m can be retained, while including dissipative effects, which mitigate the fast energy increase. This would be a rather natural resolution of the problem: physical noises cannot pump too much energy in the system, which should eventually thermalize to the (finite) temperature of the noise. This procedure successfully works for the CSL

model [145] (see also [146, 147]), however fails for the Diósi–Penrose model: in this case, thermalization can be formally achieved, but at the price of introducing very strong dissipation, causing with high probability sudden flips in momentum of the system. The momentum flip induced by the universal noise field causing the collapse of the wave function would mean that the latter can transfer an energy of the order of tens of MeV to a nucleon in a nucleus (corresponding to the average kinetic energy of a nucleon in a Fermi-gas model [148]), which would induce instantaneous matter dissociation. Accordingly, with $R_0 = 10^{-15}$ m, the Diósi–Penrose model is only an effective model for the mass ranges larger or comparable with $m_r \sim 10^{11}$ u. In other words, the Diósi–Penrose model should be applied only to mesoscopic and macroscopic systems. It is worth mentioning that the value m_r is also very different from the Planck mass $m_{\rm Pl} \sim 10^{19}$ u, which is sometimes considered as a borderline between quantum and classical masses [31].

3.2. Adler's proposal

Diósi assumes (75) and postulates (78) for the correlation function of the noise. Adler's point of view is different [125]. He starts with (70), which can be taken as the starting point to justify the collapse (74) as we showed in the introduction of section 3, and thus of (75) and (76) with the appropriate choice of the collapse operators. Next he notices that one can naturally introduced a *mass-proportional anti-hermitian* coupling to the wave function by assuming that gravity is fundamentally *classical* and its metric contains a rapidly fluctuating *complex* component. In other words:

$$g_{\mu\nu} = \overline{g}_{\mu\nu} + h_{\mu\nu},\tag{95}$$

with $\overline{g}_{\mu\nu}$ the conventional real spacetime metric, and $h_{\mu\nu}$ an irreducibly complex fluctuation. For simplicity, we will set $\overline{g}_{\mu\nu} = \eta_{\mu\nu}$, i. e. we will consider only perturbation around the flat Minkowski metric. It is easy to show how, under these premises, one obtains the desired coupling term.

The standard action of a matter field coupled to gravity is:

$$S = \int \mathrm{d}^4 x \sqrt{-g} \,\mathcal{L}_{\mathrm{matter}},\tag{96}$$

where $\mathcal{L}_{\text{matter}}$ is the matter Lagrangian, $g_{\mu\nu}$ is the metric tensor and $\sqrt{-g} = \sqrt{-\det[g_{\mu\nu}]}$. The Taylor expansion around $\eta_{\mu\nu}$ gives:

$$S = \int d^4x \left[\mathcal{L}_{\text{matter}}^{(0)} - \frac{1}{2} h^{\mu\nu} T^{(0)}_{\mu\nu} \right] + \text{higher order terms;}$$
(97)

the apex (0) denotes quantities with respect to the flat spacetime $\eta_{\mu\nu}$, and $T_{\mu\nu}$ is the stress energy tensor associated to the Lagrangian $\mathcal{L}_{\text{matter}}$. In the weak field limit, gravity couples to matter linearly through the stress energy tensor.

In the nonrelativistic limit, considering for example the Lagrangian of a Klein-Gordon field ϕ , one can show that $h^{\mu\nu}T^{(0)}_{\mu\nu} \simeq mc^2 h^{00} \phi^* \phi$. Second quantization of the

field then immediately gives the one-particle Hamiltonian $\mathbf{p}^2/2m + mc^2h^{00}/2$, therefore the N-particle Hamiltonian can be written as:

$$H = H_0 + c^2 \xi \int \mathrm{d}^3 x \, \hat{m}(\mathbf{x}) w(\mathbf{x}, t), \tag{98}$$

where H_0 is the kinetic term, $\hat{m}(\mathbf{x})$ is the mass density operator introduced in the introduction of section 3, and we have renamed $h^{00}(\mathbf{x},t)/2 = \xi w(\mathbf{x},t)$, where $w(\mathbf{x},t)$ is an a-dimensional complex random field, whose fluctuations are of order 1, while the constant ξ measures the size of the fluctuations. The real part of $w(\mathbf{x},t)$ contributes to the Hamiltonian as a standard random potential, giving rise to decoherence effects as discussed in section 2.1. Its imaginary part provides the anti-hermitian random coupling, which is the starting point for deriving the collapse dynamics.

For the purposes of this review, it is not particularly relevant to write down the collapse equation arising from (98). It is the generalization of (75) to *complex* and *non-white* noises, and in general it can be written down explicitly only as a perturbative expansion. More details can be found in [116]. One can show that the usual properties hold: wave functions are localized in space and the amplification mechanism assures that the collapse rate grows with the mass/size of the system.

It is more instructive to consider the corresponding master equation for the density matrix ρ_t . The Hamiltonian evolution is the usual one, while the Lindblad term reads:

$$\mathcal{L}_{\mathrm{A}}[\hat{\rho}_{t}] = -\frac{\xi^{2}c^{4}}{\hbar^{2}} \int \mathrm{d}^{3}x \, \int \mathrm{d}^{3}y \int_{0}^{t} \mathrm{d}\tau \, \mathcal{G}_{t-\tau}^{\mathrm{R}}(\mathbf{x}-\mathbf{y}) \left[\hat{m}(\mathbf{x}), \left[\hat{m}_{\tau-t}(\mathbf{y}), \rho_{t}\right]\right] + \\ -\frac{\mathrm{i}\xi^{2}c^{4}}{\hbar^{2}} \int \mathrm{d}^{3}x \, \int \mathrm{d}^{3}y \int_{0}^{t} \mathrm{d}\tau \, \mathcal{G}_{t-\tau}^{\mathrm{I}}(\mathbf{x}-\mathbf{y}) \left[\hat{m}(\mathbf{x}), \left\{\hat{m}_{\tau-t}(\mathbf{y}), \rho_{t}\right\}\right],$$
(99)

where $\mathcal{G}^{\mathbf{R}}$ and $\mathcal{G}^{\mathbf{I}}$ are the real and the imaginary parts of the correlation function of the noise field: $\mathcal{G}_{t,\tau}(\mathbf{x}, \mathbf{y}) = \mathbb{E}[w^*(\mathbf{x}, t)w(\mathbf{y}, \tau)]$. In writing the above equation, we assumed that the noise is statistically homogeneous over space and time: $\mathcal{G}_{t,\tau}(\mathbf{x}, \mathbf{y}) = \mathcal{G}_{t-\tau}(\mathbf{x}-\mathbf{y})$. Also, $\hat{m}_{\tau-t}(\mathbf{y})$ is the time evolved mass density operator, the time evolution generated by the Hamiltonian H_0 . The equation, which is valid up to second perturbative order in the parameter ξ , generalizes (76) to the case of complex and non-white noises. Here we can see a rather important difference between Adler's model and the Diósi–Penrose model. While both rely on the same structure for the collapse equation, the latter gives a specific form for the correlation function of the noise—the Newtonian classical gravitational potential—while Adler's model does not specify the form of the correlation function, only requiring the gravitational fluctuations to have a complex fluctuating component.

At this stage, one is not able to characterize the noise correlator in Adler's model. One might guess that, if the noise has a cosmological origin, its correlation function is highly non trivial, but apart from this nothing more can be said unless one establishes the origin of the noise, which is unknown. What one can do is to analyze bounds on the strength of the fluctuations, by comparing the predictions of the model with experimental data. To this purpose, let us consider the center-of-mass equation of a composite system. Under suitable approximations, which are typically valid for crystalline structures forming macroscopic objects, the center of mass and relative motions decouple and the Lindblad term for the center-of-mass density-matrix $\hat{\rho}_t^{\text{M}}$ takes the form [116]:

$$\mathcal{L}_{\mathrm{A}}[\hat{\rho}_{t}^{\mathrm{M}}] = -\frac{\xi^{2}c^{4}}{\hbar^{2}} \frac{1}{(2\pi\hbar)^{3}} \int_{0}^{t} \mathrm{d}\tau \int \mathrm{d}^{3}Q \,\Gamma_{t-\tau}^{\mathrm{M},\mathrm{R}}(\mathbf{Q}) \left[\mathrm{e}^{-\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}}, \left[\mathrm{e}^{\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}}_{\tau-t}, \hat{\rho}_{t}^{\mathrm{M}} \right] \right] + \\ -\frac{\mathrm{i}\xi^{2}c^{4}}{\hbar^{2}} \frac{1}{(2\pi\hbar)^{3}} \int_{0}^{t} \mathrm{d}\tau \int \mathrm{d}^{3}Q \,\Gamma_{t-\tau}^{\mathrm{M},\mathrm{I}}(\mathbf{Q}) \left[\mathrm{e}^{-\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}}, \left\{ \mathrm{e}^{\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}}_{\tau-t}, \bar{\rho}_{t} \right\} \right], \tag{100}$$

where $\Gamma_{t-\tau}^{\text{M},\text{R}/\text{I}}(\mathbf{Q}) = \Gamma_{t-\tau}^{\text{R}/\text{I}}(\mathbf{Q})|\tilde{\mu}_{\text{rel}}(\mathbf{Q})|^2$, with $\Gamma_{t-\tau}^{\text{R}/\text{I}}(\mathbf{Q})$ the Fourier transform of $\mathcal{G}_{t-\tau}^{\text{R}/\text{I}}(\mathbf{x})$ and $\tilde{\mu}_{\text{rel}}(\mathbf{Q})$ the Fourier transform of the internal mass density $\mu_{\text{rel}}(\mathbf{x})$. The above expression should be compared with the analogous expression (83) of the Diósi–Penrose model, and again it represents its generalization to complex and non-white noises.

As discussed in section 3.1, one prediction of (100) is that the collapse noise induces a Brownian diffusion, which implies of a variety of non-standard effects, like an excess diffusion of center-of-mass motion [149, 150, 151, 152, 153], spontaneous heating [154, 155], and spontaneous photon emission [156, 157, 158] from charged particles. Available experimental data allow to sets bounds on the noise correlator, based on these effects.

Discussing the experimental constraints on the noise correlator in its full generality is too difficult. We will limit the discussion to a restricted class of Gaussian correlations functions, in such a way that the collapse dynamics is controlled by only two parameters. Specifically, we consider the Markovian limit by imposing:

$$\Gamma_s^{\mathrm{R}/\mathrm{I}}(\mathbf{Q}) \simeq \Gamma^{\mathrm{R}/\mathrm{I}}(\mathbf{Q}) \tau_0 \delta(s) \tag{101}$$

with $[\tau_0] = [T]$. One can show that in this limit $\Gamma^{I}(\mathbf{Q}) = 0$. In addition, we assume that $\Gamma^{R}(\mathbf{Q})$ has the a Gaussian shape:

$$\Gamma^{\mathrm{R}}(\mathbf{Q}) = r_C^3 \exp(-r_C^2 \mathbf{Q}^2/\hbar^2), \qquad (102)$$

as in the CSL model, where $[r_C] = [L]$. With these assumptions, after some algebra, the Lindblad operator in (100) reduces to

$$\mathcal{L}_{\mathrm{A}}[\hat{\rho}_{t}^{\mathrm{M}}] = -\frac{\xi^{2}c^{4}r_{C}^{3}\tau_{0}}{(2\pi\hbar)^{3}2\hbar^{2}} \int \mathrm{d}^{3}Q \, |\tilde{\mu}_{\mathrm{rel}}(\mathbf{Q})|^{2} \mathrm{e}^{-r_{C}^{2}Q^{2}/\hbar^{2}} \left[\mathrm{e}^{-\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}}, \left[\mathrm{e}^{\frac{\mathrm{i}}{\hbar}\mathbf{Q}\cdot\hat{\mathbf{r}}}, \hat{\rho}_{t} \right] \right].$$
(103)

This equation should be compared with the CSL master equation [102]:

$$\mathcal{L}_{\rm CSL}[\hat{\rho}_t^{\rm M}] = -\frac{\lambda (4\pi r_C^2)^{3/2}}{(2\pi\hbar)^3 m_0^2} \int \mathrm{d}^3 Q \, |\tilde{\mu}_{\rm rel}(\mathbf{Q})|^2 \mathrm{e}^{-r_C^2 \mathbf{Q}^2/\hbar^2} \Big[\mathrm{e}^{-\frac{\mathrm{i}}{\hbar} \mathbf{Q} \cdot \hat{\mathbf{r}}}, \left[\mathrm{e}^{\frac{\mathrm{i}}{\hbar} \mathbf{Q} \cdot \hat{\mathbf{r}}}, \hat{\rho}_t \right] \Big]. \tag{104}$$

In particular, (103) reduces to the CSL master equation given in (104) by setting:

$$\xi = \frac{4\hbar\pi^{3/4}}{m_0 c^2} \sqrt{\frac{\lambda}{\tau_0}}.$$
(105)

With this approximations the model is fully characterized by: the magnitude ξ of the metric fluctuations, the time cut-off τ_0 and the space cut-off r_C . To further simplify the discussion, as first proposed in [113] we assume the time cut-off to be related to



Figure 1. (ξ, r_C) or equivalently (ξ, τ_0) parameter diagram of the gravity-induced collapse model. The white area is the allowed region. The blue shaded region (*X*rays) is excluded by data analysis of X-rays measurements [159]. The orange shaded region (*LISA*) is excluded from data analysis of LISA Pathfinder [160]. The green shaded region (*Macro*) is an estimate of the region excluded by the requirement that the collapse is strong enough to localize macroscopic objects [161, 162]. Note that X-ray measurements sample the high frequency region of the spectrum (~ 10¹⁸ Hz) and would disappear if the noise correlator has a cut-off below such frequencies, which is plausible. In such a case, the stronger upper bound on the left part of the plane is given by data analysis with cold atom experiments (*Cold atoms*) [155].

the space cut-off via $\tau_0 = r_C/c$. One can then give bounds on ξ by using the bounds already set for the CSL parameters λ and r_C . The most recent bounds are summarized in figure 1.

It is interesting to compare these results with the recent discovery of gravitational waves [12], observed in frequency range from 35 to 250 Hz and with a peak strain of 1.0×10^{-21} . Clearly, gravitational waves are real, while here the claim is that the collapse is caused by complex fluctuations of the metric. Also, significant gravitational waves typically have long wave lengths, while here the relevant part of the spectrum is at high frequencies. However it is interesting to see that the order of magnitude of real waves and complex fluctuations—which allow for an efficient collapse and are compatible with experimental data—are not so far away from each other.
3.3. Károlyházy model

The key idea of Károlyházy [126, 127, 128, 129, 130, 131], later developed by Frenkel [163, 164, 165, 166], is that lengths and times cannot be measured with arbitrary precision, given that measuring devices are made of atoms and therefore are ultimately quantum and must obey the Heisenberg uncertainty principle. Thus there is a fundamental uncertainty in the structure of spacetime. By using the uncertainty principle and some basic arguments, he arrives at the following relation [126]:

$$\Delta s^2 = \ell_{\rm Pl}^{4/3} s^{3/2},\tag{106}$$

where $\ell_{\rm Pl} = \sqrt{\hbar G/c^3}$ is the Planck length. The above relation expresses the best precision with which a length s can be measured.

So spacetime is undetermined beyond some scale, according to Károlyházy. He expresses this uncertainty by promoting the metric $g_{\mu\nu}(x)$ to a stochastic metric $g_{\mu\nu}(\omega, x)$, and averages are taken in the end. In the weak field limit we have:

$$g_{\mu\nu}(\omega, x) = \eta_{\mu\nu} + h_{\mu\nu}(\omega, x), \qquad (107)$$

where $h_{\mu\nu}(\omega, x)$, as usual, represent random perturbations around the flat spacetime metric (from now on we will omit the random variable ω), which are assumed to obey the wave equation, in the appropriate gauge:

$$\Box h_{\mu\nu} = 0, \qquad (108)$$

where \Box denotes the d'Alembert operator. This is true if we neglect the effect of matter on spacetime fluctuations. Equation (108) allows for a plane-wave expansion:

$$h_{\mu\nu}(x) = \frac{1}{\sqrt{L^3}} \sum_{\mathbf{k}} \left[c_{\mu\nu}(\mathbf{k}) \mathrm{e}^{\mathrm{i}\mathbf{k}\cdot\mathbf{x}-\omega_{\mathbf{k}}t} + \mathrm{c.c.} \right], \tag{109}$$

with $x = \mathbf{x}$, t and $\omega_{\mathbf{k}} = c|\mathbf{k}|$. Now the Fourier coefficients $c_{\mu\nu}(\mathbf{k})$ are random, and they are assumed to fluctuate around zero: $\mathbb{E}[c_{\mu\nu}(\mathbf{k})] = 0$, with $\mathbb{E}[\cdot]$ denoting the stochastic average. Box normalization (of side L) has been assumed.

Given the standard definition of length,

$$s^{2} = \int \left(-g_{\mu\nu} \frac{dx^{\mu}}{dt} \frac{dx^{\nu}}{dt} \right) dt, \qquad (110)$$

which again is random since $g_{\mu\nu}$ is, the fundamental relation (106), where we identify $\Delta s^2 = \mathbb{E}[(s - \mathbb{E}[s])^2]$, this yields the following spectrum for the fluctuations $c(\mathbf{k}) \equiv c_{00}(\mathbf{k})$, the only ones which will be relevant for the following discussion:

$$\mathbb{E}[c(\mathbf{k})c^{*}(\mathbf{k}')] = \delta_{\mathbf{k},\mathbf{k}'} \,\ell_{\rm Pl}^{4/3} k^{-5/3}.$$
(111)

So (106) selects a specific stochastic behavior of the metric fluctuations.

Now calculations proceed in the standard way. In the nonrelativistic and Newtonian limit, the metric perturbation enters the Schrödinger equation as a random potential $V(\mathbf{x},t) = mc^2 h_{00}(\mathbf{x},t)/2$, the generalization to many particles being straightforward. The resulting dynamics is stochastic and leads to decoherence, as different terms of

a *spatial* superposition will acquire different phases, which wash away the coherent behavior.

The decoherence time τ has been computed and it can be expressed as: $\tau = ma_c^2/\hbar$, where a_c is the shortest coherence distance, to which the stochastic dynamics assigns a phase mismatch $\simeq \pi$, which gives full decoherence. Calculations give:

$$a_c \simeq \frac{\hbar}{G} \frac{1}{m^3},\tag{112}$$

for an elementary particle of mass m. For example, for one electron one gets: $a_c \simeq 10^{35}$ cm and correspondingly $\tau \simeq 10^{70}$ s. This gravitational decoherence effect is fully negligible for microscopic systems. On the other hand, for a macroscopic object of mass M and linear dimensions R one has:

$$a_c \simeq \left(\frac{\hbar^2}{G}\right)^{1/2} \frac{R^{2/3}}{M};\tag{113}$$

for an object of size $\simeq 1$ cm and normal density, one gets $a_c \simeq 10^{-16}$ cm and $\tau \simeq 10^{-4}$ s. Macroscopic superpositions rapidly decay.

Károlyházy's model assumes the validity of the Schrödinger equation, supplemented by a random gravitational potential with a specific correlation function. As such, although often quoted as a collapse model as those described so far in this section, it is not: here the superposition principle remains valid and macroscopic superpositions are still solutions of the equations of motions. As such, the model does not represent a resolution of the quantum measurement problem. It is, however, interesting because it shows that when quantum systems couple to gravity, even if gravity is treated classically, *intrinsic* fluctuations can exist, which cause decoherence preventing to fully control the quantum state of the system.

3.4. The Schrödinger-Newton equation

The Schrödinger–Newton equation was first proposed by Diósi [132] and Penrose [133, 134], who were interested in its stationary solutions [167, 168, 169] as the final states of the wave function collapse. Its dynamical behavior received renewed attention, mainly due to its connection with the question whether gravity should really be quantized [170], and due to its falsifiability in envisaged experiments [171, 172, 173, 174] (cf. section 4). The one particle Schrödinger–Newton equation reads:

$$i\hbar \frac{\partial}{\partial t}\psi(t,\mathbf{r}) = \left(-\frac{\hbar^2}{2m}\nabla^2 - Gm^2 \int d^3r' \frac{|\psi(t,\mathbf{r}')|^2}{|\mathbf{r}-\mathbf{r}'|}\right)\psi(t,\mathbf{r}).$$
(114)

It is a nonlinear equation, the nonlinearity arising from the self-gravitational interaction among different parts of the wave function, as if the (absolute value squared of the) wave function represents the mass distribution of the system. This is an attractive effect, which opposes the quantum diffusion originating from the Laplacian. Therefore one expects a slower spreading of the wave function over time, with respect to the usual quantum dynamics, and possibly a collapse for large masses. Asymptotically, a stable solitonic state should remain, where attraction and diffusion exactly compensate. In particular, given a superposition of two states displaced in space, the Schrödinger–Newton term gives a mutual attraction, which turns the superposition into a localized state. This is why this equation is discussed in connection with the quantum measurement problem as a possible way to explain why macroscopic superpositions are not observed: for massive systems, the Schrödinger–Newton attractive term becomes dominant and kill any possible superposition in space. We will come back on this issue.

One relevant question is the justification of the Schrödinger–Newton equation, which has given rise to a rather lively debate [175, 176, 177, 178]. The claim of the supporters of this equation is that it follows from semi-classical gravity. Here one has to be clear about the meaning of *semi-classical qravity*. The majority of the physics community assumes that gravity must be quantized in some way, and that semi-classical gravity is only an *effective* theory, which holds in situations where matter is treated quantum mechanically and gravity classically (although it is fundamentally quantum). This effective description comprises so far all observable physical phenomena. From this perspective, from which the fundamental theory is fully quantum, the evolution of the wave function is linear, and one cannot expect any nonlinear term in the dynamics. The self-interactions present in the classical theory would then be treated in the same way as in Quantum Electrodynamics, namely, through the normal ordering and renormalization prescription. They would lead to a mass-renormalization of the theory rather than a potential term in the Schrödinger equation [175, 176]. Within this framework, the Schrödinger–Newton equation arises as a *mean-field* equation, like the Gross–Pitaevskii equation [179, 180] commonly used in condensed matter theory, whose validity is restricted to the case of large numbers of particles.

However, the key assumption of the proponents of the Schrödinger–Newton equation is different. The claim is that it does follow from a theory in which only matter fields are quantized, while the gravitational field remains *fundamentally* classical. If one refers to this theory as semi-classical gravity [181, 182, 183], then the Schrödinger–Newton equation does follow from it, as we now review [184].

The most natural dynamical equation of such an hybrid theory is provided by the semi-classical Einstein equations:

$$R_{\mu\nu} - \frac{1}{2}g_{\mu\nu}R = \frac{8\pi G}{c^4} \left\langle \Psi \mid \hat{T}_{\mu\nu} \mid \Psi \right\rangle, \qquad (115)$$

where the classical energy-momentum tensor in Einstein's equations is replaced by the expectation value of the corresponding quantum operator in a given quantum state Ψ . It is clear that this equation, taken seriously, gives a nonlinear evolution for the wave function. (115) has a long history, dating back to the works of Møller [185] and Rosenfeld [186]. It has been commented repeatedly that such a theory would be incompatible with established principles of physics [187, 188] but these arguments can be refuted [182, 183, 189, 190].

It is important to note that (115) implies the conservation of energy-momentum, $\partial^{\mu}\langle \Psi \mid \hat{T}_{\mu\nu} \mid \Psi \rangle = 0$, even if nonlinear terms are present, which "collapse" the wave function. This is not the case for the standard instantaneous collapse in quantum mechanics, as well as of spontaneous wave function collapse models, as discussed in section 3.1 and 3.2.

Let us expand the metric around the flat metric:

$$g_{\mu\nu} = \eta_{\mu\nu} + h_{\mu\nu} \,, \tag{116}$$

the expansion in $h_{\mu\nu}$ is well-known to yield the gravitational wave equations at leading order. (115) approximates to [191]

$$\Box h_{\mu\nu} = -\frac{16\pi G}{c^4} \left(\langle \Psi \mid \hat{T}^{(0)}_{\mu\nu} \mid \Psi \rangle - \frac{1}{2} \eta_{\mu\nu} \langle \Psi \mid \eta^{\rho\sigma} \hat{T}^{(0)}_{\rho\sigma} \mid \Psi \rangle \right) , \qquad (117)$$

where the de Donder gauge-condition $\partial^{\mu}(h_{\mu\nu} - \frac{1}{2}\eta_{\mu\nu}\eta^{\rho\sigma}h_{\rho\sigma}) = 0$ has been imposed. Note that the energy-momentum tensor at this order of the linear approximation is that for flat spacetime, while in (115) it was still in curved spacetime. In the Newtonian limit, where $\langle \Psi \mid \hat{T}_{00}^{(0)} \mid \Psi \rangle$ is large compared to the other nine components of the energymomentum tensor, (117) becomes the Poisson equation

$$\nabla^2 V = \frac{4\pi G}{c^2} \left\langle \Psi \right| \hat{T}_{00}^{(0)} \left| \Psi \right\rangle \tag{118}$$

for the potential $V = -\frac{c^2}{2}h_{00}$. This is of course a well known result: in the Newtonian limit, Einstein's equations tell that the matter distribution in space generates a Newtonian gravitational potential. In this case, however, the matter distribution is the expectation value of the quantized form of the stress-energy tensor.

The gravitational potential enters the matter Hamiltonian, which eventually drives the quantum dynamics. Within the linearized theory, the interaction Hamiltonian is (see section 3.2):

$$\hat{H}_{\rm int} = -\frac{1}{2} \int d^3 r \, h_{\mu\nu} \, \hat{T}^{\mu\nu} \,. \tag{119}$$

It is important to point out the difference with respect to a standard quantized theory of gravity. In the latter, $h_{\mu\nu}$ becomes an operator as well, simply by applying the correspondence principle to the perturbation $h_{\mu\nu}$ of the metric—and thereby treating the classical $h_{\mu\nu}$ like a field living on flat spacetime rather than a property of spacetime. In contrast to this, $h_{\mu\nu}$ here remains fundamentally classical. It is determined by the wave equation (117), which is to be understood as classical equation of motion.

In the Newtonian limit, where $\hat{T}_{00}^{(0)}$ is the dominant term of the energy-momentum tensor, the interaction Hamiltonian then becomes

$$\hat{H}_{\text{int}} = \int d^3 r \, V \, \hat{T}^{00} = -G \int d^3 r \, \int d^3 r' \, \frac{\langle \Psi \mid \hat{\varrho}(\mathbf{r}') \mid \Psi \rangle}{|\mathbf{r} - \mathbf{r}'|} \, \hat{\varrho}(\mathbf{r}) \,, \tag{120}$$

where (118) has been integrated, and $\hat{T}_{00}^{(0)} = c^2 \hat{\rho}$ in the nonrelativistic limit. The secondquantized mass density operator $\hat{\rho}$ is simply $m\hat{\psi}^{\dagger}\hat{\psi}$ when only one kind of particle is present. Therefore, following the standard procedure we end up with the Schrödinger– Newton equation in Fock space:

$$\mathrm{i}\hbar\frac{\partial}{\partial t}|\Psi\rangle = \left[\int \mathrm{d}^3 r\,\hat{\psi}^{\dagger}(\mathbf{r})\left(-\frac{\hbar^2}{2m}\nabla^2\right)\hat{\psi}(\mathbf{r})\right]$$

Gravitational Decoherence

$$-G m^{2} \int d^{3}r \int d^{3}r' \frac{\langle \Psi | \hat{\psi}^{\dagger}(\mathbf{r}')\hat{\psi}(\mathbf{r}') | \Psi \rangle}{|\mathbf{r} - \mathbf{r}'|} \hat{\psi}^{\dagger}(\mathbf{r})\hat{\psi}(\mathbf{r}) \left] | \Psi \rangle.$$
(121)

In the nonrelativistic limit the number of particles is conserved and we can immediately write the N-particle first-quantized Schrödinger–Newton equation [132]:

$$i\hbar \frac{\partial}{\partial t} \psi_N(t, \{\mathbf{r}\}) = \left[-\sum_{i=1}^N \frac{\hbar^2}{2m} \nabla_i^2 -Gm^2 \sum_{i,j=1}^N \int \prod_{k=1}^N d^3 r'_k \frac{|\psi_N(t, \{\mathbf{r}'\})|^2}{|\mathbf{r}_i - \mathbf{r}'_j|} \right] \psi_N(t, \{\mathbf{r}\}), \qquad (122)$$

where $\{\mathbf{r}\} = \mathbf{r}_1, \ldots, \mathbf{r}_N$. The one-particle Schrödinger–Newton equation (114) follows directly by taking N = 1. One should stress once again that this nonlinear equation is the weak-field, Newtonian and nonrelativistic limit of (115), which itself is nonlinear with respect to the wave function and is assumed to be a fundamental equation.

Coming back to the attractive, "collapse-like" effect of the Schrödinger–Newton equation, one should point out an important difference with respect to the von Neumann collapse postulate, and with respect to collapse models. Let us consider an experiment where a particle's position is measured. Take an initial superposition state for the particle

$$\psi(\mathbf{r}) = \frac{1}{\sqrt{2}} (\psi_1(\mathbf{r}) + \psi_2(\mathbf{r})), \qquad (123)$$

where $\psi_1(\mathbf{r})$ and $\psi_2(\mathbf{r})$ are wave packets well localized around \mathbf{r}_1 and \mathbf{r}_2 , respectively. During the measurement, this state couples with the massive measuring instrument (say, a pointer) as follows:

$$\Psi(\mathbf{r}, \mathbf{R}) = \frac{1}{\sqrt{2}} (\psi_1(\mathbf{r}) \Phi_1(\mathbf{R}) + \psi_2(\mathbf{r}) \Phi_2(\mathbf{R})), \qquad (124)$$

where $\Phi_1(\mathbf{R})$ and $\Phi_2(\mathbf{R})$ are two localized wave functions of the pointer, centered around \mathbf{R}_1 and \mathbf{R}_2 , respectively. The positions $\mathbf{R} = \mathbf{R}_{1,2}$ correspond to the particle being around positions $\mathbf{r}_{1,2}$. Since the pointer is a classical system, according to the orthodox interpretation, the wave function collapses at $\mathbf{R} = \mathbf{R}_1$ or $\mathbf{R} = \mathbf{R}_2$, revealing in this way the outcome of the measurement. This means that the particle is found half of the times around the position \mathbf{r}_1 and half of the times around the position \mathbf{r}_2 . Collapse models provide a similar description.

According to the Schrödinger–Newton equation without the standard collapse postulate, on the other hand, a superposition state as in (124) implies a gravitational attraction between the spatial wave packets Φ_1 and Φ_2 representing the massive pointer, on top of the standard quantum dynamics. The wave function of the pointer "collapses" towards the average position $(\mathbf{R}_1 + \mathbf{R}_2)/2$, simply due to the symmetry of the deterministic dynamics and the initial state. Numerical simulations confirm this behavior of spatial superpositions collapsing to an average position [169]. Such a behavior is however in obvious contradiction with the standard collapse postulate, as well as with our everyday experience, where the pointer is found with equal probability either at $\mathbf{R} = \mathbf{R}_1$ or at $\mathbf{R} = \mathbf{R}_2$, and never in the middle. Moreover the Schrödinger–Newton equation is deterministic and as such it cannot explain why quantum measurements occur randomly, distributed according to the Born rule.

There is another undesired feature of the Schrödinger–Newton equation, connected to the previous result. Being a nonlinear and deterministic equation, it allows for superluminal signaling [105, 106]. This can be seen explicitly by considering Alice and Bob, apart from each other, each having a particle, the two being in a singlet spin state [184]. If Alice measures the spin along a chosen z direction, then in a way or another Bob's particle will have spin either up or down along the z direction (if this does not happen, then the situation violates experimental evidence). If he uses a Stern-Gerlach setup to decide in which spin state his particle is, the particle will be deflected either upwards or downwards. A spot will appear either in the upper or lower part of a screen positioned right after the setup.

If on the other hand Alice measures the spin along the x direction, then Bob's particle will have definite spin along the same direction and if he uses the previous Stern–Gerlach setup to decide the spin state of the particle (Alice is far away, he does not know which type of measurement she performed), then the particle's state will become a superposition of being deflected either upwards *and* downwards. Then the Schrödinger–Newton term will (slightly) attract the two pieces of the superposition towards each other, and in the end spots appear on the screen, but not in the same position as in the previous situation. In this way Bob can understand which type of measurement Alice performed, and the two can establish a protocol for superluminal signaling.

An important remark is at order. The kind of faster-than-light signaling here discussed is an effect of the (more or less) instantaneous collapse of the wave function (as a result of Alice's measurement), together with the nonlinear character of the dynamics described by the Schrödinger-Newton equation. Therefore, even if one describes the whole situation in a fully relativistic way (i.e. by some sort of "Dirac–Newton equation", which one could eventually obtain by applying (115) to a Dirac field), one would not get rid of the instantaneous collapse of the wave function upon measurement, nor of the nonlinear character of the dynamics. What would change is the way the two parts of the superposition attract each other: in the Schrödinger–Newton equation this attraction is instantaneous, while in the relativistic framework it would likely have a finite speed. This amounts in slight differences in the self-gravitation effects, which do not play any important role for the argument proposed here. As long as there is some measurable effect of self-gravitation, Bob can always exploit it to figure out Alice's measurement setting, and thereby receive a signal with the "speed of collapse" (which is infinite in the standard collapse prescription and has been shown to exceed the speed of light by orders of magnitude in a multitude of experiments [192, 193, 194]). Note that collapse models do not have this problem, as discussed before in this section.

Given the above arguments, the Schrödinger–Newton equation poses serious problems if regarded as a resolution of the measurement problem in quantum mechanics, equivalently as a way to describe the quantum-to-classical transition. It must be heavily modified in order to achieve this goal, perhaps to the point of betraying the original arguments that led to proposing it in the first place. On the other hand, it is an alternative to *quantum gravity*, meaning with it the program of quantizing gravity, and differs from it in a quantitative and computable way. It can serve as a model to decide whether gravity is quantum or not. For example, consider the question: what is the gravitational field generated by a quantum superposition in space? Is it the superposition of the two gravitational fields, as predicted by quantum gravity, or something else? The Schrödinger–Newton equation predicts it is something else—the classical sum of the two fields generated by the two terms of the superposition. Then matter in such a state should behave differently than what predicted by quantum gravity, even in the nonrelativistic and weak field limit. These differences can be quantified by solving (114) and detected, as discussed in section 4.2. The Schrödinger–Newton equation has the merit of providing a guideline for challenging quantum gravity at low energies, where experiments are cheaper and more flexible.

This is the subject of the next section. We will review experiments probing this interplay between quantum mechanics and gravity.

4. Table-top experiments to probe gravity

Here we summarize the state-of-the art of precise Newtonian and non-Newtonian gravity measurement and the feasibility of experimental tests of gravity effects in the quantum domain. We will mainly be concerned with table-top experiments in the nonrelativistic regime as such experiments may provide a new access to shine light on the quantum and gravity interplay. Therefore the main emphasis is to explore possible routes to enter the new parameter regime, where both quantum mechanics and gravity are significant, see figure 2. This means the mass of the object has to be large enough to show gravity effects while also not being too large to still allow for the preparation of a non-classical behavior of that massive object. That regime where both physical effects, the quantum and the gravity, could be expected to be relevant is at around the Planck mass [195], which is derived from the right mixture of fundamental constants (\hbar Planck's constant, c speed of light, G gravitational constant) $m_{\rm Pl} = \sqrt{\hbar c/G} = 2.176470(51) \times 10^{-8} \text{ kg}$ (based on the official CODATA value for all fundamental constants as announced by NIST [196]) or below. No quantum experiment has been performed in that mass range, but neither has an experiment to probe gravity. The smallest source masses which have been used to verify gravity are of the order of 10^{-2} kg which was at a distance of 10^{-2} m from the test mass [8].

When we refer to quantum mechanical behavior of massive systems, we mean the center-of-mass motion of such a system, which may consist of many atoms. Surely, there are many other (we call those internal) degrees of freedom of the same system such as electronic states or vibrations and rotations which are described as relative motions of the atoms forming the large object, but here we are not concerned with those. When we



Figure 2. Exploration map of mass: Mass range of the test mass as explored by experiments. Experiments to detect gravity have been done in the classical domain, *right hand side of picture*, with comparable large masses. Quantum experiments are routinely performed by using objects of much smaller masses so that gravity effects do not become visible or relevant. Neutron and atom matter-wave interferometers are different as the test mass there is very small (the mass of a single neutron or atom), but in a spatial superposition state. The desired mass range for—at least some of—the experiments summarized in this review article is at the overlap between sufficiently large mass to see significant effects of gravity of the particle itself, while the particle can be maintained in a non-classical state. The domain where massive particles can be prepared in such non-classical states is *on the left hand side* of the picture.

talk about superpositions, we mean spatial superpositions, in the sense of the center of mass of a single particle, which can be elementary or composite, being *here* and *there* at a given time (the Schrödinger cat state). The most massive complex quantum systems, which have been experimentally put in such a superposition state, are complex organic molecules of a mass on the order of $m_{\text{max}} = 10^{-22}$ kg [197, 198, 199].

While the emphasis of this review is not so much to explore the regime where a quantum system is coupled to a large external mass as in the seminal neutron interferometer experiment by Colella, Overhauser and Werner (COW) [9] or atom interferometry [200] experiments, we will also mention the state of the art of those. Strikingly, those matter-wave interferometry experiments show that gravity effects of comparably large external masses can be detected by quantum states of much lower mass objects, which may be used as motivation for perspectives of table-top experiments to investigate the interface of quantum and gravity, while at the same time it is clear that the type of gravity investigated in such cases is of Newtonian type in the first place. In any case we will try to work out the difference between such regimes—the low mass and the high mass regimes—to also emphasize the possibility to use photons, objects of zero rest mass, to watch out for gravity induced effects on quantum systems such as decoherence [201].



Figure 3. Neutrons to measure gravity: Left Panel: The schematics of the seminal COW experiment. Neutrons matter-wave functions are split and recombined by diffraction at solid crystals to from a closed Mach-Zehnder interferometer. The dimensions of the experimental setup are on the order of cm. The whole setup is then rotated with respect to the g-field. The respective gravitational phase shift is detected as a lost of fringe visibility of the interference pattern. Reprinted figure with permission from [9]. Copyright 1975 by the American Physical Society. Right Panel: A beam of neutrons is passes horizontally above a mirror surface. The neutron matter-waves are shown to be able to only occupy a discrete set of energy states which can be explained including the effect of Newtonian gravity. Reprinted by permission from Macmillan Publishers Ltd: Nature [10], copyright 2002.

There are also other proposals for cosmological effects related to gravity in the context of dark matter [202, 203] and dark energy [204] or for the detection of gravitational waves [205, 206], which have been proposed and some experiments have been done already with atom interferometers [207] and torsion balances [208], which we will not explore in too much detail here. See section 2.1 for an account on the decoherence of matter-waves by gravitational waves.

Typically for gravity experiments there are two masses involved, the (usually large) source mass which generates a significant curvature of spacetime (i. e. gravitational field or potential), and the test mass which is probing the gravity effect generated by the source mass. Torsion balances are the classic device for typical gravity experiments [209]. One can single out two regimes interesting for experimental investigation: i) The regime where a quantum system is the test mass and interacts with a large external source mass. This is the regime where neutron and atom interferometry are already very successful and provide tools for precise measurements of gravity effects. ii) The regime where the quantum system itself carries sufficient mass to be the source mass and to allow for related quantum gravity effects to become experimentally accessible. So far there has been no convincing experiment in the second regime. Any experiment performed in that second regime will ultimately give insight into the interplay between gravity and quantum mechanics. Test of the Schrödinger–Newton equation and of quantum effects in gravity fall in the latter regime. It may very well be that there are surprises waiting

Table 1. The table is adapted and reprinted from [196], with the permission of AIP Publishing. It shows all G measurements included for the CODATA value as well as their uncertainties $(\Delta G/G)$. Most of the measurements are taken by torsion balances, while only one, LENS-14, is an atom interferometer. NIST-82: National Institute of Standards and Technology, Gaithersburg, Maryland, and Boulder, Colorado, USA; TR&D-96: Tribotech Research and Development Company, Moscow, Russian Federation; LANL-97: Los Alamos National Laboratory, Los Alamos, New Mexico, USA; UWash-00: University of Washington, Seattle, Washington, USA; BIPM-01: International Bureau of Weights and Measures, Svres, France; UWup-02: University of Wuppertal, Wuppertal, Germany; MSL-03: Measurement Standards Laboratory, Lower Hutt, New Zealand; HUST-05 & HUST-09: Huazhong University of Science and Technology, Wuhan, PRC; UZur-06: University of Zurich, Zurich, Switzerland; JILA-10: JILA, University of Colorado and National Institute of Standards and Technology, Boulder, Colorado, USA; LENS-14: European Laboratory for Non-Linear Spectroscopy, University of Florence, Florence, Italy; UCI-14: University of California, Irvine, Irvine, California, USA.

Identification	$10^{11}G \;(\mathrm{m^{3}kg^{-1}s^{-2}})$	$\Delta G/G$
NIST-82, Fiber torsion balance	$6.672\ 48(43)$	6.4×10^{-5}
TR&D-96, Fiber torsion balance	6.672 9(5)	7.5×10^{-5}
LANL-97, Fiber torsion balance	6.673 98(70)	1.0×10^{-4}
UWash-00, Fiber torsion balance	$6.674\ 255(92)$	1.4×10^{-5}
BIPM-01, Strip torsion balance	6.675 59(27)	4.0×10^{-5}
UWup-02, Suspended body	$6.674\ 22(98)$	1.5×10^{-4}
MSL-03, Strip torsion balance	6.673 87(27)	4.0×10^{-5}
HUST-05, Fiber torsion balance	$6.672\ 22(87)$	1.3×10^{-5}
UZur-06, Stationary body	$6.674\ 25(12)$	1.9×10^{-5}
HUST-09, Fiber torsion balance	$6.673 \ 49(18)$	2.7×10^{-5}
JILA-10, Suspended body	$6.672 \ 34(14)$	2.1×10^{-5}
BIPM-14, Strip torsion balance	6.675 54(16)	2.4×10^{-5}
LENS-14, Double atom interferometer	$6.671 \ 91(99)$	1.5×10^{-4}
UCI-14, Cryogenic torsion balance	$6.674 \ 35(13)$	1.9×10^{-5}

for us if we become able to probe that regime by experiments.

The outline of the experimental section is as follows. We first summarize experimental tests of classical gravity by neutron and atom interferometers as well as by torsion balances and optomechanics experiments in section 4.1. Then we discuss experimental prospects for testing the Schrödinger–Newton equation in section 4.2, then proposed tests of quantum gravity in section 4.3. We then discuss proposals for tests of gravitational decoherence in section 4.4 as well as the idea to directly measure the gravity generated by a quantum superposition state in section 4.5.

4.1. Table-top experimental tests of Newtonian gravity and general relativity

In order to reach the regimes where both gravity and quantum effects must be taken into account, one can either start from microscopic quantum systems and attempt to increase there masses such that gravitational effects become relevant, or one approaches from the opposite direction of gravitating systems, reaching for ever smaller system sizes, in order to reach the quantum regime.

For the latter approach, we will summarize the state of the art to detect *classical* Newtonian gravity, as well as some general relativity effects, which amount essentially to observing the drop of different sorts of "apples" and measure the duration and distance of the drop with exceedingly precise methods. Effectively—while testing the universality of free fall, and therefore one component of Einstein's equivalence principle, which is the basis of the theory of general relativity—such experiments perform a measurement of the gravitational constants G and g; the latter in the case of a drop in the gravitational field of Earth and the former in relation to a further large source mass close to the test mass.

We should not forget that amongst the group of broadly accepted fundamental constants of Nature, the gravitational constant $G = 6.67408(31) \times 10^{-11} \text{m}^3 \text{kg}^{-1} \text{s}^{-2}$ [196], is the one which is known with lesser precision by many orders of magnitude, namely $\Delta G/G = 4.7 \times 10^{-5}$. This fact is always assumed to be reasoned by the comparably much weaker strength of the gravitational interaction relative to the other three known fundamental forces, which makes the determination of G by experiment notoriously harder. While the interaction itself can be measured with high precision (as outlined in the following), to know the exact distribution of mass in the bodies involved in the gravity test turns out to be the limiting factor [210]. See table 1 for a summary of all experiments to measure G which are included in the evaluation of the official G-value according to CODATA [196]. The data show the surprising wide spread of the experiment values. We briefly review how the different experimental schemes work, both for measurements of g and G, in classical and quantum set-ups.

A. Optical Interferometry of dropping corner cubes.—The drop of reflective corner cubes is used to measure free fall time in Earth's gravity. Classical optical interferometry is used for very precise measurement of the drop/fall time. This technology is portable [212] and a version has been used since the Apollo missions for lunar optical experiments [213]. The most precise version of a similar idea has been used in LISA pathfinder to realize a drag-free navigation of the satellite [214] by mutually stabilizing two masses by means of optical interferometry, while the gravitational wave detectors such as LIGO [12] can be regarded as a static version of this experimental scheme.

B. Neutron experiments.—The experiment by COW [9] is a recoverable matterwave dephasing effect rather than the destructive effect of gravitational decoherence. Depending on the orientation of the two arms of the Mach Zehnder type neutron interferometer with respect to the Earth's gravitational potential, the matter-waves acquire different gravitational phases, which upon recombination result in an overall phase shift, see figure 3 (left) for an illustration of the original experimental setup.

A further series of stunning experiment has been performed with neutrons traveling in the gravitational potential at the surface of the Earth [10]. Such experiments reveal the existence of a discrete set of states to be occupied by the neutron wave function in the classical gravitational potential of the Earth, see figure 3 (right).



Figure 4. Cold atoms to measure classical gravity: Left Panel: First experiment to detect gravitational acceleration by Earth (g) in atom interferometry. Reprinted by permission from Macmillan Publishers Ltd: Nature [200], copyright 1999. Middle Panel: The 10 m cold atom BEC fountain to generate single atom superposition states of 0.5 m size. Reprinted by permission from Macmillan Publishers Ltd: Nature [211], copyright 2015. Right Panel: The assembly of big source masses is around the tube containing two clouds of cold atom test masses. The systematic comparison of the result of the atom interferometry is performed on both clouds while moving the source masses. The experiment has been used for a measurement of G. Reprinted by permission from Macmillan Publishers Ltd: Nature [210], copyright 2014.

C. Atom interferometers.—The first atom interferometer experiment of the lightmatter beam splitter type to measure a gravitational effect has been performed by Peters [200] by measuring the effect of acceleration q on a cloud of cold atoms. The basic idea is similar to the COW neutron interferometer; the matter-wave couples to the external gravitational potential and acquires a related phase shift, which can in turn be measured very precisely by the interferometer, see figure 4 (left). That same experiment has been used for a high-precision gravity measurement [215]. Many revolutionary contributions to the field of detecting gravity with atom interferometers have been made by Kasevich [216, 217, 218, 211], see figure 4 (middle) for an illustration of the 10 m atomic fountain at Stanford. A very precise measurement of g has also been performed by an atom interferometer experiment with Bose–Einstein condensed (BEC) atoms [219], while G has also been measured by using a differential technique with two atomic clouds in different positions relative to an assembly of bigger source masses [210], see figure 4 (right). The technology of cold atom and BEC interferometers for gravity sensing is very mature and is on the verge to commercialization [220, 221]. It has been demonstrated to work in extreme environments such as drop towers based on BEC interferometry on atom chip technology [222, 223] and has been proposed to be used in space experiments [224] with first demonstrations of sounding rocket flights into space by the MAIUS collaboration [225].

D. Torsion pendulums.—While the first experiment has been performed by



Figure 5. Cavendish type torsion balance experiments: Left Panel: The scheme of the historic Cavendish experiment. The attraction between the big source mass and the smaller test masses is measured precisely by detecting the slow torsion angle change. Picture taken from [209]. Right Panel. The version of a modern torsion pendulum gravity sensor by Kapner et al. This is an upgraded version of the so-called missing mass torsion balance instrument. This experiment has been use to test extensions of Newtons square-law gravity by Yukawa type extensions as predicted by models for dark energy. Reprinted figure with permission from [208]. Copyright 2007 by the American Physical Society.

Cavendish in 1798 [209], the same basic principle is still used in the best experiments to measure classical gravity. A set of two test masses attached to a torsion pendulum is positioned close to a set of larger masses and the gravitational attraction is monitored by measuring precisely the torsion angle, while the test mass assembly is moving slowly towards the bigger masses, see figure 5 (left).

Modern versions of the same idea of a torsion balance are amongst the most sensitive gravity sensors, to test the inverse-square Newton law—even below the dark energy length scale [208], see figure 5 (right). The very diverse optomechanical devices are related and will be discussed in the next section.

E. Optomechanics experiments.—Optomechanical devices span a wide range of sizes and masses. Crucially the optomechanical coupling allows for both ultraprecise detection of the mechanical motion and preparation of sensitive classical and non-classical states of the mechanics [226]. Optomechanical devices have reached the quantum domain which promises for ultra-precise gravity sensing and even the exploration of the large mass quantum domain, see figure 2.

First optomechanical devices including torsion balance type sensors have been realized and approached the standard quantum limit in sensing [227]. A first proposal has been suggested to use optomechanical devices to perform gravity tests with mg masses [228], which would push down the low mass limit for gravity experiments. Here we refer to the mass of the test mass itself, instead of the interaction with a big external mass as for atom interferometry. Clamped cantilever type devices coupled to optical [229] and microwave cavities [230] have been realized as well as levitated optomechanical devices [138, 231, 232, 233, 234, 235, 236, 237, 238, 239]. The latter are very young, being seemingly a hybrid between the nanofabricated optomechanical structures and the atomic and molecular matter-wave systems, have a great potential for gravity sensing [240] as their mechanical quality factor (Q) is unprecedentedly high, as for example achieved for the driven rotation of a levitated nanorod: Q = 10^{11} [241]. Beside optical levitation, also magnetic levitation of superconductors is under development [242, 243]. All this is certainly of high potential, but still has to be shown to work as an ultra-precise gravity sensor in the quantum domain.

A clear challenge for all the nanometer and micrometer scale gravity tests—such as the optomechanics based experiments—is the competition with other interactions in the close proximity of surfaces, such as for example van der Waals and Casimir– Polder interactions. The latter can easily overwhelm the effects of gravity, while clever experimental tricks have been proposed to keep such dispersion force effects constant while the gravity effect is modulated [240]. Needless to say, all sorts of noise and heating effects generated by the trapping field, the environment, collisions with background gas particles, and so on, will set bounds on the detectable magnitude of the desired gravity effect.

F. Tests of general relativity by different types of experiments.—Beside the experiments to measure classical Newtonian gravity there are also attempts to detect effects as predicted by classical general relativity. At present, most experiments are on the proposal level or are attempted based on technologies with the highest sensitivities as general relativity effects are usually smaller than the Newtonian effects. No deviations from predictions of general relativity have been reported. Torsion pendulums, cold atom interferometers, and atomic clocks are typical (proposed and used) devices for such tests. This is a very broad and active field and we can here only mention some experiments performed. Examples include: i) Tests of the equivalence principle [244, 245, 246, 247]. A very creative idea for an experiment in this context involved mimicking of gravitation by acceleration of nanomechanical elements [248]. ii) Tests of the universality of free fall, where two test masses of different mass are dropped at the same time and their free fall times are compared. Such experiments have been performed with two different atomic species on Earth [249] and have been proposed for space [250]. iii) Tests of Lorentz invariance, a topic, with a dedicated review article [251]. iv) The proposals for using split path interferometers (for light or for matter) to detect general relativistic time dilation effects. We will come back to this topic in section 4.4 on gravitational decoherence. v) And last but not least, the general relativistic effect of *gravitational red shift* which has attracted a lot of discussion in the atom interferometry community [252, 253, 254, 255, 256, 257, 258].

There are many more experiments to test general relativity such as the Shapiro experiment [259], which are beyond the table-top of a laboratory and therefore we will not discuss them here.



Figure 6. Optomechanical experiments to test gravity: Left Panel: A testmass m (a) is loaded on a micromechanical device (b). A source mass M (c) is located at a COM distance d_0 from the test mass and is modulated through a drive motor (d) with maximum amplitude d_S . The displacement of the test mass cantilever is read out optically (e). Other, non-gravitational forces are further suppressed by a shielding membrane (f). (g) labels the mounting support structure. Picture and caption taken from [228]. Right Panel. Levitated optomechanics proposal and experiment to test gravity. Proposed experimental geometry. A sub-wavelength dielectric microsphere of radius a is trapped with light in an optical cavity. The sphere is positioned at an anti-node occurring at distance z from a gold-coated SiN membrane. Light of a second wavelength $\lambda_{cool} = 2\lambda_{trap}/3$ is used to simultaneously cool and measure the center-of-mass motion of the sphere. The sphere displacement δz results in a phase shift $\delta \phi$ in the cooling light reflected from the cavity. For the short-range gravity measurement, a source mass of thickness t with varying density sections is positioned on a movable element behind the mirror surface that oscillates harmonically with an amplitude δy . The source mass is coated with a thin layer of gold to provide an equipotential. Reprinted figure and caption with permission from [240]. Copyright 2010 by the American Physical Society.

4.2. Proposals for experimental tests of the Schrödinger-Newton equation

The Schrödinger–Newton equation assumes classical gravity to affect the wave function directly. In a way, the wave function acts as a mass (and not only a probability) distribution and the self-gravity of that spatial mass distribution affects the evolution of the spatial wave function. The Schrödinger–Newton equation is presented in section 3.4. So far, there has been no experimental test (direct or indirect) of predicted Schrödinger–Newton effects, while there are a number of proposals for such tests. We briefly review proposals for both direct and indirect experimental tests. We note that if the evolution of a matter-wave can be studied experimentally, it will as well provide a direct test of the quantum superposition principle [104].

The regime of the Schrödinger–Newton equation is sometimes called the semiclassical regime as gravity remains classical—even Newtonian—but this should not be confused with the regime of quantum field theory on curved spacetime, which is also sometimes called semi-classical. In the latter case, quantum matter fields are described in locally flat Minkowski frames on an overall curved (general relativistic) spacetime, which leads to the prediction of, e.g., the Unruh effect and Hawking radiation, and is therefore the mathematical foundation of black hole thermodynamics [18]. The curvature of spacetime is, however, given as an external feature in this latter case. The former case of the Schrödinger–Newton equation, to the contrary, assumes that spacetime must be treated as fundamentally classical, and describes the curvature resulting from the quantum matter itself, i. e. self-gravity.

It is obvious that semi-classical gravity, in connection with the Copenhagen view on quantum mechanics, allows for faster-than-light signaling, and is incompatible with the more radical extension of that view, the many worlds interpretation. The Schrödinger-Newton equation, therefore, can only seem reasonable when it is discussed in combination with the question of wave function collapse [184]. That obvious conflict between the two approaches holds the potential for an experimental test where two alternative views on the world predict a different outcome of the same experiment. The experiment will be of the type discussed in section 4.5. There has been already an attempt to perform an experiment [188] which was supposed to test semi-classical gravity in the sense of the Schrödinger-Newton equation. It is, however, a purely classical experiment together with incomplete theoretical arguments [182] which failed the condition to have, for the same degree of freedom, both sufficiently large mass, in order to see a gravity effect, and to be in a quantum mechanical state. It can therefore not be regarded as a valid experiment to test semi-classical gravity. (See also section 3.4 for a discussion of formal problems of the Schrödinger-Newton equation.)

Formally, the Schrödinger–Newton equation is very similar to the physics of manybody systems in the Hartree approximation, such as the nonlinear Gross–Pitaevskii equation [179, 180]. However, it is worth noting that the Schrödinger–Newton nonlinearity is delocalized over the range of the affected physical system, contrary to the Gross–Pitaevskii equation. The nonlinearity of the equation makes analytical solutions of the Schrödinger–Newton equation rather difficult to obtain, although some analytical properties are known [168].

4.2.1. Proposed direct tests of Schrödinger–Newton equation: wave function expansion The direct test of the Schrödinger–Newton equation is by studying the free expansion of the wave function of sufficiently massive objects. Then a contraction of the wave function according to the Schrödinger–Newton self-gravity effect should have a consequence on that expansion, competing with its natural Schrödinger's dynamics spread. Clearly, because of the weakness of the gravitation interaction, the mass has to be sufficiently large while the object has to remain in a quantum mechanical state which can be described by a center-of-mass quantum wave function, meaning the spatial extent of the wave function should be detectable for the full duration of the evolution. See figure 7 for the mass-time parameter space required to observe the predicted Schrödinger–Newton effect directly, which has been studied extensively [167, 170, 171, 260, 184, 261, 262], while analytical solutions of the Schrödinger–Newton equation are difficult and even numerical simulations are non-trivial.

One possible experimental scenario would be a molecule interferometry experiment [197], as such matter-wave experiments probe spatial superposition states of



Figure 7. Direct Test of Schrödinger–Newton wave function evolution: The mass-time plot to illustrate the parameter range which needs to be reached for direct Schrödinger–Newton wave function evolution experiments. This clearly needs to be done without external gravity and other forces/interactions and therefore an experiment in space appears a likely option. The red area shows the parameter range for a proposed space mission to test the Schrödinger–Newton effect.

large molecules. The Schrödinger–Newton contraction effect could also be observed for a free expansion of a singular wave function originated from a point in space, without the necessity of interferometric methods. The key is that the mass of the evolving quantum object has to be comparably large, much larger than the mass achieved in present molecule interferometry experiments. Cold atoms and even BECs of atoms (see section 4.1 C), which benefit from the multitude of coherent manipulation, control, and cooling schemes do not seem to have large enough mass in order to show the Schrödinger–Newton inhibition effect on the free expansion. Clearly, in order to measure the SN effect one needs both a large mass of the object and access to the coherent quantum evolution of the wave function. The high mass and the long expansion times to be studied challenge the experimental realization.

Should direct tests of the Schrödinger–Newton equation be done in space? At this point, there seems to be no other way to allow the wave function expansion for long enough, typically some hundred seconds, see figure 7. Proposals to levitate massive particles (optically or magnetically) and, therefore, to compensate for the drop in Earth's gravity have not been realized and are more problematic for Schrödinger–Newton tests. The levitated tests rely on proposed techniques to accelerate the wave function expansion artificially by optical or magnetic field gradients [242, 263]. That acceleration would have the potential to wash out completely the fragile Schrödinger–Newton effect.

4.2.2. Proposed indirect tests of Schrödinger–Newton equation Indirect Schrödinger– Newton effects have been predicted for optomechanical systems which are comparably



Figure 8. Indirect Tests of the Schrödinger-Newton equation: Left Panel: Phase space plot of mechanical squeezed state with extra rotation of state distribution according to the Schrödinger-Newton effect. Left side: according to standard quantum mechanics, both the vector $\langle \langle x \rangle, \langle p \rangle$ and the uncertainty ellipse of a Gaussian state for the center of mass of a macroscopic object rotate clockwise in phase space, at the same frequency $\omega = \omega_{CM}$. Right side: according to the center-of-mass Schrödinger-Newton equation, $(\langle x \rangle, \langle p \rangle)$ still rotates at ω_{CM} , but the uncertainty ellipse rotates at $\omega_q = (\omega_{CM}^2 + \omega_{SN}^2)^2$. Reprinted figure with permission from [173]. Copyright 2013 by the American Physical Society. Right Panel: Schematic overview of the effect of the Schrödinger–Newton equation the spectrum. The top part shows the first three energy eigenvalues and their shift due to the first order perturbative expansion of the Schrödinger–Newton potential. The bottom part shows the resulting spectrum of transition frequencies. In the narrow wave function regime (middle part), all energy levels are shifted down by an n-independent value minus an n-proportional contribution that scales with the inverse trap frequency. In the intermediate regime, where the wave function width becomes comparable to the localization length scale of the nuclei, this n-proportionality does no longer hold, leading to a removal of the degeneracy in the spectrum. Picture and caption taken from [174].

massive and on the verge to be quantum, see figure 2. Such effects are very small, can be overwhelmed by noise effects in the experiments, but can be done on the tabletop. Therefore, these tests represent a serious experimental challenge, while proposed to be possible with available technology. Two optomechanics experimental cases, as well as the study of the Schrödinger–Newton dynamics in nonlinear optics analogs are mentioned:

A. Schrödinger-Newton rotation of squeezed states.—The mechanical motion of an optomechanical system, clamped or levitated, is squeezed. Quantum squeezing of clamped optomechanics has been realized experimentally already [264, 265], while a classical analog has been demonstrated for a levitated system [239]. An optical homodyne detection of both field quadratures of the mechanical state is plotted and shows the cigar-shaped state, see in figure 8 (left). The Schrödinger–Newton equation predicts an extra rotation of the squeezed phase-space distribution [173], i. e. the internal oscillation of the squeezed state width $\langle (x - \langle x \rangle)^2 \rangle$ is not in phase with the external oscillation of position $\langle x \rangle$.

B. Schrödinger-Newton energy shifts of mechanical harmonic oscillator.— A further theoretical study [174] predicts Schrödinger-Newton related shifts of the energy

Gravitational Decoherence

levels of the stationary states of the quantum harmonic oscillator describing the optomechanical system, see for an illustration of the multiple energy shift effects figure 8 (right). There, different effects for the so-called wide and narrow wave function regimes are predicted for the situations that the spatial extent of the center-of-mass wave-function is larger (wide wave function regime) or smaller (narrow wave function regime) than the physical size of the massive object. A detailed experimental scenario has been worked out and awaits its realization in an actual laboratory.

C. Nonlinear optics simulation of the Schrödinger–Newton equation.—Specific delocalized nonlinearities in optical systems, typically just a piece of glass with a large refractive index, show a very similar type of dynamics for the propagation of light through that system if compared to Schrödinger–Newton dynamics. The analogue holds at least in (1+2) spacetime dimensions. The analogue provides an interesting option to study the dynamics of the Schrödinger–Newton equation in a parameter regime complementary to numeric simulations. Some experiments have been already performed [266, 267] to study cosmological settings of the Schrödinger–Newton equation such as exotic Boson stars. The main question remains, what can we ultimately learn from optics analogue experiments. Although these experiments do not genuinely test gravity, the study of the formally analogue dynamics, which is hard to calculate and even simulate otherwise, can lead to a better understanding of gravity, as well. See section 4.3 C for more tests of gravity based on analogues.

4.3. Proposed tests of Quantized Gravity

Here we look into predicted experimental tests for approaches that assume that gravity is quantum on the fundamental level, string theory and loop quantum gravity being the two most prevalent candidate theories. Arguments why gravity should be fundamentally quantum include a thermodynamic argument [268], and we will revisit in our brief section on the holographic principle. Again we focus on tests which are predicted for the low energy regime of table-top experiments.

The consequences of quantum gravity theories for experiments are very tiny effects and no test has been successfully performed yet. Also none of the candidate theories is regarded as a fully consistent theory of quantum gravity at their present stage of theoretical development. String theory is essentially an extension of quantum field theoretical principles from point particles to extended objects, strings, which uses flat Minkowski spacetime as its starting point [2], and introduces gravity by predicting a new particle, the graviton. A direct test in particle accelerator type experiments would take extremely high energies—far beyond the scales achievable in nonrelativistic table-top experiments. A potentially testable prediction on the table-top is related to the higher dimensional background spacetime used in string theory. This predicts a deviation of Newton's inverse square law for sufficiently large gravitational interaction strengths, i. e. in the close proximity of two masses. It is similar to fifth force tests by measuring the gravitational interaction at close proximity of masses where also effects of dark energy candidates [204], e. g. the Chameleon field are expected. Again, the most precise gravity sensors (torsion balances, atom interferometers, optomechanics) compete in these experiments, while no evidence for a deviation from the inverse square law has been measured [208, 204, 207, 269]. Loop quantum gravity follows a more conservative approach, starting from a re-formulation of general relativity, and attempting to directly (canonically) quantize it, thereby introducing operators for geometrical features such as length, area and volume [3, 270].

There are other quantum gravity predicted experimental consequences to be testable on the table-top for quantized gravity and we will mention some here.

A. Minimal length scale tests.—The basic assumption is that spacetime has a minimum length scale at which point it becomes discrete and therefore quantum, as explained in section 2.2.3. While it has been speculated that the Planck length could provide such a minimal length scale, there is no consensus that this is actually the case. There is already an excellent review of minimum length scale models and their possible experimental tests [271], so we do not repeat a review of those topics here. We only mention some aspects, which seem most interesting to us—mostly related to proposed experimental tests.

Some of the proposed experiments to measure physics at the Planck length are based on the related modification of quantum mechanical relations, such as the canonical uncertainty relation $\Delta x \Delta p \geq \hbar/2$ [272], which can be measured precisely in quantum optomechanical experiments [273]. The proposal is to measure Δx and Δp of the mechanical harmonic oscillator by quantum optics techniques to the ultimate precision, in order to detect a deviation as predicted by Planck-scale physics, such as doubly special relativity [274, 275], string theory [276, 277, 278], the principle of relative locality [279], and the deformation of the canonical commutator [280, 281]. The predicted effect is however tiny and optomechanics experiments have not reached the required sensitivity to observe any deviation.

No deviation from the standard quantum mechanical relations has been detected yet, while experiments at the sub-millikelvin cooled mechanical motion of the gravitational wave detector AURIGA have been used to set upper bounds on Planck scale modifications [282]. However, AURIGA weights several tons and, arguably, it is not a table-top experiment anymore. Such upper bounds have been lowered by an optomechanical experiment using masses of around the Planck mass, but in a purely classical motional state [283].

B. Bekenstein's idea for a table-top experiment to measure discreteness of spacetime.—This proposal is related to the minimum length scale ideas as it aims to resolve experimentally the discrete structure of spacetime as in principle predicted by the quantized version of general relativity. Bekenstein's idea was to hunt for discrete steps in the precisely detected motion of a macroscopic glass block. The discrete steps are thought to originate from ripples of spacetime at the Planck (or any other minimum) length. While not surprising, the interesting line of thought is that a mechanism is needed to amplify the (proposed) tiny rippled structure of spacetime in order to make

it detectable. In Bekenstein's proposal the amplification is realized by nonlinear optics effects. This is an approach which remains to appear fruitful to follow further.

Two major problems have been identified with this idea: 1) There is no definite size of those ripples of spacetime predicted by any quantum gravity theory or model. 2) The optical amplification principle is speculative and has not been demonstrated even for larger modulations of the motion of a macroscopic glass block. More clearly, the motion of the glass block has to be detected relative to the propagation of a single photon in a nonlinear optical medium (glass), while it is not clear if the motion of the photon itself is not also affected by the same quantum gravity effect which affects the motion of the macroscopic glass block.

However we remain positive about the initial idea as the proposal has certainly inspired experimentalist to think about possible tests of quantum gravity on the tabletop.

C. The idea of the holographic principle.—The idea of the holographic principle provides an interesting connection between gravity and thermodynamics. It is, that all physics (via the statistical interpretation of entropy) within an enclosed volume is encoded on its surface, was proposed first in the so-called brick wall model [284] and soon found adaptation for string theory in the so-called AdS/CFT correspondence [285]. AdS/CFT is very popular in string theory, as it provides an approach to circumvent non-trivial calculations of a higher dimension (n+1) theory (in anti deSitter spacetime, hence AdS) to the corresponding lower dimensional (n) conformal field theory (CFT). This approach ultimately leads to the lively discussed idea of the so-called firewall [286, 287].

The link to thermodynamics comes as physical degrees of freedom are considered, as they might appear on the event horizon of a black hole [288]. The mathematical framework of thermodynamics, which is the very same as quantum field theory on curved spacetime as discussed in section 4.2, allows for quantum effects in that gravity scenario, such as Hawking radiation [289]. In turn thermodynamic properties such as temperature and entropy can be associated with the extreme gravity situation of a black hole, and is therefore going beyond general relativity. While the holographic principle provides an exciting link between quantum mechanics and gravity, it remains very hard to test by any experiment.

If true, this principle means that the total number of degrees of freedom in the universe is finite, according to the Bekenstein bound [290], in contradiction to infinite Hilbert-space quantum mechanics. The fascinating question is if the principle is universal and therefore could in principle also be tested in a less general relativistic domain - such as by quantum table-top experiments. There is also an interesting and potentially fruitful link to information theory, through the link to von Neumann entropy. Interestingly, there is a link to quantum optics to so-called area laws, which use the same informational link of entropy between the area and the enclosed volume [291, 292]. This may open the door for experimental tests of the holographic principle by scenarios involving quantum entanglement in the laboratory.

Experiments with optical interferometers have been proposed to be used to test

Planck scale quantum geometry models, some of which are mentioned in section 4.3 A, including those with holographic bounds which induces noise in the interferometer measurements [293].

Again, as black holes and especially quantum physics on the black hole horizon are notoriously hard to reach by experiments, analogue experiments have been proposed to test formally equivalent physics on artificial event horizons, which can be realized in the laboratory [294, 295, 296, 297, 298]. Physical systems used to generate event horizons and to detect quantum physics there are nonlinear optics [299] and flowing water [300] amongst others. While analogue experiments clearly give insight into the dynamics of formal equivalent mathematics, it remains unclear, per definition, whether the same physics applies to the case involving actual gravity. Gravity analogue experiments certainly help to ask more precise questions about the case with gravity.

4.4. Gravitational decoherence effects

Tests of gravitational decoherence are based on the the straight-forward approach to generate a spatial superposition state (or any other non-classical state) of a massive particle and test if such a state decoheres according to (classical or quantum) gravity. Clearly, the experimental challenge is the preparation of such a state of sufficient mass. Typical experiments involve matter-wave interferometers and quantum optomechanics. The state of the art of experiments have been already covered in section 4.1. While the largest mass is given again by molecule interferometry, some of the effects (such as time dilation) are more promising to be tested in smaller mass systems such as cold atom interferometry, as those can be prepared in larger size superposition states to pick up a larger dephasing or decoherence effect. Although on first sight it appears that only massive systems can be used for the test, it becomes clear that general relativity effects also exist for photons [201].

A. Gravitational decoherence affecting superpositions.—In matter-wave interferometry experiments, the proposed effects from general relativistic time dilation [62, 20], discussed in section 2.3 may be tested. Time dilation leads to a dephasing effect in a matter-wave interferometer for the propagation of the wave function along the two different arms—ultimately resulting in a reduction of the visibility of the interference pattern. The effect has been predicted to scale with the number of all internal degrees of freedom, which are involved in the energy-momentum tensor on the right hand side of Einstein's equations (and which, therefore, affect the spacetime curvature and result in a gravitational force).

Atom interferometry tests of the time dilation effect appear most promising at the moment. They profit from the high control of the center-of-mass motion of cold atoms, e.g. in a 10 m fountain and with sensitivity on the verge of 10^{-19} , while the theoretical details of the effect are still debated. As a universal decoherence effect to explain the evident macroscopic quantum to classical transition, it is clear that time dilation decoherence, should it exist, is weaker by many order of magnitude than know environmental effects such as decoherence due to collisions by an even very diluted background gas [68], which leaves the usefulness of the general relativity effect in question.

To be more precise, each (internal) degree of freedom of the particle is regarded as a clock running at a typical frequency, but depending via general relativistic time dilation on the local gravitational environment. Then each single clock, if separated between the two different paths of an interferometer, will be sensitive to the relative duration of time and, therefore, dephase. This experiment has been realized as a proof of principle experiment with atomic chips [301], where the much larger spatial separation in other atomic interferometers [211] will help to improve the sensitivity to observe the predicted effect, discussed in section 2.3.

B. Gravitational effects in dynamical reduction models.—Dynamical reduction or collapse models have been formulated to explain the quantum to classical transition on a fundamental level and in complement to decoherence models [104]. While the physical reason for the collapse to occur is explained by the existence of a universal classical and random noise field, the physical origin of that field is still debated. The Diósi–Penrose model and Adler's model suggest that gravity is the cause of the collapse. This has been reviewed in section 3.

The best way to test those models is by large mass matter-wave interferometry, where the mass has to be beyond the presently reached limit of molecule interferometry by many orders of magnitude. This means that testing such models requires the preparation of large masses in non-classical states and optomechanical or magnetomechanical systems look most promising for the test [302, 303, 304]. Proposed experiments along those lines involve [305, 242].

Also indirect, non-interferometric experiments can be performed. Figure 1 shows the state of the art with respect to Adler's model. Future experiments, proposed in order to close the remaining gap in the parameter plot, involve those to generate large and massive quantum superpositions [306, 307, 242, 308, 309]. Such experiments are currently under development in the laboratories. A similar analysis for the Diósi– Penrose model (and Károlyházy's model) is still missing.

C. Gravitational wave induced decoherence.—Gravitational waves, which are a predicted consequence of general relativity [310, 311], have attracted much attention recently as a first experimental proof of their existence has been made by the LIGO experiment [12] in 2016, which is promising that gravitational waves may soon serve as new tool for astronomy. The LIGO experiment is a sophisticated version of a Michelson interferometer with kilometer long arms. It is, therefore, a large-scale physics experiment and we will not discuss it here in much more details. A first step has been taken to implement a space based version of a Michelson type gravitational wave interferometer with the very successful LISA pathfinder mission [214], which is of course also way beyond the scale of the table-top experiments discussed here.

There are further proposals to detect the effect of gravitational waves in more table-top like experiments. However, it will become clear that also the scale of these seemingly table-top ideas goes easily beyond the laboratory scale. As we have discussed in section 4.1 C, in atom interferometers there are sensitive spatial superposition states of massive particles which have been proposed to be used as a probe for gravitational waves. One can distinguish two ways to detect gravitational waves by atom interferometry, depending on whether gravitational waves are assumed to 'decohere' or 'not to decohere' matter-waves: The effect described by Lamine *et al* [31] is *decoherence* of the matterwave by collision with a gravitational wave, while in Dimopoulos *et al* [205] the matterwave undergoes a *dephasing*, which is precisely measured by a differential measurement between two widely separated atom interferometers using a common laser.

For the latter case and with technology such as the 10 m Stanford atomic fountain on ground with strain sensitivity of up to $10^{-19}/\sqrt{\text{Hz}}$ in the frequency band of 1– 10 Hz, the gravitational wave measurement would be complementary to LIGO. A recent proposal is about a technique to use atom interferometers for a single baseline detector for gravitational waves with improved strain sensitivity on ground [312]. Similar ideas based on atom interferometry have been proposed for a space based experiment in the LISA-type arrangement and the achievable strain sensitivities are predicted to be $10^{-20}/\sqrt{\text{Hz}}$ for the same gravitational wave frequency spectrum as LISA. A further low orbit scenario with a 30 km baseline has been proposed [313], aiming for strain sensitivity of less than $10^{-18}/\sqrt{\text{Hz}}$ in the gravitational wave frequency band of 50 mHz to 10 Hz, which would be complementary to LIGO-type instruments. This idea is the basis for a proposed space mission.

For the case of decoherence due to scattering with gravitational waves, where the gravitational wave hits the matter-wave, there is the HYPER proposal by Reynaud and co-workers [30]. They evaluate the significance of the decoherence process associated with the stochastic background of gravitational waves, and they show it has a tiny effect on HYPER-like atomic interferometers. In the work of Lamine *et al* [31] they work out that gravitational waves would ultimately limit matter-wave interferometers by a decoherence effect, while the effects are very tiny and much smaller than all other decoherence effects discussed in this review. They estimate that the decoherence effect on a matter-wave would be visible for a molecule interferometer type of experiment [83] of path separation on the order of 1 m, for 10^{-18} kg molecules traveling at a speed of 1 km/s—which is clearly way beyond existing experiments and technology in the laboratories, while not un-thinkable for a space based experiment. The interferometric path separation of 1 m is clearly the biggest experimental challenge.

Other ideas to probe gravitational waves on more table-top like experiments are to look for unique heating and noise effects in optical lattice atomic clocks [314], or with superfluid ⁴He droplets [315]. Furthermore, levitated optomechanics has been proposed to be used to probe gravitational waves [206], where they also work out how the effect to detect gravitational waves would scale with mass. Such an instrument could detect gravitational waves in the 50 kHz–330 kHz frequency range with a 10^3 times improved sensitivity compared to LIGO assuming a 150 nm radius levitated microdisk in a 100 m optical cavity. Clearly the 100 m cavity line is beyond a strict table-top scale.

Gravitational Decoherence

Long-term competitors of LIGO type optical interferometers to detect gravitational waves are the Schenberg instruments or parametric transducers [316] in Leiden and Sao Paulo which have the expected strain sensitivity of $10^{-22}/\sqrt{\text{Hz}}$ in the narrow frequency band of 3 kHz to 3.4 kHz. There, the gravitational wave scattering effect on the precisely controlled motion of a magnetically trapped macroscopic perfect sphere is used to detect gravitational waves.

D. Competing effects for matter-wave interferometry.—In order to be able to see such gravity effects and how they collapse or decohere the wave function in matterwave based experiments, all competing environmental decoherence processes have to be suppressed, which is the major experimental challenge in order to perform the experiments. Dominating decoherence effects are due to collisions with background gas, collisional decoherence [317], and the effects due to exchange of thermal radiation between the quantum system and the environment [318, 319, 308]. Magnetic levitation of superconducting microparticles, by definition, avoids all effects related to internal temperature radiation as the experiment is cryogenic and on top of that all noises related to lasers are removed as well [242]. This represents a huge advantage compared to optomechanics test. Furthermore, vibrations set serious constraints to all mechanics based tests of wave function collapse and gravity.

E. The case for space.—Ultimately a test of gravity decoherence and gravity induced collapse of the wave function would benefit from large masses of the particles in superposition states as well as long lifetimes of those superposition states in order to observe the extremely weak effects. The space proposal on macroscopic quantum resonators (MAQRO) [320] would be able to fulfill all those conditions. A community has started to work towards such a test in space and to propose a related mission. There are also many attempts to perform atom interferometry in space, which have been already mentioned and cited in section 4.1 C on atom interferometry and section 4.4 C on gravitational wave decoherence.

4.5. The gravity of a quantum state

A somehow different approach to the question of the interplay between quantum mechanics and gravity is given here. The setting goes back to Feynman [321] and amounts to the question: what is the gravitational field generated by a massive quantum superposition? Is it the superposition of the two gravitational fields generated by the two terms of the superposition, as predicted by quantum gravity? (Feynman thought this was the most logic answer.) Or is it the sum of the two gravitational fields, as predicted by the Schrödinger–Newton equation and perhaps by any consistent theory, which keeps gravity fundamentally classical? Or does the superposition decay because of gravity, as assumed by Penrose and developed in gravity-related collapse models?

This question challenges directly our understanding of gravity. But it goes further than this. It is directly related to another fundamental question: what is the mass, in a quantum universe? Is it a parameter entering the Schrödinger equation, more or less like the electric charge is? Or does it have a different origin, as general relativity suggests? Is the mass delocalized along with the wave function, or is it always localized in space?

For years these questions were purely speculative, because a detectable gravitational field requires a large enough mass, which can be hardly put in a spatial quantum superposition. This is why an experiment along these lines has never been done. But recent technological developments in quantum optomechanics are changing things. Though not in a truly quantum regime yet, appreciably large masses are controlled and manipulated with extreme accuracy, and very sensitive force and position measurements can be done. Now a feasible experimental proposal can be thought of, although the details still need to be sorted out [322].

A related proposal is to use the nonlinearity of gravity interaction to generate quantum entanglement between two distant massive systems in order to prove the fundamental non-classicality of gravity [323].

It is worth mentioning another interesting effect—retarded gravity—which has been speculated. Using the language of a classical gravity, the claim is that the center of the gravitational field does not coincide with the center of mass originating the field. As a consequence, there is a retarded gravity effect, which can be observed in specific experimental conditions [324, 325]. Experiments, most likely torsion balances or optomechanical devices, could aim to resolve it.

5. Conclusions

After about half a century of intense research in unifying quantum theory and general relativity, only recently a significant part of the scientific community started reformulating the problem from scratch, in a perhaps less ambitious but equally interesting way: to challenge the interplay between the two theories, both theoretically and experimentally, rather than trying to combine them in a single framework.

The field of gravitational decoherence is relatively young, and this explains the variety of approaches and the lack of unity, as it emerges from the present review. In this summary we will present some of the most relevant open problems, at least in our view, and possible future directions of research.

From the theoretical point of view, two lines of research have been initiated: to understand the effect of gravitational noise on quantum systems, at the nonrelativistic level, which is more easily accessible from the experimental point of view; and to speculate whether gravity might modify quantum theory, as a resolution of the quantum measurement problem. The first strategy was reviewed in section 2, the second in section 3.

As far as the first line is concerned, a rather large variety of models exists which take different approaches and, therefore, are hard to compare and put on a common ground. Most models are based on, or at least motivated by, assumptions about quantum gravity, such as the existence of a minimal length scale or a bath of thermalized gravitons. Interestingly, much less work has been focused on effects of classical spacetime fluctuations, and a comparison of the different approaches discussed in section 2 would be desirable. A comparison of the effects of a classical spacetime to quantum gravity seems necessary also, in order to identify effects that can truly be considered as quantum gravity fingerprints. Despite being the most straightforward one from its theoretical origin, the time dilation induced decoherence effect [20], which should be considered as an effect of non-inertial reference frames rather than a genuinely gravitational decoherence, only raised interest very recently. There is an ongoing debate about the interpretation and testability of this effect.

Coming to the second research line, the fact that the program of quantizing gravity turned out not to be as straightforward as with the other known forces, and is yet unresolved, together with the conceptual problems of quantum theory, has lead part of the scientific community to explore the possibility that quantum theory needs first to be reconsidered and that gravity might play a role in this; Penrose [4] refers to this as "gravitization of quantum mechanics", as opposed to the "quantization of gravity". As we have seen, the formulation of the problem is simple: quantum mechanics does not explain the absence of macroscopic superpositions. On the other hand, gravity is the stronger, the larger the system. Hence one might speculate that gravity plays a role in preventing macroscopic superpositions.

We have reviewed four options in section 3: the Diósi–Penrose model; Adler's idea on the existence of an irreducibly complex, rapidly fluctuating component of the metric; Károlyházy's model; and the Schrödinger-Newton equation. They first three models predict the existence of a stochastic gravitational background. The first two also predict a non-standard (in particular, nonlinear) coupling between matter and the background, which induces the collapse of the wave function in space, in agreement with the Born rule. The third one instead appears more like a standard decoherence model of a system in a noisy gravitational environment. The fourth model predicts a self-gravitational attraction of different parts of the wave function, opposing its natural spread.

These models do not resolve the quantum-gravity problem, and have very much the flavor of provisional models. So in this sense they cannot compete with much better developed theories such as string theory. However they are interesting and relevant, as they challenge the, so far incomplete, mainstream view according to which the world is quantum, gravity included. They are also interesting because they all predict effects, which can be tested with (a reasonable improvement of) current technology, as discussed in section 4.

From the theoretical point view, the obvious work which needs to be done, is to consolidate these models, ideally deriving them from what can be rightfully regarded as a new theory of matter and gravity. From a phenomenological point of view, it is crucial to conceive new experimental scenarios, where their effects can be tested, possibly in cheap table-top settings. And then experiments need to be done. Such experiments would probe the quantum-gravity interplay, one of the most fascinating and mysterious areas of current research.

From the experimental point of view, precise experiments have been performed to detect classical Newtonian gravity (G and g), which, however, are still less precise than the tests of all other fundamental constants. Such experiments include atom interferometers where the test mass is in a non-classical superposition state. Different values for G are found from different types of experiments [210]. The exact knowledge of the precise distribution of atoms in the typically big source mass has been identified as the limiting factor to increase precision further. It can be speculated that more precise tests of both G and g can be performed when both the test and the source mass are comparable in size, sufficiently massive to observe gravity effects, but well maintained within the quantum regime.

While matter-wave interferometer quantum experiments have been performed in the low mass regime, see figure 2, the higher mass range, all the way up to milligram masses—the working regime of torsion balances—is almost unexplored by any experiment and definitely not by any quantum experiment. Optomechanical devices are one candidate to bridge this enormous mass gap, while being in a quantum mechanical state and very massive at the same time. Especially levitated mechanical systems hold promise to test new physics in that new mass range. A variety of theoretical proposals and ideas for the interplay between quantum mechanics and gravity will become testable in the same mass range. Such ideas include the here discussed gravitationally induced decoherence and collapse models, the Schrödinger–Newton equation, the gravity of a quantum state, and quantum gravity effects.

Acknowledgments

We wish to thank colleagues for enlightening discussions on the topics discussed in this review—some of them quite some time ago: Stephen L. Adler, Peter Aichelburg, Markus Arndt, Markus Aspelmeyer, Peter Barker, James Bateman, Robert Beig, Jacob Bekenstein, Sougato Bose, Matteo Carlesso, Lajos Diósi, Sandro Donadi, Detlef Dürr, Daniele Faccio, Giulio Gasbarri, Domenico Giulini, Saikat Gosh, Klaus Hornberger, Bei-Lok Hu, Rainer Kaltenbaek, Myungshik Kim, Renate Loll, Stefan Nimmrichter, Mauro Paternostro, Igor Pikovski, Jess Riedel, Tejinder P. Singh, Andrea Smirne, Marko Toroš, Michael Vanner, Andrea Vinante. AB acknowledges financial support from the University of Trieste (FRA 2016) and INFN, AG acknowledged funding by the German Research Foundation (DFG), and HU acknowledges funding by the Leverhulme Trust (RPG-2016-046). The authors further thank ICTS Bangalore for hospitality during the program "Fundamental Problems in Quantum Physics", during which several of the ideas presented here were discussed.

References

 Will C M 1981 Theory and Experiment in Gravitational Physics (Cambridge: Cambridge University Press)

- [2] Green M B, Schwarz J H and Witten E 1987 Superstring theory (Vol. 1 and 2) (Cambridge: Cambridge University Press)
- [3] Rovelli C 1998 Liv. Rev. Relativ. 1 1 doi:10.12942/lrr-1998-1
- [4] Penrose R 2014 Found. Phys. 44 557–575 doi:10.1007/s10701-013-9770-0
- [5] Sorkin R D 2009 J. Phys.: Conf. Ser. 174 012018 doi:10.1088/1742-6596/174/1/012018
- [6] Dowker F 2011 J. Phys.: Conf. Ser. **306** 012016 doi:10.1088/1742-6596/306/1/012016
- [7] 't Hooft G 2016 The Cellular Automaton Interpretation of Quantum Mechanics (Fundamental Theories of Physics vol 185) (Cham: Springer)
- [8] Ritter R C, Goldblum C E, Ni W T, Gillies G T and Speake C C 1990 Phys. Rev. D 42 977–992 doi:10.1103/PhysRevD.42.977
- [9] Colella R, Overhauser A W and Werner S A 1975 Phys. Rev. Lett. 34 1472-1474 doi:10.1103/PhysRevLett.34.1472
- [10] Nesvizhevsky V V, Börner H G, Petukhov A K, Abele H, Baeßler S, Rueß F J, Stöferle T, Westphal A, Gagarski A M, Petrov G A et al. 2002 Nature 415 297–299 doi:10.1038/415297a
- [11] Fixler J B, Foster G T, McGuirk J M and Kasevich M A 2007 Science 315 74-77 doi:10.1126/science.1135459
- [12] Abbott B P et al (LIGO Scientific Collaboration and Virgo Collaboration) 2016 Phys. Rev. Lett.
 116 061102 doi:10.1103/PhysRevLett.116.061102
- [13] Amelino-Camelia G, Lämmerzahl C, Macias A and Müller H 2005 AIP Conf. Proc. 758 30 (Preprint gr-qc/0501053) doi:10.1063/1.1900507
- [14] Gorini V, Kossakowski A and Sudarshan E C G 1976 J. Math. Phys. 17 821–825 doi:10.1063/1.522979
- [15] Lindblad G 1976 Commun. Math. Phys. 48 119–130 doi:10.1007/BF01608499
- [16] Zeh H D 1986 Phys. Lett. A 116 9-12 doi:10.1016/0375-9601(86)90346-4
- [17] Kiefer C 1992 Phys. Rev. D 46 1658–1670 doi:10.1103/PhysRevD.46.1658
- [18] Wald R M 1994 Quantum Field Theory in Curved Spacetime and Black Hole Thermodynamics Chicago Lectures in Physics (Chicago: The University of Chicago Press)
- [19] Fredenhagen K, Rehren K H and Seiler E 2007 Quantum field theory: Where we are Approaches to Fundamental Physics. An Assessment of Current Theoretical Ideas Lecture Notes in Physics ed Stamatescu I O and Seiler E (Berlin: Springer) pp 61–87
- [20] Pikovski I, Zych M, Costa F and Brukner C 2015 Nat. Phys. 11 668-672 (Preprint 1311.1095 [quant-ph]) doi:10.1038/nphys3366
- [21] Wentzel G 1926 Z. Phys. 38 518-529 doi:10.1007/BF01397171
- [22] Kramers H A 1926 Z. Phys. 39 828-840 doi:10.1007/BF01451751
- [23] Brillouin L 1926 Comptes Rendus de l'Academie des Sciences 183 24–26
- [24] Plebanski J 1960 Phys. Rev. 118 1396–1408 doi:10.1103/PhysRev.118.1396
- [25] Linet B and Tourrenc P 1976 Can. J. Phys. 54 1129–1133 doi:10.1139/p76-136
- [26] Stodolsky L 1979 Gen. Relativ. Gravit. 11 391-405 doi:10.1007/BF00759302
- [27] Cai Y Q and Papini G 1989 Class. Quantum Grav. 6 407-418 doi:10.1088/0264-9381/6/3/017
- [28] Misner C W, Thorne K S and Wheeler J A 1973 Gravitation (San Francisco: W. H. Freeman and Company)
- [29] Sagnac G 1913 Comptes Rendus 157 708–710
- [30] Reynaud S, Lamine B, Lambrecht A, Maia Neto P and Jaekel M T 2004 Gen. Relativ. Gravit.
 36 2271–2288 doi:10.1023/B:GERG.0000046183.31629.02
- [31] Lamine B, Hervé R, Lambrecht A and Reynaud S 2006 Phys. Rev. Lett. 96 050405 doi:10.1103/PhysRevLett.96.050405
- [32] Sánchez Gómez J L 1993 Stochastic Evolution of Quantum States in Open Systems and in Measurement Processes (Singapore: World Scientific) chap Decoherence Through Stochastic Fluctuations of the Gravitational Field, pp 88–93
- [33] Power W L and Percival I C 2000 Proc. Roy. Soc. Lond. A 456 955-968 (Preprint quant-ph/9811059) doi:10.1098/rspa.2000.0544

- [34] Wald R M 1984 General Relativity (Chicago: University of Chicago Press)
- [35] Kar G, Sinha M and Roy S 1993 Int. J. Theor. Phys. 32 593-607 doi:10.1007/BF00673762
- [36] Breuer H P and Petruccione F 2002 The theory of open quantum systems (Oxford: Oxford University Press)
- [37] Göklü E and Lämmerzahl C 2008 Class. Quantum Grav. 25 105012 doi:10.1088/0264-9381/25/10/105012
- [38] Kiefer C and Singh T P 1991 Phys. Rev. D 44 1067-1076 doi:10.1103/PhysRevD.44.1067
- [39] Breuer H P, Göklü E and Lämmerzahl C 2009 Class. Quantum Grav. 26 105012 doi:10.1088/0264-9381/26/10/105012
- [40] Hu B L and Verdaguer E 2008 Liv. Rev. Relativ. 11 3 (Preprint 0802.0658 [gr-qc]) doi:10.12942/lrr-2008-3
- [41] Blencowe M P 2013 Phys. Rev. Lett. 111 021302 (Preprint 1211.4751 [quant-ph]) doi:10.1103/PhysRevLett.111.021302
- [42] Arteaga D, Parentani R and Verdaguer E 2004 Phys. Rev. D 70 044019 doi:10.1103/PhysRevD.70.044019
- [43] Calzetta E A and Hu B L B 2008 Nonequilibrium Quantum Field Theory Cambridge Monographs on Mathematical Physics (Cambridge: Cambridge University Press)
- [44] Feynman R P and Vernon F L J 1963 Ann. Phys. (N. Y.) 24 118–173 doi:10.1006/aphy.2000.6017
- [45] Caldeira A O and Leggett A J 1983 Physica A 121 587–616 doi:10.1016/0378-4371(83)90013-4
- [46] Anastopoulos C and Hu B L 2013 Class. Quantum Grav. 30 165007 doi:10.1088/0264-9381/30/16/165007
- [47] Arnowitt R, Deser S and Misner C W 1959 Phys. Rev. 116 1322–1330 doi:10.1103/PhysRev.116.1322
- [48] Oniga T and Wang C H T 2016 Phys. Rev. D 93 044027 (Preprint 1511.06678 [quant-ph]) doi:10.1103/PhysRevD.93.044027
- [49] Oniga T and Wang C H T 2016 Quantum dynamics of bound states under spacetime fluctuations (*Preprint* 1612.01097 [gr-qc])
- [50] Wang C H T, Bingham R and Mendonça J T 2006 Class. Quantum Grav. 23 L59–L65 doi:10.1088/0264-9381/23/18/L01
- [51] Peters A, Chung K Y, Young B, Hensley J and Chu S 1997 Phil. Trans. R. Soc. London A 355 2223–2233 ISSN 1364-503X doi:10.1098/rsta.1997.0121
- [52] Garay L J 1998 Phys. Rev. Lett. 80 2508-2511 doi:10.1103/PhysRevLett.80.2508
- [53] Garay L J 1999 Int. J. Mod. Phys. A 14 4079-4120 doi:10.1142/S0217751X99001913
- [54] Carlip S 1997 Phys. Rev. Lett. 79 4071-4074 doi:10.1103/PhysRevLett.79.4071
- [55] Erdős L 2010 Lecture notes on quantum Brownian motion (*Preprint* 1009.0843 [math-ph])
- [56] Anastopoulos C and Hu B L 2007 J. Phys.: Conf. Ser. 67 012012 doi:10.1088/1742-6596/67/1/012012
- [57] Mavromatos N E and Sarkar S 2006 Phys. Rev. D 74 036007 doi:10.1103/PhysRevD.74.036007
- [58] Kok P and Yurtsever U 2003 Phys. Rev. D 68 085006 doi:10.1103/PhysRevD.68.085006
- [59] Milburn G J 1991 Phys. Rev. A 44 5401-5406 doi:10.1103/PhysRevA.44.5401
- [60] Bonifacio R 1999 Il Nuovo Cimento B **114** 473–488
- [61] Anastopoulos C and Hu B L 2008 Class. Quantum Grav. 25 154003 doi:10.1088/0264-9381/25/15/154003
- [62] Zych M, Costa F, Pikovski I and Brukner C 2011 Nat. Commun. 2 505 doi:10.1038/ncomms1498
- [63] Bonder Y, Okon E and Sudarsky D 2016 Nat. Phys. 12 2 (Preprint 1507.05320 [quant-ph]) doi:10.1038/nphys3573
- [64] Bonder Y, Okon E and Sudarsky D 2015 Phys. Rev. D 92 124050 (Preprint 1509.04363 [gr-qc]) doi:10.1103/PhysRevD.92.124050
- [65] Diósi L 2015 Centre of mass decoherence due to time dilation: paradoxical frame-dependence

(*Preprint* 1507.05828 [quant-ph])

- [66] Adler S L and Bassi A 2016 Phys. Lett. A 380 390-393 (Preprint 1506.04414 [quant-ph]) doi:10.1016/j.physleta.2015.10.064
- [67] Zeh H D 2015 Comment on decoherence by time dilation (*Preprint* 1510.02239 [quant-ph])
- [68] Carlesso M and Bassi A 2016 Phys. Lett. A 380 2354-2358 (Preprint 1602.01979 [quant-ph]) doi:10.1016/j.physleta.2016.05.034
- [69] Pang B H, Chen Y and Khalili F Y 2016 Phys. Rev. Lett. 117 090401 (Preprint 1603.01984 [quant-ph]) doi:10.1103/PhysRevLett.117.090401
- [70] Pikovski I, Zych M, Costa F and Brukner C 2017 New J. Phys. 19 025011 arXiv:1508.03296
 [quant-ph] (Preprint 1508.03296 [quant-ph]) doi:10.1088/1367-2630/aa5d92
- [71] Toroš M, Großardt A and Bassi A 2017 Quantum mechanics for non-inertial observers (*Preprint* 1701.04298 [quant-ph])
- [72] Pikovski I, Zych M, Costa F and Brukner Č 2017 Comment on "quantum mechanics for noninertial observers" (*Preprint* 1702.06670 [quant-ph])
- [73] Toroš M, Großardt A and Bassi A 2017 Reply to comment on "quantum mechanics for non-inertial observers" (*Preprint* 1704.02285 [quant-ph])
- [74] Currie D G, Jordan T F and Sudarshan E C G 1963 Rev. Mod. Phys. 35 350-375 doi:10.1103/RevModPhys.35.350
- [75] Foldy L L and Wouthuysen S A 1950 Phys. Rev. 78 29-36 doi:10.1103/PhysRev.78.29
- [76] Lämmerzahl C 1995 Phys. Lett. A 203 12–17 doi:10.1016/0375-9601(95)00345-4
- [77] Pryce M H L 1948 Proc. Roy. Soc. Lond. A 195 62-81 doi:10.1098/rspa.1948.0103
- [78] Krajcik R A and Foldy L L 1974 Phys. Rev. D 10 1777-1795 doi:10.1103/PhysRevD.10.1777
- [79] Fulling S A 1973 Phys. Rev. D 7 2850-2862 doi:10.1103/PhysRevD.7.2850
- [80] Davies P C W 1975 J. Phys. A: Math. Gen. 8 609-616 doi:10.1088/0305-4470/8/4/022
- [81] Unruh W G 1976 Phys. Rev. D 14 870-892 doi:10.1103/PhysRevD.14.870
- [82] Hornberger K, Gerlich S, Haslinger P, Nimmrichter S and Arndt M 2012 Rev. Mod. Phys. 84 157-173 (Preprint 1109.5937 [quant-ph]) doi:10.1103/RevModPhys.84.157
- [83] Arndt M and Hornberger K 2014 Nat. Phys. 10 271-277 doi:10.1038/nphys2863
- [84] Cronin A D, Schmiedmayer J and Pritchard D E 2009 Rev. Mod. Phys. 81 1051-1129 doi:10.1103/RevModPhys.81.1051
- [85] Bell J S 1964 Physics 1 195–200
- [86] Bell J S 2004 Speakable and Unspeakable in Quantum Mechanics 2nd ed (Cambridge: Cambridge University Press)
- [87] Aspect A, Dalibard J and Roger G 1982 Phys. Rev. Lett. 49 1804-1807 doi:10.1103/PhysRevLett.49.1804
- [88] Dowling J P and Milburn G J 2003 Phil. Trans. R. Soc. London A 361 1655-1674 doi:10.1098/rsta.2003.1227
- [89] Schrödinger E 1935 Die Naturwissenschaften 23 807–812
- [90] von Neumann J 1955 Mathematical Foundations of Quantum Mechanics (Princeton, NJ: Princeton University Press)
- [91] Weinberg S 2012 Phys. Rev. A 85 062116 doi:10.1103/PhysRevA.85.062116
- [92] Broglie L D 1927 J. Phys. Radium 8 225-241 doi:10.1051/jphysrad:0192700805022500
- [93] Bohm D 1952 Phys. Rev. 85 166-179 doi:10.1103/PhysRev.85.166
- [94] Bohm D 1952 Phys. Rev. 85 180-193 doi:10.1103/PhysRev.85.180
- [95] Dürr D, Goldstein S and Zanghí N 1992 J. Stat. Phys. 67 843–907 (Preprint quant-ph/0308039) doi:10.1007/BF01049004
- [96] Holland P R 1993 The Quantum Theory of Motion: An Account of the de Broglie-Bohm Causal Interpretation of Quantum Mechanics (Cambridge: Cambridge University Press)
- [97] Dürr D and Teufel S 2009 Bohmian Mechanics: The Physics and Mathematics of Quantum Theory (Berlin: Springer)
- [98] Everett H I 1957 Rev. Mod. Phys. 29 454-462 doi:10.1103/RevModPhys.29.454

- [99] DeWitt B S 1970 Physics Today 23 30-35 doi:10.1063/1.3022331
- [100] Wallace D 2002 Studies in History and Philosophy of Science Part B: Studies in History and Philosophy of Modern Physics 33 637-661 (Preprint quant-ph/0103092) doi:10.1016/S1355-2198(02)00032-1
- [101] Ghirardi G C, Rimini A and Weber T 1986 Phys. Rev. D 34 470-491 doi:10.1103/PhysRevD.34.470
- [102] Ghirardi G C, Pearle P and Rimini A 1990 Phys. Rev. A 42 78-89 doi:10.1103/PhysRevA.42.78
- [103] Bassi A and Ghirardi G 2003 Phys. Rep. 379 257-426 (Preprint quant-ph/0302164) doi:10.1016/S0370-1573(03)00103-0
- [104] Bassi A, Lochan K, Satin S, Singh T P and Ulbricht H 2013 Rev. Mod. Phys. 85 471-527 (Preprint 1204.4325 [quant-ph]) doi:10.1103/RevModPhys.85.471
- [105] Gisin N 1989 Helv. Phys. Acta 62 363-371 doi:10.5169/seals-116034
- [106] Polchinski J 1991 Phys. Rev. Lett. 66 397-400 doi:10.1103/PhysRevLett.66.397
- [107] Wiseman H M and Diósi L 2001 Chem. Phys. 268 91-104 (Preprint quant-ph/0012016) doi:10.1016/S0301-0104(01)00296-8
- [108] Adler S L 2004 Quantum Theory as an Emergent Phenomenon: The Statistical Mechanics of Matrix Models as the Precursor of Quantum Field Theory (Cambridge: Cambridge University Press)
- [109] Bassi A, Dürr D and Hinrichs G 2013 Phys. Rev. Lett. 111 210401 arXiv:1303.4284 (Preprint 1303.4284 [quant-ph]) doi:10.1103/PhysRevLett.111.210401
- [110] Arnold L 1974 Stochastic Differential Equations: Theory and Applications (Wiley)
- [111] Adler S L, Brody D C, Brun T A and Hughston L P 2001 J. Phys. A: Math. Gen. 34 8795–8820 doi:10.1088/0305-4470/34/23/302
- [112] Adler S L and Brun T A 2001 J. Phys. A: Math. Gen. 34 4797–4809 doi:10.1088/0305-4470/34/23/302
- [113] Bassi A and Ghirardi G 2002 Phys. Rev. A 65 042114 doi:10.1103/PhysRevA.65.042114
- [114] Adler S L and Bassi A 2007 J. Phys. A: Math. Theor. 40 15083-15098 (Preprint 0708.3624 [quant-ph]) doi:10.1088/1751-8113/40/50/012
- [115] Adler S L and Bassi A 2008 J. Phys. A: Math. Theor. 41 395308 (Preprint 0807.2846 [quant-ph]) doi:10.1088/1751-8113/41/39/395308
- [116] Gasbarri G, Toroš M, Donadi S and Bassi A 2017 Gravity induced wave function collapse (Preprint 1701.02236 [quant-ph])
- [117] Ghirardi G, Grassi R and Rimini A 1990 Phys. Rev. A 42 1057–1064 doi:10.1103/PhysRevA.42.1057
- [118] Ghirardi G C, Grassi R and Benatti F 1995 Found. Phys. 25 5-38 doi:10.1007/BF02054655
- [119] Bassi A and Salvetti D G M 2007 J. Phys. A: Math. Theor. 40 9859–9876 (Preprint quant-ph/0702011) doi:10.1088/1751-8113/40/32/011
- [120] Bassi A, Ghirardi G and Salvetti D G M 2007 J. Phys. A: Math. Theor. 40 13755–13772 (Preprint 0707.2940) doi:10.1088/1751-8113/40/45/015
- [121] Diósi L 1989 Phys. Rev. A 40 1165–1174 doi:10.1103/PhysRevA.40.1165
- [122] Diósi L 2007 J. Phys. A: Math. Theor. 40 2989–2995 (Preprint quant-ph/0607110) doi:10.1088/1751-8113/40/12/S07
- [123] Diósi L 2013 Newton force from wave function collapse: speculation and test (Preprint 1312.6404 [quant-ph])
- [124] Diósi L 2014 New J. Phys. **16** 105006 (Preprint 1404.6644 [quant-ph]) doi:10.1088/1367-2630/16/10/105006
- [125] Adler S L 2016 Gravitation and the noise needed in objective reduction models Quantum Nonlocality and Reality: 50 Years of Bell's Theorem ed Bell M and Gao S (Cambridge: Cambridge University Press)
- [126] Károlyházy F 1966 Il Nuovo Cimento A 42 390-402 doi:10.1007/BF02717926
- [127] Károlyházy F, Frenkel A and Lukács B 1986 On the possible role of gravity in the reduction of

the wave function *Quantum Concepts in Space and Time* ed Penrose R and Isham C J (Oxford: Clarendon Press) pp 109–128

- [128] Károlyházy F 1974 Magyar Fizikai Folyóirat 22 23-85
- [129] Károlyházy F 1990 The breakdown of the superposition principle Sixty-Two Years of Uncertainty: Historical, Philosophical, and Physical Inquiries into the Foundations of Quantum Mechanics ed Miller A I NATO Scientific Affairs Division (New York: Plenum Press) pp 215–233
- [130] Károlyházy F 1995 State vector reduction via spacetime imprecision Fundamental Problems in Quantum Physics ed Ferrero M and van der Merwe A (Dordrecht: Springer) chap 16, pp 157–169
- [131] Károlyházy F, Frenkel A and Lukács B 1982 On the possibility of observing the eventual breakdown of the superposition principle *Physics as Natural Philosophy: Essays in Honor* of Laszlo Tisza on His Seventy-Fifth Birthday ed Shimony A and Feshbach H (Cambridge, MA: MIT Press) chap 3, pp 204–240
- [132] Diósi L 1984 Phys. Lett. A 105 199-202 doi:10.1016/0375-9601(84)90397-9
- [133] Penrose R 1996 Gen. Relativ. Gravit. 28 581-600 doi:10.1007/BF02105068
- [134] Penrose R 1998 Phil. Trans. R. Soc. London A 356 1927-1939 doi:10.1098/rsta.1998.0256
- [135] Diósi L 1986 Egy kvantum-sztochasztikus gravitációmodell és a hullámfüggvényredukció modell (aquantum-stochastic qravity and the wave function reduction) "candidate" thesis Hungarian Academy of Sciences Budapest URL http://www.rmki.kfki.hu/%7Ediosi/pub%5FQG.html
- [136] Diósi L 1987 Phys. Lett. A 120 377–381 doi:10.1016/0375-9601(87)90681-5
- [137] Mandel L and Wolf E 1995 Optical Coherence and Quantum Optics (Cambridge: Cambridge University Press) chap 15.5.3, p 771
- [138] Chang D E, Regal C A, Papp S B, Wilson D J, Ye J, Painter O, Kimble H J and Zoller P 2010 Proc. Natl. Acad. Sci. U.S.A. 107 1005–1010 doi:10.1073/pnas.0912969107
- [139] Savage C M and Walls D F 1985 Phys. Rev. A 32 2316-2323 doi:10.1103/PhysRevA.32.2316
- [140] Smirne A and Vacchini B 2010 Phys. Rev. A 82 042111 doi:10.1103/PhysRevA.82.042111
- [141] Penrose R 1994 Non-locality and objectivity in quantum state reduction Quantum Coherence and Reality: In Celebration of the 60th Birthday of Yakir Aharonov ed Anandan J S and Safko J L (Singapore: World Scientific) chap 6, p 238
- [142] Penrose R 1994 Shadows of the Mind: A Search for the Missing Science of Consciousness (Oxford: Oxford University Press)
- [143] Penrose R 1998 Gravity and quantum mechanics General Relativity and Gravitation 11. Part I: Plenary Lectures 1992 ed Gleiser R J, Kozameh C N and Moreschi O M (Bristol: IOP Publishing)
- [144] Bahrami M, Smirne A and Bassi A 2014 Phys. Rev. A 90 062105 doi:10.1103/PhysRevA.90.062105
- [145] Smirne A and Bassi A 2015 Sci. Rep. 5 12518 doi:10.1038/srep12518
- [146] Bassi A, Ippoliti E and Vacchini B 2005 J. Phys. A: Math. Gen. 38 8017–8038 doi:10.1088/0305-4470/38/37/007
- [147] Smirne A, Vacchini B and Bassi A 2014 Phys. Rev. A 90 062135 doi:10.1103/PhysRevA.90.062135
- [148] Bertulani C A and Danielewicz P 2004 Introduction to Nuclear Reactions (London: IOP Publishing)
- [149] Bahrami M, Paternostro M, Bassi A and Ulbricht H 2014 Phys. Rev. Lett. 112 210404 doi:10.1103/PhysRevLett.112.210404
- [150] Nimmrichter S, Hornberger K and Hammerer K 2014 Phys. Rev. Lett. 113 020405 doi:10.1103/PhysRevLett.113.020405
- [151] Diósi L 2015 Phys. Rev. Lett. **114** 050403 (Preprint 1411.4341) doi:10.1103/PhysRevLett.114.050403
- [152] Li J, Zippilli S, Zhang J and Vitali D 2016 Phys. Rev. A **93** 050102(R)

doi:10.1103/PhysRevA.93.050102

- [153] Vinante A, Bahrami M, Bassi A, Usenko O, Wijts G and Oosterkamp T 2016 Phys. Rev. Lett. 116 090402 doi:10.1103/PhysRevLett.116.090402
- [154] Laloë F, Mullin W J and Pearle P 2014 Phys. Rev. A 90 052119 doi:10.1103/PhysRevA.90.052119
- [155] Bilardello M, Donadi S, Vinante A and Bassi A 2016 Physica A 462 764-782 (Preprint 1605.01891 [quant-ph]) doi:10.1016/j.physa.2016.06.134
- [156] Adler S L, Bassi A and Donadi S 2013 J. Phys. A: Math. Theor. 46 245304 (Preprint 1011.3941 [quant-ph]) doi:10.1088/1751-8113/46/24/245304
- [157] Donadi S, Deckert D A and Bassi A 2014 Ann. Phys. (N. Y.) 340 70-86 (Preprint 1307.1021 [quant-ph]) doi:10.1016/j.aop.2013.10.009
- [158] Curceanu C, Hiesmayr B C and Piscicchia K 2016 J. Adv. Phys. 4 263-266 (Preprint 1502.05961 [quant-ph]) doi:10.1166/jap.2015.1193
- [159] Curceanu C, Bartalucci S, Bassi A, Bazzi M, Bertolucci S, Berucci C, Bragadireanu A, Cargnelli M, Clozza A, De Paolis L, Di Matteo S, Donadi S, D'Uffizi A, Egger J P, Guaraldo C, Iliescu M, Ishiwatari T, Laubenstein M, Marton J, Milotti E, Pichler A, Pietreanu D, Piscicchia K, Ponta T, Sbardella E, Scordo A, Shi H, Sirghi D, Sirghi F, Sperandio L, Vazquez Doce O and Zmeskal J 2016 Found. Phys. 46 263–268 doi:10.1007/s10701-015-9923-4
- [160] Carlesso M, Bassi A, Falferi P and Vinante A 2016 Phys. Rev. D 94 124036 doi:doi.org/10.1103/PhysRevD.94.124036
- [161] Toroš M and Bassi A 2016 Bounds on collapse models from matter-wave interferometry (Preprint 1601.03672 [quant-ph])
- [162] Toroš M and Bassi A 2016 Bounds on collapse models from matter-wave interferometry: Calculational details (*Preprint* 1601.02931 [quant-ph])
- [163] Frenkel A 2002 Found. Phys. 32 751-771 doi:10.1023/A:1016057026165
- [164] Frenkel A 1990 Found. Phys. 20 159-188 doi:10.1007/BF00731645
- [165] Frenkel A 1995 The hazy spacetime of the károlyházy model of quantum mechanics Fundamental Problems in Quantum Physics ed Ferrero M and van der Merwe A (Dordrecht: Springer) chap 11, pp 89–103
- [166] Frenkel A 1997 The model of f. károlyházy and the desiderata of a. shimony for a modified quantum dynamics Experimental Metaphysics: Quantum Mechanical Studies for Abner Shimony, Volume One (Boston Studies in the Philosophy of Science vol 193) ed Cohen R S, Horne M and Stachel J (Dordrecht: Kluwer Academic Publishers) pp 39–61
- [167] Moroz I M, Penrose R and Tod P 1998 Class. Quantum Grav. 15 2733-2742 doi:10.1088/0264-9381/15/9/019
- [168] Tod P and Moroz I M 1999 Nonlinearity 12 201–216 doi:10.1088/0951-7715/12/2/002
- [169] Harrison R, Moroz I and Tod K P 2003 Nonlinearity 16 101-122 (Preprint math-ph/0208045 and math-ph/0208046) doi:10.1088/0951-7715/16/1/307
- [170] Carlip S 2008 Class. Quantum Grav. **25** 154010 (Preprint 0803.3456 [gr-qc]) doi:10.1088/0264-9381/25/15/154010
- [171] Giulini D and Großardt A 2011 Class. Quantum Grav. 28 195026 (Preprint 1105.1921 [gr-qc]) doi:10.1088/0264-9381/28/19/195026
- [172] Giulini D and Großardt A 2013 Class. Quantum Grav. 30 155018 (Preprint 1212.5146 [gr-qc]) doi:10.1088/0264-9381/30/15/155018
- [173] Yang H, Miao H, Lee D S, Helou B and Chen Y 2013 Phys. Rev. Lett. 110 170401 (Preprint 1210.0457 [gr-qc]) doi:10.1103/PhysRevLett.110.170401
- [174] Großardt A, Bateman J, Ulbricht H and Bassi A 2016 Phys. Rev. D 93 096003 (Preprint 1510.01696 [quant-ph]) doi:10.1103/PhysRevD.93.096003
- [175] Anastopoulos C and Hu B L 2014 Newton-Schrödinger equations are not derivable from General Relativity + Quantum Field Theory (*Preprint* 1402.3813 [gr-qc])
- [176] Anastopoulos C and Hu B L 2014 New J. Phys. 16 085007 arXiv:1403.4921 (Preprint

1403.4921 [quant-ph]) doi:10.1088/1367-2630/16/8/085007

- [177] Adler S L 2007 J. Phys. A: Math. Theor. 40 755-763 (Preprint quant-ph/0610255) doi:10.1088/1751-8113/40/4/011
- [178] Christian J 1997 Phys. Rev. D 56 4844-4877 (Preprint gr-qc/9701013) doi:10.1103/PhysRevD.56.4844
- [179] Gross E P 1961 Il Nuovo Cimento 20 454-477 doi:10.1007/BF02731494
- [180] Pitaevskii L P 1961 Sov. Phys. JETP 13 451-454
- [181] Kibble T W B 1981 Is a semi-classical theory of gravity viable? Quantum Gravity 2. A Second Oxford Symposium ed Isham C J, Penrose R and Sciama D W (New York: Oxford University Press) pp 63–80
- [182] Mattingly J 2005 Is quantum gravity necessary? Einstein Studies Volume 11. The Universe of General Relativity Einstein Studies ed Kox A J and Eisenstaedt J (Boston: Birkhäuser) chap 17, pp 327–338
- [183] Kiefer C 2007 Quantum Gravity 2nd ed (International Series of Monographs on Physics vol 124) (Oxford: Clarendon Press)
- [184] Bahrami M, Großardt A, Donadi S and Bassi A 2014 New J. Phys. 16 115007 (Preprint 1407.4370 [quant-ph]) doi:10.1088/1367-2630/16/11/115007
- [185] Møller C 1962 Les théories relativistes de la gravitation Colloques Internationaux CNRS vol 91 ed Lichnerowicz A and Tonnelat M A (CNRS, Paris)
- [186] Rosenfeld L 1963 Nucl. Phys. 40 353–356 doi:10.1016/0029-5582(63)90279-7
- [187] Eppley K and Hannah E 1977 Found. Phys. 7 51–68 doi:10.1007/BF00715241
- [188] Page D N and Geilker C D 1981 Phys. Rev. Lett. 47 979-982 doi:10.1103/PhysRevLett.47.979
- [189] Mattingly J 2006 Phys. Rev. D **73** 064025 (Preprint gr-qc/0601127) doi:10.1103/PhysRevD.73.064025
- [190] Albers M, Kiefer C and Reginatto M 2008 Phys. Rev. D 78 064051 (Preprint 0802.1978 [gr-qc]) doi:10.1103/PhysRevD.78.064051
- [191] Misner C W, Thorne K S and Wheeler J A 1973 Gravitation (San Francisco: W. H. Freeman and Company) chap 18, pp 435–447
- [192] Stefanov A, Zbinden H, Gisin N and Suarez A 2002 Phys. Rev. Lett. 88 120404 doi:10.1103/PhysRevLett.88.120404
- [193] Scarani V, Bancal J D, Suarez A and Gisin N 2014 Found. Phys. 44 523-531 (Preprint 1304.0532 [quant-ph]) doi:10.1007/s10701-014-9785-1
- [194] Bancal J D, Pironio S, Acín A, Liang Y C, Scarani V and Gisin N 2012 Nat. Phys. 8 867–870 doi:10.1038/nphys2460
- [195] Planck M 1906 Vorlesungen über die Theorie der Wärmestrahlung (Leipzig: Barth)
- [196] Mohr P J, Newell D B and Taylor B N 2016 J. Phys. Chem. Ref. Data 45 043102 doi:10.1063/1.4954402
- [197] Gerlich S, Hackermüller L, Hornberger K, Stibor A, Ulbricht H, Gring M, Goldfarb F, Savas T, Müri M, Mayor M and Arndt M 2007 Nat. Phys. 3 711–715 doi:10.1038/nphys701
- [198] Gerlich S, Eibenberger S, Tomandl M, Nimmrichter S, Hornberger K, Fagan P J, Tüxen J, Mayor M and Arndt M 2011 Nat. Commun. 2 263 doi:10.1038/ncomms1263
- [199] Eibenberger S, Gerlich S, Arndt M, Mayor M and Tüxen J 2013 Phys. Chem. Chem. Phys. 15 14696-14700 doi:10.1039/C3CP51500A
- [200] Peters A, Chung K Y and Chu S 1999 Nature 400 849-852 doi:10.1038/23655
- [201] Zych M, Costa F, Pikovski I, Ralph T C and Brukner Č 2012 Class. Quantum Grav. 29 224010 doi:10.1088/0264-9381/29/22/224010
- [202] Riedel C J 2013 Phys. Rev. D 88 116005 doi:10.1103/PhysRevD.88.116005
- [203] Bateman J, McHardy I, Merle A, Morris T R and Ulbricht H 2015 Sci. Rep. 5 8058 (Preprint 1405.5536 [hep-th]) doi:10.1038/srep08058
- [204] Burrage C, Copeland E J and Hinds E 2015 Journal of Cosmology and Astroparticle Physics 2015 042 (Preprint 1408.1409 [astro-ph.CO]) doi:10.1088/1475-7516/2015/03/042

- [205] Dimopoulos S, Graham P W, Hogan J M, Kasevich M A and Rajendran S 2009 Phys. Lett. B 678 37-40 doi:10.1016/j.physletb.2009.06.011
- [206] Arvanitaki A and Geraci A A 2013 Phys. Rev. Lett. 110 071105 doi:10.1103/PhysRevLett.110.071105
- [207] Hamilton P, Jaffe M, Haslinger P, Simmons Q, Müller H and Khoury J 2015 Science 349 849–851 (Preprint 1502.03888 [physics.atom-ph]) doi:10.1126/science.aaa8883
- [208] Kapner D J, Cook T S, Adelberger E G, Gundlach J H, Heckel B R, Hoyle C D and Swanson H E 2007 Phys. Rev. Lett. 98 021101 doi:10.1103/PhysRevLett.98.021101
- [209] Cavendish H 1798 Phil. Trans. R. Soc. London 88 469–526
- [210] Rosi G, Sorrentino F, Cacciapuoti L, Prevedelli M and Tino G 2014 Nature 510 518-521 doi:10.1038/nature13433
- [211] Kovachy T, Asenbaum P, Overstreet C, Donnelly C, Dickerson S, Sugarbaker A, Hogan J and Kasevich M 2015 Nature 528 530-533 doi:10.1038/nature16155
- [212] Zumberge M A, Rinker R and Faller J 1982 Metrologia 18 145–152 doi:10.1088/0026-1394/18/3/006
- [213] Dickey J O, Bender P L, Faller J E, Newhall X X, Ricklefs R L, Ries J G, Shelus P J, Veillet C, Whipple A L, Wiant J R, Williams J G and Yoder C F 1994 Science 265 482–490 doi:10.1126/science.265.5171.482
- [214] Armano M, Audley H, Auger G, Baird J T, Bassan M, Binetruy P, Born M, Bortoluzzi D, Brandt N, Caleno M, Carbone L, Cavalleri A, Cesarini A, Ciani G, Congedo G, Cruise A M, Danzmann K, de Deus Silva M, De Rosa R, Diaz-Aguiló M, Di Fiore L, Diepholz I, Dixon G, Dolesi R, Dunbar N, Ferraioli L, Ferroni V, Fichter W, Fitzsimons E D, Flatscher R, Freschi M, García Marín A F, García Marirrodriga C, Gerndt R, Gesa L, Gibert F, Giardini D, Giusteri R, Guzmán F, Grado A, Grimani C, Grynagier A, Grzymisch J, Harrison I, Heinzel G, Hewitson M, Hollington D, Hoyland D, Hueller M, Inchauspé H, Jennrich O, Jetzer P, Johann U, Johlander B, Karnesis N, Kaune B, Korsakova N, Killow C J, Lobo J A, Lloro I, Liu L, López-Zaragoza J P, Maarschalkerweerd R, Mance D, Martín V, Martin-Polo L, Martino J, Martin-Porqueras F, Madden S, Mateos I, McNamara P W, Mendes J, Mendes L, Monsky A, Nicolodi D, Nofrarias M, Paczkowski S, Perreur-Lloyd M, Petiteau A, Pivato P, Plagnol E, Prat P, Ragnit U, Raïs B, Ramos-Castro J, Reiche J, Robertson D I, Rozemeijer H, Rivas F, Russano G, Sanjuán J, Sarra P, Schleicher A, Shaul D, Slutsky J, Sopuerta C F, Stanga R, Steier F, Sumner T, Texier D, Thorpe J I, Trenkel C, Tröbs M, Tu H B, Vetrugno D, Vitale S, Wand V, Wanner G, Ward H, Warren C, Wass P J, Wealthy D, Weber W J, Wissel L, Wittchen A, Zambotti A, Zanoni C, Ziegler T, and Zweifel P 2016 Phys. Rev. Lett. 116 231101 doi:10.1103/PhysRevLett.116.231101
- [215] Peters A, Chung K Y and Chu S 2001 Metrologia 38 25-61 doi:10.1088/0026-1394/38/1/4
- [216] Kasevich M and Chu S 1991 Phys. Rev. Lett. 67 181-184 doi:10.1103/PhysRevLett.67.181
- [217] Snadden M, McGuirk J, Bouyer P, Haritos K and Kasevich M 1998 Phys. Rev. Lett. 81 971–974 doi:10.1103/PhysRevLett.81.971
- [218] Chiow S w, Kovachy T, Chien H C and Kasevich M A 2011 Phys. Rev. Lett. 107 130403 doi:10.1103/PhysRevLett.107.130403
- [219] Poli N, Wang F Y, Tarallo M, Alberti A, Prevedelli M and Tino G 2011 Phys. Rev. Lett. 106 038501 doi:10.1103/PhysRevLett.106.038501
- [220] Geiger R, Landragin A, Bresson A, Bidel Y, Battelier B, Bouyer P, Villing A, Moron F, Zahzam N, Cheinet P et al. 2011 Nat. Commun. 2 474 (Preprint 1109.5905 [physics.atom-ph]) doi:10.1038/ncomms1479
- [221] Bodart Q, Merlet S, Malossi N, Dos Santos F P, Bouyer P and Landragin A 2010 Appl. Phys. Lett. 96 134101 doi:10.1063/1.3373917
- [222] van Zoest T, Gaaloul N, Singh Y, Ahlers H, Herr W, Seidel S, Ertmer W, Rasel E, Eckart M, Kajari E et al. 2010 Science 328 1540-1543 doi:10.1126/science.1189164
- [223] Müntinga H, Ahlers H, Krutzik M, Wenzlawski A, Arnold S, Becker D, Bongs K, Dittus H,
Duncker H, Gaaloul N, Gherasim C, Giese E, Grzeschik C, Hänsch T W, Hellmig O, Herr W, Herrmann S, Kajari E, Kleinert S, Lämmerzahl C, Lewoczko-Adamczyk W, Malcolm J, Meyer N, Nolte R, Peters A, Popp M, Reichel J, Roura A, Rudolph J, Schiemangk M, Schneider M, Seidel S T, Sengstock K, Tamma V, Valenzuela T, Vogel A, Walser R, Wendrich T, Windpassinger P, Zeller W, van Zoest T, Ertmer W, Schleich W P and Rasel E M 2013 *Phys. Rev. Lett.* **110** 093602 (*Preprint* 1301.5883 [physics.atom-ph]) doi:10.1103/PhysRevLett.110.093602

- [224] Tino G M, Sorrentino F, Aguilera D, Battelier B, Bertoldi A, Bodart Q, Bongs K, Bouyer P, Braxmaier C, Cacciapuoti L, Gaaloul N, Gürlebeck N, Hauth M, Herrmann S, Krutzik M, Kubelka A, Landragin A, Milke A, Peters A, Rasel E M, Rocco E, Schubert C, Schuldt T, Sengstock K and Wicht A 2013 Nucl. Phys. B 243 203-217 doi:10.1016/j.nuclphysbps.2013.09.023
- [225] Lachmann M, Seidel S, Becker D, Ahlers H, Wendrich T, Grosse J, Müntinga H, Weps B, Dinkelaker A, Schkolnik V et al. 2017 Creating the first Bose-Einstein condensate in space 48th Annual Meeting of the APS Division of Atomic, Molecular and Optical Physics (Sacramento, CA: American Physical Society) URL https://meetings.aps.org/Meeting/DAMOP17/Session/J7.4
- [226] Aspelmeyer M, Kippenberg T J and Marquardt F 2014 Rev. Mod. Phys. 86 1391-1452 doi:10.1103/RevModPhys.86.1391
- [227] Kim P, Hauer B, Doolin C, Souris F and Davis J 2016 Nat. Commun. 7 13165 doi:10.1038/ncomms13165
- [228] Schmöle J, Dragosits M, Hepach H and Aspelmeyer M 2016 Class. Quantum Grav. 33 125031 doi:10.1088/0264-9381/33/12/125031
- [229] Norte R A, Moura J P and Gröblacher S 2016 Phys. Rev. Lett. 116 147202 doi:10.1103/PhysRevLett.116.147202
- [230] Teufel J, Donner T, Li D, Harlow J, Allman M, Cicak K, Sirois A, Whittaker J D, Lehnert K and Simmonds R W 2011 Nature 475 359–363 doi:10.1038/nature10261
- [231] Romero-Isart O, Juan M L, Quidant R and Cirac J I 2010 New J. Phys. 12 033015 doi:10.1088/1367-2630/12/3/033015
- [232] Barker P F 2010 Phys. Rev. Lett. 105 073002 doi:10.1103/PhysRevLett.105.073002
- [233] Neukirch L P and Vamivakas A N 2015 Contemporary Physics 56 48-62 doi:10.1080/00107514.2014.969492
- [234] Yin Z q, Geraci A A and Li T 2013 Int. J. Mod. Phys. B 27 1330018 doi:10.1142/S0217979213300181
- [235] Kiesel N, Blaser F, Delić U, Grass D, Kaltenbaek R and Aspelmeyer M 2013 Proc. Natl. Acad. Sci. U.S.A. 110 14180–14185 doi:10.1073/pnas.1309167110
- [236] Gieseler J, Deutsch B, Quidant R and Novotny L 2012 Phys. Rev. Lett. 109 103603 doi:10.1103/PhysRevLett.109.103603
- [237] Millen J, Fonseca P Z G, Mavrogordatos T, Monteiro T S and Barker P F 2015 Phys. Rev. Lett. 114 123602 doi:10.1103/PhysRevLett.114.123602
- [238] Asenbaum P, Kuhn S, Nimmrichter S, Sezer U and Arndt M 2013 Nat. Commun. 4 2743 doi:10.1038/ncomms3743
- [239] Rashid M, Tufarelli T, Bateman J, Vovrosh J, Hempston D, Kim M S and Ulbricht H 2016 Phys. Rev. Lett. 117 273601 doi:10.1103/PhysRevLett.117.273601
- [240] Geraci A A, Papp S B and Kitching J 2010 Phys. Rev. Lett. 105 101101 doi:10.1103/PhysRevLett.105.101101
- [241] Kuhn S, Stickler B A, Kosloff A, Patolsky F, Hornberger K, Arndt M and Millen J 2017 Optically driven ultra-stable nanomechanical rotor (*Preprint* 1702.07565 [quant-ph])
- [242] Pino H, Prat-Camps J, Sinha K, Venkatesh B P and Romero-Isart O 2016 Magnetic skatepark for quantum interference of a superconducting microsphere (*Preprint* 1603.01553 [quant-ph])
- [243] Prat-Camps J, Teo C, Rusconi C C, Wieczorek W and Romero-Isart O 2017 Ultrasensitive inertial

and force sensors with diamagnetically levitated magnets (*Preprint* 1703.00221 [quant-ph]) [244] Schlamminger S, Choi K Y, Wagner T A, Gundlach J H and Adelberger E G 2008 *Phys. Rev.*

- Lett. 100 041101 doi:10.1103/PhysRevLett.100.041101
- [245] Vessot R F C and Levine M W 1979 Gen. Relativ. Gravit. 10 181–204 doi:10.1007/BF00759854
- [246] Lämmerzahl C, Everitt C W F and Hehl F W (eds) 2001 Gyros, Clocks, Interferometers: Testing Relativistic Gravity in Space (Lecture Notes in Physics vol 562) (Berlin: Springer) doi:10.1007/3-540-40988-2
- [247] Hohensee M A, Chu S, Peters A and Müller H 2011 Phys. Rev. Lett. 106 151102 doi:10.1103/PhysRevLett.106.151102
- [248] Katz B, Blencowe M and Schwab K 2015 Phys. Rev. A 92 042104 doi:10.1103/PhysRevA.92.042104
- [249] Schlippert D, Hartwig J, Albers H, Richardson L L, Schubert C, Roura A, Schleich W P, Ertmer W and Rasel E M 2014 Phys. Rev. Lett. 112 203002 doi:10.1103/PhysRevLett.112.203002
- [250] Aguilera D N, Ahlers H, Battelier B, Bawamia A, Bertoldi A, Bondarescu R, Bongs K, Bouyer P, Braxmaier C, Cacciapuoti L, Chaloner C, Chwalla M, Ertmer W, Franz M, Gaaloul N, Gehler M, Gerardi D, Gesa L, Gürlebeck N, Hartwig J, Hauth M, Hellmig O, Herr W, Herrmann S, Heske A, Hinton A, Ireland P, Jetzer P, Johann U, Krutzik M, Kubelka A, Lämmerzahl C, Landragin A, Lloro I, Massonnet D, Mateos I, Milke A, Nofrarias M, Oswald M, Peters A, Posso-Trujillo K, Rasel E, Rocco E, Roura A, Rudolph J, Schleich W, Schubert C, Schuldt T, Seidel S, Sengstock K, Sopuerta C F, Sorrentino F, Summers D, Tino G M, Trenkel C, Uzunoglu N, von Klitzing W, Walser R, Wendrich T, Wenzlawski A, Weßels P, Wicht A, Wille E, Williams M, Windpassinger P and Zahzam N 2014 Class. Quantum Grav. **31** 115010 doi:10.1088/0264-9381/31/11/115010
- [251] Mattingly D 2005 Liv. Rev. Relativ. 8 5 doi:10.12942/lrr-2005-5
- [252] Müller H, Peters A and Chu S 2010 Nature 463 926–930 doi:10.1038/nature08776
- [253] Wolf P, Blanchet L, Bordé C J, Reynaud S, Salomon C and Cohen-Tannoudji C 2010 Nature 467 E1 doi:10.1038/nature09340
- [254] Müller H, Peters A and Chu S 2010 Nature 467 E2 doi:10.1038/nature09341
- [255] Wolf P, Blanchet L, Bordé C J, Reynaud S, Salomon C and Cohen-Tannoudji C 2011 Class. Quantum Grav. 28 145017 doi:10.1088/0264-9381/28/14/145017
- [256] Hohensee M A, Chu S, Peters A and Müller H 2012 Class. Quantum Grav. 29 048001 doi:10.1088/0264-9381/29/4/048001
- [257] Giulini D 2012 Equivalence principle, quantum mechanics, and atom-interferometric tests Quantum Field Theory and Gravity. Conceptual and Mathematical Advances in the Search for a Unified Framework ed Finster F, Müller O, Nardmann M, Tolksdorf J and Zeidler E (Basel: Birkhäuser/Springer) pp 345–370 (Preprint 1105.0749 [gr-qc])
- [258] Lan S Y, Kuan P C, Estey B, English D, Brown J M, Hohensee M A and Müller H 2013 Science 339 554–557 doi:10.1126/science.1230767
- [259] Shapiro I I, Pettengill G H, Ash M E, Stone M L, Smith W B, Ingalls R P and Brockelman R A 1968 Phys. Rev. Lett. 20 1265–1269 doi:10.1103/PhysRevLett.20.1265
- [260] Giulini D and Großardt A 2012 Class. Quantum Grav. 29 215010 (Preprint 1206.4250 [gr-qc]) doi:10.1088/0264-9381/29/21/215010
- [261] Colin S, Durt T and Willox R 2016 Phys. Rev. A 93 062102 (Preprint 1402.5653 [quant-ph]) doi:10.1103/PhysRevA.93.062102
- [262] Großardt A 2016 Phys. Rev. A 94 022101 (Preprint 1503.02622 [gr-qc]) doi:10.1103/PhysRevA.94.022101
- [263] Scala M, Kim M, Morley G, Barker P and Bose S 2013 Phys. Rev. Lett. 111 180403 doi:10.1103/PhysRevLett.111.180403
- [264] Wollman E E, Lei C, Weinstein A, Suh J, Kronwald A, Marquardt F, Clerk A and Schwab K 2015 Science 349 952–955 (Preprint 1507.01662 [quant-ph]) doi:10.1126/science.aac5138
- [265] Pirkkalainen J M, Damskägg E, Brandt M, Massel F and Sillanpää M A 2015 Phys. Rev. Lett.

115 243601 doi:10.1103/PhysRevLett.115.243601

- [266] Bekenstein R, Schley R, Mutzafi M, Rotschild C and Segev M 2015 Nat. Phys. 11 872–878 doi:10.1038/nphys3451
- [267] Roger T, Maitland C, Wilson K, Westerberg N, Vocke D, Wright E M and Faccio D 2016 Nat. Commun. 7 13492 doi:10.1038/ncomms13492
- [268] Padmanabhan T 2011 J. Phys.: Conf. Ser. 306 012001 (Preprint 1012.4476 [gr-qc]) doi:10.1088/1742-6596/306/1/012001
- [269] Hsu J F, Ji P, Lewandowski C W and D?Urso B 2016 Sci. Rep. 6 30125 doi:10.1038/srep30125
- [270] Ashtekar A and Lewandowski J 2004 Class. Quantum Grav. 21 R53–R152 doi:10.1088/0264-9381/21/15/R01
- [271] Plato A D K, Hughes C N and Kim M S 2016 Contemporary Physics 57 477-495 doi:10.1080/00107514.2016.1153290
- [272] Garay L J 1995 Int. J. Mod. Phys. A 10 145-165 doi:10.1142/S0217751X95000085
- [273] Pikovski I, Vanner M R, Aspelmeyer M, Kim M S and Brukner C 2012 Nat. Phys. 8 393–397 (Preprint 1111.1979 [quant-ph]) doi:10.1038/nphys2262
- [274] Amelino-Camelia G 2002 Int. J. Mod. Phys. D 11 1643-1669 doi:10.1142/S021827180200302X
- [275] Magueijo J and Smolin L 2003 Phys. Rev. D 67 044017 doi:10.1103/PhysRevD.67.044017
- [276] Gross D J and Mende P F 1988 Nucl. Phys. B 303 407-454 doi:10.1016/0550-3213(88)90390-2
- [277] Amati D, Ciafaloni M and Veneziano G 1987 Phys. Lett. B 197 81–88 doi:10.1016/0370-2693(87)90346-7
- [278] Witten E 1996 Physics Today 49 24-30 doi:10.1063/1.881493
- [279] Amelino-Camelia G, Freidel L, Kowalski-Glikman J and Smolin L 2011 Phys. Rev. D 84 084010 doi:10.1103/PhysRevD.84.084010
- [280] Kempf A, Mangano G and Mann R B 1995 Phys. Rev. D 52 1108-1118 doi:10.1103/PhysRevD.52.1108
- [281] Das S and Vagenas E C 2008 Phys. Rev. Lett. **101** 221301 doi:10.1103/PhysRevLett.101.221301
- [282] Marin F, Marino F, Bonaldi M, Cerdonio M, Conti L, Falferi P, Mezzena R, Ortolan A, Prodi G A, Taffarello L et al. 2013 Nat. Phys. 9 71–73 doi:10.1038/nphys2503
- [283] Bawaj M, Biancofiore C, Bonaldi M, Bonfigli F, Borrielli A, Di Giuseppe G, Marconi L, Marino F, Natali R, Pontin A et al. 2015 Nat. Commun. 6 7503 doi:10.1038/ncomms8503
- [284] 't Hooft G 1985 Nucl. Phys. B 256 727-745 doi:10.1016/0550-3213(85)90418-3
- [285] Maldacena J, Gamboa Saravi R E, Falomir H and Schaposnik F A 1999 The large N limit of superconformal field theories and supergravity AIP Conference Proceedings vol 484 (AIP) pp 51–63 doi:10.1063/1.59653
- [286] Bousso R 1999 J. High Energy Phys. 1999 004 (Preprint hep-th/9905177) doi:10.1088/1126-6708/1999/07/004
- [287] Hayden P and Preskill J 2007 J. High Energy Phys. 2007 120 (Preprint 0708.4025 [hep-th]) doi:10.1088/1126-6708/2007/09/120
- [288] Bekenstein J D 1973 Phys. Rev. D 7 2333–2346 doi:10.1103/PhysRevD.7.2333
- [289] Hawking S W 1975 Commun. Math. Phys. 43 199-220 doi:10.1007/BF02345020
- [290] Bekenstein J D 1974 Phys. Rev. D 9 3292-3300 doi:10.1103/PhysRevD.9.3292
- [291] Verstraete F, Wolf M M, Perez-Garcia D and Cirac J I 2006 Phys. Rev. Lett. 96 220601 doi:10.1103/PhysRevLett.96.220601
- [292] Eisert J, Cramer M and Plenio M B 2010 Rev. Mod. Phys. 82 277-306 doi:10.1103/RevModPhys.82.277
- [293] Kwon O and Hogan C J 2016 Class. Quantum Grav. 33 105004 doi:10.1088/0264-9381/33/10/105004
- [294] Philbin T G, Kuklewicz C, Robertson S, Hill S, König F and Leonhardt U 2008 Science 319 1367-1370 (Preprint 0711.4796 [gr-qc]) doi:10.1126/science.1153625
- [295] Leonhardt U 2002 Nature 415 406-409 doi:10.1038/415406a

- [296] Unruh W G and Schützhold R (eds) 2007 Quantum Analogues: From Phase Transitions to Black Holes and Cosmology (Lecture Notes in Physics vol 718) (Berlin: Springer)
- [297] Novello M, Visser M and Volovik G E (eds) 2002 Artificial black holes (Singapore: World Scientific)
- [298] Volovik G E 2003 The Universe in a Helium Droplet (International Series of Monographs in Physics vol 117) (Oxford: Oxford University Press)
- [299] Belgiorno F, Cacciatori S, Clerici M, Gorini V, Ortenzi G, Rizzi L, Rubino E, Sala V and Faccio D 2010 Phys. Rev. Lett. 105 203901 doi:10.1103/PhysRevLett.105.203901
- [300] Weinfurtner S, Tedford E W, Penrice M C, Unruh W G and Lawrence G A 2011 Phys. Rev. Lett. 106 021302 doi:10.1103/PhysRevLett.106.021302
- [301] Margalit Y, Zhou Z, Machluf S, Rohrlich D, Japha Y and Folman R 2015 Science 349 1205-1208 (Preprint 1505.05765 [quant-ph]) doi:10.1126/science.aac6498
- [302] Bose S, Jacobs K and Knight P 1997 Phys. Rev. A 56 4175-4186 doi:10.1103/PhysRevA.56.4175
- [303] Bose S, Jacobs K and Knight P L 1999 Phys. Rev. A 59 3204-3210 doi:10.1103/PhysRevA.59.3204
- [304] Vanner M R, Pikovski I, Cole G D, Kim M, Brukner Č, Hammerer K, Milburn G J and Aspelmeyer M 2011 Proc. Natl. Acad. Sci. U.S.A. 108 16182–16187 doi:10.1073/pnas.1105098108
- [305] Marshall W, Simon C, Penrose R and Bouwmeester D 2003 Phys. Rev. Lett. 91 130401 doi:10.1103/PhysRevLett.91.130401
- [306] Romero-Isart O, Pflanzer A C, Blaser F, Kaltenbaek R, Kiesel N, Aspelmeyer M and Cirac J I 2011 Phys. Rev. Lett. 107 020405 doi:10.1103/PhysRevLett.107.020405
- [307] Asadian A, Brukner C and Rabl P 2014 Phys. Rev. Lett. 112 190402 doi:10.1103/PhysRevLett.112.190402
- [308] Bateman J, Nimmrichter S, Hornberger K and Ulbricht H 2014 Nat. Commun. 5 4788 (Preprint 1312.0500 [quant-ph]) doi:10.1038/ncomms5788
- [309] Wan C, Scala M, Morley G W, Rahman A A, Ulbricht H, Bateman J, Barker P, Bose S and Kim M 2016 Phys. Rev. Lett. 117 143003 doi:10.1103/PhysRevLett.117.143003
- [310] Einstein A 1916 Näherungsweise integration der feldgleichungen der gravitation
- [311] Ehlers J and Kundt W 1962 Exact solutions of the gravitational field equations Gravitation: An Introduction to Current Research ed Witten L (New York: Wiley) pp 49–101
- [312] Graham P W, Hogan J M, Kasevich M A and Rajendran S 2013 Phys. Rev. Lett. 110 171102 doi:10.1103/PhysRevLett.110.171102
- [313] Hogan J M, Johnson D M, Dickerson S, Kovachy T, Sugarbaker A, Chiow S w, Graham P W, Kasevich M A, Saif B, Rajendran S et al. 2011 Gen. Rel. Grav. 43 1953-2009 (Preprint 1009.2702 [physics.atom-ph]) doi:10.1007/s10714-011-1182-x
- [314] Kolkowitz S, Pikovski I, Langellier N, Lukin M D, Walsworth R L and Ye J 2016 Phys. Rev. D 94 124043 doi:10.1103/PhysRevD.94.124043
- [315] Singh S, Lorenzo L A D, Pikovski I and Schwab K C 2016 Detecting continuous gravitational waves with superfluid ⁴He (*Preprint* 1606.04980 [gr-qc])
- [316] Aguiar O D, Andrade L A, Barroso J J, Castro P J, Costa C A, de Souza S T, de Waard A, Fauth A C, Frajuca C, Frossati G, Furtado S R, Gratens X, Maffei T M A, Magalhes N S, Marinho Jr R M, Oliveira Jr N F, Pimentel G L, Remy M A, Tobar M E, Abdalla E, Alves M E S, Bessada D F A, Bortoli F S, Brando C S S, Costa K M F, de Arajo H A B, de Araujo J C N, de Gouveia Dal Pino E M, de Paula W, de Rey Neto E C, Evangelista E F D, Lenzi C H, Marranghello G F, Miranda O D, Oliveira S R, Opher R, Pereira E S, Stellati C and Weber J 2008 Class. Quantum Grav. 25 114042 doi:10.1088/0264-9381/25/11/114042
- [317] Hornberger K, Uttenthaler S, Brezger B, Hackermüller L, Arndt M and Zeilinger A 2003 Phys. Rev. Lett. 90 160401 doi:10.1103/PhysRevLett.90.160401
- [318] Hackermüller L, Hornberger K, Brezger B, Zeilinger A and Arndt M 2004 Nature 427 711-714 doi:10.1038/nature02276

- [319] Romero-Isart O 2011 Phys. Rev. A 84 052121 doi:10.1103/PhysRevA.84.052121
- [320] Kaltenbaek R, Aspelmeyer M, Barker P F, Bassi A, Bateman J, Bongs K, Bose S, Braxmaier C, Brukner Č, Christophe B, Chwalla M, Cohadon P F, Cruise A M, Curceanu C, Dholakia K, Diósi L, Döringshoff K, Ertmer W, Gieseler J, Gürlebeck N, Hechenblaikner G, Heidmann A, Herrmann S, Hossenfelder S, Johann U, Kiesel N, Kim M, Lämmerzahl C, Lambrecht A, Mazilu M, Milburn G J, Müller H, Novotny L, Paternostro M, Peters A, Pikovski I, Zanoni A P, Rasel E M, Reynaud S, Riedel C J, Rodrigues M, Rondin L, Roura A, Schleich W P, Schmiedmayer J, Schuldt T, Schwab K C, Tajmar M, Tino G M, Ulbricht H, Ursin R and Vedral V 2016 *EPJ Quant. Tech.* 3 5 (*Preprint* 1503.02640 [quant-ph]) doi:10.1140/epjqt/s40507-016-0043-7
- [321] Feynman R P, Morinigo F B and Wagner W G 1995 Feynman Lectures on Gravitation (Reading, MA: Addison-Wesley)
- [322] Bahrami M, Bassi A, McMillen S, Paternostro M and Ulbricht H 2015 Is gravity quantum? (Preprint 1507.05733 [quant-ph])
- [323] Kafri D, Milburn G J and Taylor J M 2015 New J. Phys. 17 015006 (Preprint 1404.3214 [quant-ph]) doi:10.1088/1367-2630/17/1/015006
- [324] Diósi L 2014 Equation and test of possible delay time of Newton force EPJ Web of Conferences vol 78 (EDP Sciences) p 02001 (Preprint 1406.1338 [gr-qc]) doi:10.1051/epjconf/20147802001
- [325] Yang H, Price L R, Smith N D, Adhikari R X, Miao H and Chen Y 2015 Towards the laboratory search for space-time dissipation (*Preprint* 1504.02545 [gr-qc])